

CIVIL SOCIETY MONITORING OF HARM REDUCTION IN EUROPE 2020

DATA REPORT

® Correlation -

European Harm Reduction Network, 2021.

This publication of Correlation – European Harm Reduction Network is protected by copyright. Reproduction is authorised provided the source is acknowledged.

Recommended citation:

Rigoni, R, Tammi, T, van der Gouwe, D, Oberzil, V,; Csak R, Schatz, E. (2021) Civil Society Monitoring of Harm Reduction in Europe, 2020. Executive Summary. Amsterdam, Correlation – European Harm Reduction Network.

*For the full report - Civil Society Monitoring of Harm Reduction in Europe, 2020. Data Report – please go to: https://www.correlation-net.org/monitoring/

Correlation – European Harm Reduction Network

c/o Foundation De REGENBOOG GROEP Droogbak 1d 1013 GE Amsterdam The Netherlands

www.correlation-net.org







Correlation - European Harm Reduction Network is co-funded by the European Union

CIVIL SOCIETY MONITORING OF HARM REDUCTION IN EUROPE 2020

DATA REPORT

Contributors C-EHRN Focal points

Country	ountry City Organization Main conte		Main contact	Function	Other acknowledged contributors		
Albania	Tirana	Aksion Plus	Besnik Hoxha	Psychologist			
Austria	Vienna	Suchthilfe Wien gGmbH	Birgit Braun	Operational manager, Daycentre Jed- mayer	Suchthilfe Wien, Z6, Kontaktladen and Streetwork Graz		
Belgium	Antwerp	vzw Free Clinic	Tessa Windelinckx	Coordinator GIG - health promotion in injecting drug use, Flanders			
Croatia	Rijeka	NGO for helping people with drug use disorders "Vida"	Karla Gla□ar	Social Peda- gogue	NGO Nada (Zadar) NGO Terra (Rijeka) NGO Zajednica Susret (Zagreb)		
Cyprus	Nicosia	Cyprus National Ad- dictions Authority	Josefina Mavrou	Officer	Ioanna Yiasemi		
					Evi Kyprianou		
					Vasilis Chrysanthou		
					Alexandros Tifas		
Czech Re- public	Prague	SANANIM	David Pesek	Manager of NSP drop-in centre			
Denmark	Copenha- gen	Health Team for the Homeless	Henrik Thiesen	Senior physi- cian, Manager	Jonas Demant, (Drug Users Academy), Peer Brehm Christensen (Profes- sor, Senior physician), Odense Kim Gosmer. (Chemical scientist, dev. & analysis medical cannabis)		
Estonia	Tallinn	NGO Convictus Estonia	Greete Org	Project manag- er, peer coun- sellor	Kristel Kivimets and Katri Abel - Ollo (National Institute for Health Dev., Centre for Drugs and Infec- tious Diseases) Mart Kalvet and Aleksandra Iru (Estonian association of people who use psychotropic substanc- es LUNEST)		
Finland	Helsinki	EHYT Ry/	Mika Mikkonen/	Manager/	Janne Nahkuri (A-Clinic Founda- tion), Juha Sederaren (Tukikohta		
		A-Clinic Foundation	Juho Sarvanko	Project coordi- nator	ry).		
France	Paris	Fédération Addic- tion	Marine Gaubert	Senior project manager	Laurent Michel (CSAPA Pierre Nicole and Sevag Chenorhokian (Sida Parole)		



Country	City	Organization	Main contact	Function	Other acknowledged contributors
Georgia	Tbilisi	Georgian Harm Re- duction Network	Maka Gogia	Programmes Director	Dali Usharidze (New Way), Zaza Karchkhadze (Rubiconi), Nino Janashia (Association Xe- non), George Soselia and Ina Inaridze (MDM Georgia), Temo Khatiashvili (Mandala)
Germany	Berlin	Deutsche Aidshilfe	Dirk Schäffer	Head of division	
Greece	Athens-Sa- Ionika	Positive Voice	Marios Atzemis	Harm Reduction Officer	
Hungary	Budapest	Rights Reporter Foundation	Peter Sarosi	Director	
Ireland	Dublin	Ana Liffey Drug Project	Tony Duffin	CEO	Nicola Perry (Community Re- sponse)
Italy	Milan/ Rome	Fondazione LILA Mi- lano Italian League for Fighting AIDS/ Forum Droghe	Lella Cosmaro / Antonella Cam- poseragna	Senior preven- tion officer/ Senior research- er	Massimo Oldrini, (LILA – Italian League for Fighting AIDS)
Lithuania	Vilnius	Coalition "I Can Live"	Jurgita Poskevi- ciute	Director of Ad- ministration	
Luxembourg	Luxem- bourg	Jugend - an Dro- genhëllef	Martina Kap	Chef de service	
Netherlands	Amsterdam	Mainline Founda- tion	Sara Woods	National policy officer	Cedric Charvet (De Regenboog Groep) Daniela van Santen, (Municipal Health Services Am- sterdam), Cas Isfordink (AMC) Floor van Bakkum and Raoul Koning (Jel- linek)
Norway	Kristiansand	proLAR Nett	Ronny Bjørnestad	Leader	
Poland	Krakow	MONAR – Krakow	Grzegorz Wodows- ki	MONAR-Krakow Coordinator	Bartosz Michalewski
Portugal	Vila Nova de Gaia	Agência Piaget Para o Desenvolvi- mento	Teresa Fernandes Sousa	Outreach Team Coordinator	Adriana Curado Grupo de Ativi- stas em Tratamento – GAT Kosmicare
North Macedonia	Skopje	HOPS - Healthy Op- tion Project Skopje	Silvana Naumova	Manager of Harm Reduction Programme	
Romania	Bucharest	CARUSEL	Zamfi Pinzariu	Social worker	
Russia	St. Peters- burg/	Charitable Fund "Humanitarian Ac- tion"/	Aleksey Lakhov/	Deputy Direc- tor/	
	Amsterdam	AFEW	Anke van Dam	Director	

Country	City	Organization	Main contact	Function	Other acknowledged contributors
Scotland ¹	Glasgow	Scottish Drugs Fo- rum	Dave Liddell	CEO	Kirsten Horsburgh, Katy McLeod and Leon Wylie (SDF)
Serbia	Novi Sad	Prevent	Nebojsa Durasovic	President of Pre- vent, President of Drug Policy Network South- East Europe (DPNSEE)	Milutin Milosevic, Drug Policy Network South-East Europe
Slovakia	Bratislava	Odyseus	Dominika Jase- kova	Executive Director	AABVIE Slovakia, medical staff, PWUD, Odyseus staff
Slovenia	Ljubljana	Association Stigma	Katja Krajnc	Social Worker	
Spain	Barcelona	Red Cross Catal- onia	Patricia Colomera Aguilà	Director of the attention and Monitoring cen- tre and harm reduction area	Carlos Belmar Ramírez. (Subdi- rección General de Coordina- ción de Programas Delegación del Gobierno para el Plan Nacio- nal sobre Drogas Ministerio de Sanidad, Consumo y Bienestar Social) Mireia Ventura (and team). Energy Control Asociación Bie- nestar y Desarrollo Elena Adan, Teresa de Gispert, Laia Gasulla, Joan Colom (and team). Sub-direcció General de Drogo- dependències Salut/Agència de Salut Pública de Catalunya Oleguer Parés i Gabriela Bar- gbalia (and team) Servei de Prevenció i Atenció a les Drogodependències Agèn- cia de Salut Pública de Barcelo- na (ASPB)
Sweden	Stockholm	Stockholm Drug Users Union	Niklas Eklund	Chairman	
Switzerland	Bern	Infodrog	Marc Marthaler	Harm Reduction coordinator	
Ukraine	Kiev	ICF "AIDS Foun- dation East-West" (AFEW-Ukraine)	Elena Voskresenskaya, Natalia Dvynskikh	Executive Director	Vielta Parkhomenko, (Chair of Club Eney, Coordinator of PUD. UA (VOLNA)); Evgeniya Kuvshy- nova (Executive director of Con- victus); Iryna Ivanchuk, (Head of Sector SMT and VH, Public Health Cen- ter MOH)
United King- dom	London	Release	Niamh Eastwood	Executive Director	Stuart Smith, Hepatitis C Trust UK



Scientific expert group

Daan van der Gouwe, Trimbos Institute Dirk Schäffer, Deutsche Aids Hilfe Mojca Matičič, University Ljubljana Rafaela Rigoni, C-EHRN Tuukka Tammi, THL

Expert groups

Overdose

Dirk Schäffer, Deutsche Aids Hilfe (Coordinator) Heino Stöver, Frankfurt University Mika Mikkonen, Finnish Association for Substance Abuse Prevention Ehyt ry

New Drug Trends

Daan van der Gouwe, Trimbos Institute (Coordinator) Magali Martinez, OFDT Mireia Ventura, Energy Control Tony Duffin, Ana Liffey Drug Project

Hepatitis C

Mojca Maticic, University Ljubljana (Coordinator) Erika Duffell, ECDC Juha Sedergren, Tukikohta ry Marie jauffret Roustide, Sante Publique France (Inserm) Ruth Zimmermann, Robert Koch Institut (RKI)

Authors

Chapters 1, 3, and 6: Rafaela Rigoni (C-EHRN) Chapter 2: Robert Csak (HRI) Chapters 4 and 5: Tuukka Tammi (THL) Chapter 7: Daan van der Gouwe (Trimbos) Chapter 8 : Victoria Oberzil (C-EHRN)

Report revision

Eberhard Schatz, Erika Duffell, Heino Stöver, Isabelle Giraudon, Jane Mounteney, Juha Sedergren, Katrin Schiffer Magali Martinez, Marie jauffret Roustide, Mireia Ventura, Mojca Matičič, Peter Sarosi, Rafaela Rigoni, Ruth Zimmermann, Sofia da Costa Cabral, Thomas Seyler and Tony Duffin.

Editor

Graham Shaw ® Correlation – European Harm Reduction Network, 2021.

This publication of Correlation – European Harm Reduction Network is protected by copyright. Reproduction is authorised provided the source is acknowledged.

This project has been supported by the European Commission.

Correlation - European Harm Reduction Network is co-funded by the European Union.

Recommended citation: Rigoni, R.; Tammi, T.; van der Gouwe, D.; Oberzil, V.; Csak, R.; Schatz, E. (2021) Civil Society Monitoring of Harm Reduction in Europe, 2020. Data Report. Amsterdam, Correlation – European Harm Reduction Network.

Contents

Preface	12
Acronyms and abbreviations	13
1: Introduction	15
The importance of CSO's in monitoring harm reduction	
Methodology	
Limitations	
Report structure	
2: Contextualising Harm Reduction in Europe	21
Needle and syringe programme implementation	24
Provision of opioid agonist therapy	25
Overdose prevention and drug consumption rooms	25
Viral hepatitis, tuberculosis and HIV	26
Harm reduction during the COVID-19 pandemic	26
3: Participation of Civil Society Organisations in policymaking	
Introduction	32
Cooperation between CSO's and policymakers	32
Cooperation mechanisms	32
Cooperation at the national level	33
Cooperation at the municipal level	40
CSO contributions are not always taken seriously	42
Civil Society networks and platforms	43
Civil Society contribution to data reporting	44
Conclusions	45



4: Essential harm reduction services	47
Differences between services and user groups	48
Highlight: Harm reduction in Tirana, Albania	51
How do cities compare with the national situation?	53
Improvements needed to harm reduction services	54
Conclusions	56
5: Hepatitis C	59
Introduction	60
Results	60
HCV prevention at current national policy level	60
National guidelines and real-life practices	61

6: Overdose prevention	
Introduction	70
OD prevention in official policies	70
What ideal guidelines should contain	72
Overdose context	73
Drugs involved in ODs	73
Characteristics and circumstances of OD victims	74
Challenges in OD prevention	75
Overdose prevention campaigns	76
Overdose prevention training	76
Desired improvements in overdose prevention	78
FP context compared to national context	
OD prevention for opioids	79
Naloxone availability	79
FP context compared to national context	85
Opioid Substitution Therapy (OST)	85
FP context compared to national context	88
Conclusions	89
7: Highlights in New Drug Trends	91
Introduction	
Adjustments to the questionnaire and introduction of focus groups	
New drug trends in C-EHRN FP cities	
Emergence of new substances	93
Known substances newly used by your target groups	94
New routes of administration	95
New combinations of substances	95
Changes in the target groups	96
New target groups	96



Challenges	97
Corona crisis	
No focus groups (as planned)	
Issues in interpreting the data	
What people think is in the substance versus lab-tested information	
Definition of 'a trend'	
Reliability of some answers	
Conclusions	
8: COVID-19 and Harm Reduction	
Introduction	
Results	
Challenges faced by harm reduction services	
Positive changes in harm reduction during the pandemic	
Difficulties faced by PWUD during the pandemic from the perspective of	107
harm reduction service providers	
Voices of PWUD during the pandemic	
Changes in drug use by PWUD during the pandemic	
Health of PWUD during the COVID-19 pandemic	
PWUD life under the COVID-19 pandemic	
Overdose and the COVID-19 pandemic	
Government response for PWUD during the COVID-19 pandemic	111
Role of CSO's during the pandemic	

Preface

This is the second civil society-led monitoring report produced by Correlation – European Harm Reduction Network (C-EHRN) within the context of our operational grant from the European Commission. The purpose of this report is to enrich the information and knowledge base of harm reduction interventions in Europe from the viewpoint of civil society organisations. We believe that this approach is a necessary, and useful, contribution to the development of drug policy in the region.

We learned a lot from the former process and the 2019 edition and modified the approach, the focus and certain questions to enable respondents to report closer to their own experiences. Consequently, the information provided in this report sometimes represents the situation in a particular city or region and informs us as to the experiences of a specific organisation in the field. Such 'real life' information can contribute significantly to an understanding of the advantages, barriers and challenges of drug policy. Even stronger, we directly approached representatives of networks and organizations of people who use drugs to share their view on the developments with us.

2020, however, was not a regular year and the world wide pandemic had its influence on people, organisations and the care system in general. Accordingly, we added a section on the impact of COVID-19 on harm reduction services and we will continue to measure this impact.

We will use the insights and information collected in this report within our advocacy efforts to strengthen harm reduction policies in Europe and, we hope, our partners and contributors will do the same in their environment at a regional and national level.

More than one hundred organisations and individuals from 34 European countries have contributed to the collection of data with an amazingly high response rate; we thank all contributors for their great work and commitment. Without their engagement, this work would never have been undertaken at all. In particular, we would like to thank the authors of this report, Rafaela Rigoni, Tuukka Tammi, Daan van der Gouwe, and Victoria Oberzil, who were supported by the coordinators of the expert groups and the reviewers of this report. We are also grateful to HRI and Robert Czack for contributing with a chapter to this report. A special thanks to Dagmar Hedrich and her EMCDDA colleagues for their ongoing and patient support.

We thank the European Commission, DG Sante, for their financial support and to the Regenboog Groep, Amsterdam, for their ongoing support of Correlation – European Harm Reduction Network.

Eberhard Schatz and Karin Schiffer

On behalf of the C-EHRN team



Acronyms and abbreviations

3-MMC	3-Methylmethcathinone, also known as Metaphedrone
AFVD	Association Francophone Pour Vaincre Les Douleurs; Francophone association to overcome pain (France)
Aides	A French community-based non-profit organisation
APDES	Agência Piaget para o Desenvolvi- mento
APSEP	Association des Professionnels de Santé exercant en Prison; Association of Health Professionals working in Prison (France)
ART	Antiretroviral Therapy
ASUD	Autosupport des usagers de dro- gues; Self-support for drug users (France)
C-EHRN	Correlation – European Harm Reduction Network
COPD	Chronic Obstructive Pulmonary Disease
COVID	Coronavirus Disease
CSFD	Civil Society Forum on Drugs
CSIDP	Civil Society Involvement in Drug Policy Project
CS	Civil Society
CSO	Civil Society Organisation
DAA	Direct-Acting Antiviral
DCR	Drug Consumption Room
EASL	European Association for the Study of the Liver
ECDC	European Centre for Disease Prevention and Control
EMCDDA	European Monitoring Centre for Drugs and Drug Addiction
ENPUD	European Drug User Union

ER	Emergency Room
EU	European Union
EuroNPUD	European Network of People Who Use Drugs
FP	Focal Point
FSW	Female Sex Worker
GFATM	Global Fund to Fight AIDS, Tuberculosis and Malaria
GHB	Gammahydroxybutrate, a depressant
GP	General Practitioner
HAT	Heroin Assisted Treatment
HCV	Hepatitis C Virus
HIV	Human Immunodeficiency Virus
HR	Harm Reduction
HRI	Harm Reduction International
HSE	Health Service Executive (Ireland)
ID	Identification
IEC	Information, Education, Communication
IHRD	International Harm Reduction Devel- opment Programme (Soros Foundation)
INPUD	International Network of People who Use Drugs
Lab	Laboratory
LGBT	Lesbian, Gay, Bisexual, Trans
LGBTI	Lesbian, Gay, Bisexual, Trans, and Intersex
lgbtqia+	Lesbian, Gay, Bisexual, Trans, Queer/ Questioning, Intersex, Asexual, all of the other sexualities, sexes, and gen- ders
LSD	Lysergic acid diethylamide, also known colloquially as acid; an hallucinogenic drug
Lube	Lubricant
MDMA	3,4-Methylenedioxymethamphet- amine, commonly known as ecstasy
MENA	Middle East and North Africa
MMT	Methadone Maintenance Treatment
MSM	Men having Sex with Men

NAAC	Cyprus National Addictions Authority	SEG	Scientific Expert Group
NDT	New Drug Trend	SFETD	La Société Française d'Etude et Trait-
NEP	Needle Exchange Programme		ement de la Douleur; French Society
NFP	National Focal Point		for the Study and Treatment of Pain
Nk	Not Known	SFSPO	Société Francophone des Sciences Pharmaceutiques Officinales;
NPS	New Psychoactive Substance		Francophone Society of Official
NSP	Needle and Syringe Programme		Pharmaceutical Sciences
OAT	Opioid Agonist Therapy	SICAD	Serviço de Intervenção nos
OD	Overdose		Comportamentos Aditivos e nas
OFMA	Observatoire Francais des Medica-		Dependências (Intervention Service in Addictive Behaviours and
	ments Antalgiques; French Observa- tory for Analgesic Drugs		Dependencies) (Portugal)
OST	Opioid Substitution Therapy	STI	Sexually Transmitted Infection
PDF	Portable Document Format	TB	Tuberculosis
PsychoActif	A community dedicated to informa-	THN	Take-Home Naloxone
,	tion, mutual aid, the exchange of	UNFPA	United Nations Population Fund, for-
	experiences and the construction of		merly the United Nations Fund
	knowledge on drugs, in a risk reduc- tion approach	UNODC	for Population Activities United Nations Office on
PWI	People Who Inject	UNODC	Drugs and Crime
PWID	People Who Inject Drugs	WHO	World Health Organization
PWS	People Who Smoke	-	

- PWU People Who Use
- PWUD People Who Use Drugs
- Reitox Réseau Européen d'Information sur les Drogues et les Toxicomanies; the European information network on drugs and drug addiction, which includes designated national institutions or agencies responsible for data collection and reporting to the EM-CDDA on drugs and drug addiction.
- RESPADD Réseau de prévention des addictions; a non-profit association which brings together more than 600 health establishments (hospitals, clinics, EH-PAD, medico-social establishments, etc.) engaged in the prevention and management of addictive practices. (France).



Introduction



The importance of CSO's in monitoring harm reduction

In the field of harm reduction, civil society organisations (CSO's) play an essential role in developing and implementing effective measures to address the negative consequences of drug use. They work directly for, and with, people who use drugs (PWUD) and have a good understanding of their daily needs. Their inside knowledge and information are critical in developing adequate drug policies and practices.

CSOs can act as transmission belts that filter societal preferences and channel them to policymakers. In practice, however, their capacity to effectively interact with policymakers varies considerably. Where a constructive and respectful relationship between policymakers and CSOs is missing, decision-makers may have minimal knowledge about what PWUD need, resulting in a lack of adequate, inclusive policies, based on mutual understanding and real necessities.

It has long been shown that community monitoring can play an essential role in improving service delivery (1). Civil society is increasingly assuming the role of holding governments and donors accountable, among others, by engaging in independent monitoring and evaluation of services and programmes (2). In combination with advocacy, monitoring tools are crucial strategies to hold governments accountable and to improve the implementation of policies and programmes in line with the needs of PWUD and their environments (3).

A complementary role

Other agencies already have well-established monitoring activities in the field of drug use and harm reduction, both globally and in Europe. At a global level, Harm Reduction International (HRI) has conducted a biannual survey since 2008, publishing its data in the report The Global State of Harm Reduction (4). Data collection involves a coordinated effort across practitioners, academics, advocates and activists, and provides an independent analysis of the state of harm reduction in the world. In Europe, the European Monitoring Centre for Drugs and Drug Addiction (EMCDDA) has conducted systematic monitoring since 2007. Harm reduction data is collected by its 30 National Focal Points (Reitox NFP Network), that are often lead by research or administrative institutions. These include all European Union (EU) countries plus Norway and Turkey. Core publications include the annual European Drug Report (5) and the Health and social responses to drug problems: a European guide (6).

Despite these excellent efforts, there are still gaps in information about harm reduction and the needs of PWUD. In some cases, data might not systematically reflect the perspective of harm reduction CSOs and their service users on availability, accessibility and quality of harm reduction interventions and ways of improvements. In others, it reflects civil society perspectives but only offers a generic overview of the European region, without much detail on policy implementation and experiences at the service delivery level in each country. These are the gaps that the C-EHRN intends to fill, complementing the work undertaken by others. This monitoring tries to reflect the 'street experiences' of harm reduction service providers and their service users, focusing on how drug policies and specific harm reduction guidelines are (or are not) being implemented at the street level. Such in-depth and rich information is crucial for the development of policies and services for PWUD, and can be of great value for CSO advocacy and for policymakers.



Methodology

C-EHRN has established four expert groups to support the development of the monitoring framework, draft the questionnaires, assess the data, and review the final report: A scientific expert group (SEG) and three thematic expert groups for Hepatitis C (HCV), overdose prevention (OD), and new drug trends (NDT)². These groups, together with C-EHRN staff, have contributed to the development of the framework of C-EHRN monitoring and have added to the formulation of the questionnaires. To gather data on the experiences of harm reduction service providers and service users at ground level, C-EHRN builds on a network of national Focal Points (FPs).

C-EHRN Focal Points

The Focal Points are organisational members of C-EHRN selected by:

- Their willingness to commit to the network's principles, mission and vision at the national and European level;
- Proven thematic expertise in the field of drug use and harm reduction;
- Connectedness at the national and European level; and,
- Ability to fulfil the role of an intermediary at a national level.

C-EHRN strives to select at least one FP per country, but some countries can have more than one representative if additional thematic expertise is needed, or no FP when no member is available for such a role. C-EHRN currently has 35 FPs in 34 countries³. Map 1 shows the location of the FPs partaking in the 2020 monitoring.

Map 1: Location of C-EHRN Focal Points



FP's cities

The tasks of FPs include being consulted for specific thematic or regional expertise; providing inputs and information, particularly for monitoring purposes, including answering the monitoring questionnaire annually. FPs do not receive financial support to perform their functions. Nevertheless, they receive a few benefits, such as being invited to the annual C-EHRN conference (one scholarship available per country); free C-EHRN seminars and training; being able to promote their activities on the network's website and through the network's other communication channels, and in speaking on behalf of the network at the national level.

Profile of FPs

More than 70% of C-EHRN FPs⁴ have as their main priority of their organisation the provision of services, making them highly appropriate in describing how harm reduction activities play out in practice. That is followed by advocacy and policy activities (17%), training and capacity building (10%), and, to a much lesser extent, research (2,5%) (see Figure 1).

Figure 1: Priorities (1 to 4) of FPs organisations.



The populations to which FP organisations are able to provide services can be seen in Figures 2 and 3. The main services provided (offered by more than 50% of FPs) are outreach work, HCV and HIV prevention, testing and treatment, drop-in centres, needle and syringe exchange, STI prevention, and legal support. Less than 15% of FPs provide housing or shelter, Heroin Assisted Treatment (HAT) or Drug Consumption Rooms (DCRs).

Figure 2: Populations to which FP organisations provide services (related to drug of choice)



PW: people who NPS: new psychoactive substance(s)





Even though research is not a priority for the vast majority of C-EHRN FPs, all of them are involved in some type of research activity. Besides C-EHRN monitoring, 83% of FPs are involved in data collection for monitoring and evaluating within their own organisations, 53% perform needs assessments, and 52% the monitoring of drug trends; more than 80% use the data collected for advocacy purposes. Virtually all FPs are involved in some kind of



policy and advocacy activity, moslty at the local/ regional or national level. Figure 4 shows the main research activities of FPs.



Figure 4: Main activities for those undertaking research

A new questionnaire in 2020

Based on feedback from both FPs and SEG members, the 2020 questionnaire was reformulated and shortened. In comparison to 2019, more questions now focus on the implementation (local) level, and on the experiences of FPs and their clients rather than on the national level. This was done to: 1) improve the reliability of data, since the experiences and knowledge of most FPs focuses on the local/regional level; and, 2) better address the main objective of this monitoring tool, which is to reflect fundamental qualitative data at the service delivery level.

Also in 2020, new sections were included on basic harm reduction services and on the harm reduction response to the COVID-19 pandemic.

A short questionnaire for PWUD

In 2020, on an experimental basis, a parallel short questionnaire was sent to PWUD in partnership with the European Network of People Who Use Drugs (EuroNPUD). The questions addressed access to harm reduction services, involvement in planning, monitoring and delivery of harm reduction services, and influence of the COVID-19 pandemic in life and access to harm reduction services.

Data gathering and analysis

FPs gathered data for this report based on a questionnaire distributed to them both as an online survey and as a PDF attached to an e-mail. The PDF was intended as a working document to be shared with contributors to the data gathering. The questionnaire is available at the C-EHRN website⁵. The short questionnaire for PWUD was sent via e-mail to contacts selected by EuroNPUD. From those answering, only feedback from respondents living in the same cities as FPs were retained; those were, in general, 1-2 respondents per city.

Closed questions were analysed for general percentages or represented in tables with descriptions of features per city/country. Open ended responses were analysed with thematic analysis (7) and key findings illustrated with quotes. Data were verified and analysed by the report authors. The different chapters were revised by the respective thematic expert groups.

Limitations

C-EHRN Monitoring is still in an early developmental phase; 2020 is only the second year of reporting. Following the feedback of the expert groups and the FPs, several adjustments were made to the 2020 questionnaire as compared to 2019. In addition, the influence of the COVID-19 pandemic might have influenced results. For this reason, in many cases it is still not possible to trace a comparable line as expected in a monitoring process. Both expert groups and FPs are still adjusting, and trying to find the best and more feasible indicators, as well as better ways to collect reliable and consistent data.

Given the nature of this monitoring structure and the focus of the work of C-EHRN FP organisations, data in this report cannot claim to be representative of Europe or the nations in which FPs are based. Most FPs work locally, or regionally, and have an in-depth knowledge of how harm reduction plays out in the streets. Respecting this experience was chosen over national representativeness, to provide a more nuanced analysis of the implementation of harm reduction at the local level. If, on the one hand, the monitoring loses in its ability to reflect a broader European situation focusing on developments at the national level, it gains in reflecting fundamental qualitative data at the service delivery level that can only be collected by CSOs, and which is lacking in several national/global reports. Other specific limitations can be found in the respective chapters.

A more complete account of the methodology, challenges and lessons learned with the C-EHRN monitoring can be found elsewhere (8).

Report structure

The report consists of 8 chapters.

This first introductory chapter provides information on the importance of CSOs in monitoring harm reduction; the methodology used for the present monitoring and the differences regarding the first monitoring report in 2019; the profile of the C-EH-RN Focal Points (FPs) collecting data for this report; and the limitations of this monitoring.

Chapter 2 is a special chapter written by HRI. It contextualises the state of harm reduction in Europe in comparison to other world regions, based on the data reported in the *Global State of Harm Reduction 2020* (4). The other chapters are based on data collected by C-EHRN FPs in 2020.

Chapter 3 reports data about civil society involvement in drug policy and related decision-making processes in European countries.

Chapter 4 is a newly introduced chapter, describing the state of essential harm reduction services in FP cities.

Chapter 5 describes experiences with the availability and accessibility of interventions that constitute the continuum of care for hepatitis C.

Chapter 6 describes the status of, the need for, and changes to, overdose prevention in the previous year at the local level in Europe.

Chapter 7 focuses on the perceived New Drug Trends in FP cities. Finally,

Chapter 8 discusses how the COVID-19 pandemic has affected harm reduction services and the lives of people who use drugs in different European cities.

References

- Bjorkman M, Svensson J. Power to the People: Evidence from a Randomized Field Experiment of a Community-Based Monitoring Project in Uganda. Q J Econ. 2009;124:735–769.
- Gildemyn M. Understanding the Influence of Independent Civil Society Monitoring and Evaluation at the District Level: A Case Study of Ghana. Am J Eval [Internet]. 2014 Mar 4;35(4):507–24. Available from: https://doi. org/10.1177/1098214014525257
- Joshi A, Houtzager PP. Widgets or Watchdogs? Public Manag Rev [Internet]. 2012 Feb 1;14(2):145–62. Available from: https://doi.org/10. 1080/14719037.2012.657837
- Harm Reduction International. The Global State of Harm Reduction 2020 [Internet]. London; 2020. Available from: https://www.hri.global/ files/2020/10/26/Global_State_HRI_2020_BOOK_ FA.pdf
- 5. European Monitoring Centre for Drugs and Drug Addiction. European Drug Report 2020: Trends and Developments. Luxembourg: Publications Office of the European Union; 2020.
- 6. EMCDDA. Health and social responses to drug problems: a European guide. Luxembourg; 2017.
- Braun V, Clarke V, Hayfield N, Terry G. Thematic Analysis. In: Liamputtong P, editor. Handbook of Research Methods in Health Social Sciences. Singapore: Springer International Publishing; 2019.
- 8. Rigoni R, Tammi T, van der Gouwe D, Schatz E. Harm reduction in Europe: a framework for civil society-led monitoring. Harm Reduct J. TBC.

CONTEXTUALISING HARM REDUCTION IN EUROPE



Harm Reduction International's report, the Global State of Harm Reduction 2020⁶, shows that implementation of vital harm reduction services has worsened globally since 2018, after having stalled since 2014. The number of countries where needle and syringe programmes (NSPs) are available remained level at 86, and the number of countries where opioid agonist therapy (OAT) is available decreased by two to 84.^[1] Europe⁷ represents one of the regions with the greatest number of harm reduction service available: almost half of the countries worldwide where NSP and OAT are available are in Europe, and ten out of twelve countries with officially sanctioned drug consumption rooms (DCRs) are European.

However, geographic gaps, and an uneven distribution of services, exist even in countries where harm reduction has been available for decades; for example, rural communities are particularly underserved in many countries in Europe. In addition to the geographic gaps in coverage, there are sub-groups of people who use drugs that experience barriers in Europe, including women who use drugs, men who have sex with men, people who use stimulants, or non-injecting methods of drug use, and people experiencing homelessness. Though these barriers in access to harm reduction services exist in Europe, and harm reduction coverage and funding is far from sufficient, there is no other region in the world where more than ninety percent of the countries have at least one NSP or OAT site, and more than ninety percent of the countries reference harm reduction in their national drug policies.

While there are still serious issues in harm reduction implementation in European countries, even in countries pioneering harm reduction, this region is in a privileged position compared to Asia, Latin America and the Caribbean, the Middle East and North Africa (MENA), and sub-Saharan Africa, where unfavourable drug policy environments limit harm reduction service implementation and some governments adopt punitive drug strategies that pose serious threats to the rights of people who use drugs.

Table 1: Global availability of harm reduction services by region	

Global State of Harm Reduction region (number of countries in the region)	Explicit sup- portive refer- ence to harm reduction in national policy docu- ments	At least one needle and syringe programme operational	At least one opioid ago- nist therapy programme operational	At least one drug con- sumption room	Peer distribution of naloxone	OAT in at least one prison	NSP in at least one prison
Asia (25)	11	14	15	0	4	5	0
Eurasia (29)	26	27	26	0	2	21	5
Latin America and the Caribbean (24)	9	4	4	0	2	0	0
Middle East and North Africa (19)	7	7	5	0	0	6	0
North America (2)	2	2	2	1	2	2	1
Oceania (12)	2	2	2	1	1	2	0
Sub-Saharan Africa (38)	11	10	9	0	0	3	0
Western Europe (24)	20	20	21	10	4	20	4
Total	88	86	84	12	15	59	10



Table 2: European availability⁸ of harm reduction services in practice and policy

Country or territory	Explicit supportive reference to harm reduc- tion in na- tional policy documents	At least one needle and syringe programme operational	At least one opioid ago- nist therapy programme operational	At least one drug con- sumption room	Peer distribu- tion of nalox- one	OAT in at least one prison	NSP in at least one prison
Albania	1	1	1	×	×	1	×
Bosnia and Herzegovina	1	1	1	×	×	1	×
Bulgaria	1	×	1	×	×	1	×
Croatia	1	1	1	×	×	1	×
Czechia	1	1	1	×	×	1	×
Estonia	1	1	1	×	1	1	×
Georgia	1	1	1	×	×	1	×
Hungary	1	1	1	×	×	1	×
Kosovo	1	1	1	×	×	×	×
Latvia	1	1	1	×	×	1	×
Lithuania	1	1	1	×	×	1	×
North Macedonia	1	1	1	×	×	1	1
Moldova	1	1	1	×	×	1	1
Montenegro	1	1	1	×	×	1	×
Poland	1	1	1	×	×	1	×
Romania	1	1	1	×	×	1	x
Russia	×	1	×	×	×	x	×
Serbia	1	1	1	×	×	1	×
Slovakia	1	1	1	×	×	x	x
Slovenia	1	1	1	×	×	1	x
Ukraine	1	1	1	×	1	1	×
Andorra	nk	nk	nk	×	nk	nk	nk
Austria	1	1	1	×	×	1	×
Belgium	1	1	1	1	×	1	×
Cyprus	1	1	1	×	×	1	x
Denmark	1	1	1	1	1	1	×
Finland	1	1	1	×	×	1	x
France	1	1	1	1	x	1	×
Germany	1	1	1	1	×	1	1
Greece	1	1	1	×	×	1	x
Iceland	1	1	1	×	×	1	x
Ireland	1	1	1	×	x	1	×
Italy	1	1	1	×	1	1	x
Luxembourg	1	1	1	1	×	1	1
Malta	1	1	1	×	×	1	×
Monaco	nk	nk	nk	×	nk	nk	nk

Country or territory	Explicit supportive reference to harm reduc- tion in na- tional policy documents	At least one needle and syringe programme operational	At least one opioid ago- nist therapy programme operational	At least one drug con- sumption room	Peer distribu- tion of nalox- one	OAT in at least one prison	NSP in at least one prison
Netherlands	1	1	1	1	×	1	×
Norway	1	1	1	1	1	1	×
Portugal	1	1	1	1	×	1	×
San Marino	nk	nk	nk	×	nk	nk	nk
Spain	1	1	1	1	×	1	1
Sweden	1	1	1	×	×	1	×
Switzerland	1	1	1	1	×	1	1
United Kingdom	1	1	1	×	1	1	×
TOTAL	40	40	40	10	6	38	6
Percentage of implement- ing countries worldwide	45%	47%	48%	83%	40%	64%	60%
Percentage of European countries	91%	91%	91%	23%	14%	86%	14%

Needle and syringe programme implementation

Needle and syringe programmes are the most widely implemented harm reduction service globally, and, as of 2020, eighty-six countries implemented such services overall. However, there is a large disparity between world regions in NSP implementation, both in terms of availability and coverage. Of 206 countries worldwide, 179 report some injecting drug use.^[2] In Eurasia, North America, Oceania, and Western Europe almost all countries with reported injecting drug use implement NSPs, but it is severely lacking in many countries in other regions including Asia, MENA and sub-Saharan Africa.

There are large differences among countries with NSPs in terms of coverage and accessibility. NSPs in Australia distributed 698 syringes per person who injects drugs (per year) in 2019,^[3] whereas in Benin, in sub-Saharan Africa, NSPs only provide ten syringes per person visiting the programme per month⁹. In India, despite the increase in the number of NSPs in the country, the number of syringes distributed per person is just 35, even though new estimates

suggest that injecting drug use is more prevalent in the country than previously documented.^[5] Gaps in geographic distribution of services is a problem across all regions. Remote areas and rural populations are underserved even in high-income countries such as the United States, Canada, Australia, New Zealand and many countries throughout Western Europe. Another cross-regional issue is NSP provision for people who use stimulants. The needs of this population are generally not met despite the fact that stimulant injecting has been associated with local HIV outbreaks in four countries (Ireland, Germany, Luxembourg, and the United Kingdom) in Western Europe in the past five years.^[6–8]

Stigma and discrimination against people who inject drugs continues to exist and hinder service access in all contexts and all regions,^[9-11] and the lack of appropriate, gender-specific programmes for women who use drugs is a serious issue. Moreover, black, brown, and indigenous people experience serious additional barriers and discrimination, for example in North America and Oceania. Migrants who inject drugs face similar problems in access to harm reduction services in Western Europe.



Provision of opioid agonist therapy

Methadone is the most frequently prescribed OAT medication worldwide, followed by buprenorphine or buprenorphine-naloxone. Heroin assisted treatment is the least available option, being available only¹⁰ in Canada, Denmark, Germany, Luxembourg, the Netherlands, Switzerland and the United Kingdom. While OAT is available in nearly all countries in Europe, OAT provision is insufficient in many other regions. For example, in sub-Saharan Africa, OAT is available in eight out of 49 countries with reported presence of injecting drug use in the region. OAT remains unavailable in Zimbabwe and Nigeria, despite significant populations of people who inject opioids and high HIV prevalence in both countries. In Latin America and the Caribbean, OAT is only available in Argentina, Colombia, Mexico and Puerto Rico; however, this is partly because in Latin America and the Caribbean, opioid use is relatively uncommon, and the region has the lowest proportion of opioid users among treatment admissions worldwide.^[12]

Despite availability, significant barriers exist in the accessibility of OAT for certain communities. Women, transgender people, indigenous people, and people experiencing homelessness all face significant issues related to access in all regions. Cost is also a serious barrier in access to OAT in many countries; practices in implementation generate a serious financial burden for clients, such as significant out-of-pocket expenses in some cases in Australia, Lebanon and Mexico. Furthermore, geographic distribution of services is uneven, with particularly insufficient coverage in rural areas, a serious issue in every country.

Overdose prevention and drug consumption rooms

Globally, officially sanctioned drug consumption rooms (DCRs) are available only in Australia, Belgium, Canada, Denmark, France, Germany, Luxembourg, Netherlands, Norway, Portugal, Spain and Switzerland. The number of countries where DCRs are implemented has increased since 2018, with Portugal opening a mobile service in 2019.^[13] Canada has the highest number of DCRs in the world with 39; in addition, a further 20 overdose prevention sites, which are primarily volunteer-run and funded, have also been opened in the country. There are two DCRs in Australia, the second facility opening in 2018 in Melbourne, and an independent review of the first 18 months of its operation found that it was beneficial not just in reducing harms, but in providing access to health and support services.^[14] In Western Europe, DCRs increasingly include supervised inhalation spaces to adapt to the needs of people who smoke drugs and the decline in injecting;^[15] for example, in the Netherlands, all 24 DCRs in the country allow smoking, while injecting is allowed in only 19.^[16]

Naloxone provision to those who are at risk of an opioid overdose, or who might witness an overdose, is an effective means of preventing overdose deaths.^[17,18] However, restrictive legal environments (for example where naloxone is available only in medical, emergency or treatment settings) hinders the implementation of naloxone distribution programmes across the world. Asia has only four countries where some form of naloxone distribution is available (Afghanistan, India, Myanmar and Vietnam). In the MENA region, Iran is the only country where naloxone is available outside of medical settings, and take-home naloxone is implemented; and in sub-Saharan Africa, naloxone remains largely unavailable or difficult to access. Naloxone remains highly limited in Latin America and the Caribbean, even in areas where opioid use is prevalent. However, there is a peer distribution network of naloxone in northern Mexico, and naloxone became available in Puerto Rico after a long advocacy campaign by civil society. Though naloxone is available in several countries of Eurasia, Ukraine is the only country in the region where naloxone is available without prescription.

The number of countries found to facilitate peer distribution of naloxone, whereby individuals can pass on naloxone without each recipient requiring a personal prescription, increased from 12 in 2018 to 15 countries¹¹ globally. Six of those are in Europe: Denmark, Estonia¹², Italy, Norway, Ukraine and the United Kingdom.

Viral hepatitis, tuberculosis and HIV

According to the latest report from the United Nations Office on Drugs and Crime (UNODC), an estimated 11.3 million people inject drugs globally, while HIV prevalence is estimated to be 12.6% and hepatitis C prevalence 48.5% among this population.^[19] People who inject drugs are particularly vulnerable to HIV and viral hepatitis, but other groups, such as people who smoke opioids or stimulants, are also at greater risk than the general population.^[20,21] using STI diagnosis history as an indicator. A cross-sectional study was conducted in 323 NIDUs of two facilities for alcohol and/or drug dependence treatment in the Goiás State, Central Brazil. All participants were interviewed about risk behaviors and STI history. Multivariable analysis was performed in order to identify predictors of STIs. Adjusted prevalence ratio (APR People who use drugs are disproportionately represented among tuberculosis (TB) cases and are at greater risk of developing more serious TB,^[22] while people living with HIV who inject drugs are two to six times more likely to develop TB than the general population.[23-25]

The early implementation of harm reduction approaches (such as NSPs and OAT), and the sustained harm reduction response is credited with maintaining low prevalence of HIV among people who inject drugs in Australia, New Zealand and Switzerland, among others.[26-28] On the other hand, one of the most reported barriers to HIV and hepatitis C testing and treatment is services implemented in settings that are not appropriate to the needs of key populations. Treatment and care among people who use drugs has focused mainly on the needs of people who inject opioids. In Latin America, data shows that use of stimulant drugs has also been associated with higher risk of HIV transmission through unsafe sexual behaviours. ^[29,30] Furthermore, the sharing of pipes and higher-risk sexual practices among people who use stimulants are associated with increased hepatitis C infection.[12,31,32]

Stigma and discrimination towards people who use drugs, as well as unstable housing, poverty, criminalisation and incarceration, continue to

act as major barriers to people accessing testing and treatment in every region. Additionally, there are still barriers to hepatitis C treatment for those actively using drugs, despite evidence showing strong treatment benefit with current treatment regimens for such patients.^[33,34] Scaling up of, and access to, harm reduction interventions (not just the aforementioned NSP, OAT, naloxone distribution, but also community-based testing and treatment) are included among key measures in decreasing the prevalence of HIV and hepatitis in international and regional guidelines.[35-38] Community-led programmes are an effective way to reduce the barriers to diagnosis and treatment for key populations beside people who use drugs, such as transgender people and people experiencing homelessness.

Harm reduction during the COVID-19 pandemic

The pandemic showed that harm reduction services are essential public health interventions, crucial in reaching key populations. Networks of people who use drugs also played an important role during the pandemic, developing guidelines for avoiding COVID-19 and harm reduction tips for people who use drugs.^[39,40] Peers played a crucial role beyond advocacy during the pandemic, showing that greater community involvement is fundamental to increase accessibility of services. They contributed to service delivery and in filling the gap in service provision with peer-to-peer needle and syringe distribution, providing input for other services working on-the-ground, and disseminating crucial information among the drug user community.

People who use drugs, especially people who smoke or inject drugs, face additional risks and vulnerabilities to COVID-19¹³ infection compared to the general population.^[42-44] Furthermore, people who use drugs may be less able or willing to adhere to quarantine and physical distancing measures in general, since they may need to seek out harm reduction services like NSPs and OAT programmes, or need to procure drugs to



avoid withdrawal symptoms.^[42] Therefore, maintaining services for people who use drugs is even more vital during a public health crisis such as the COVID-19 pandemic. All the more so, since interruption of harm reduction services – closures, staffing restrictions, decreasing coverage, or reducing funds – can lead to a spike in HIV and hepatitis C infections.^[45]

Although the pandemic seriously affected service delivery and the coverage of harm reduction services throughout the globe, in North America, Oceania and Western Europe, the impact was less severe compared to other regions. For example, the majority of European Union countries reported only a slight decrease or no change in availability of harm reduction services.^[46] In Asia however, key populations experienced delays in accessing HIV and harm reduction services.^[47] In Latin America, contact with harm reduction programmes has been limited due to physical distancing, and in MENA, harm reduction services had to reduce the number of working days, or close entirely, in all countries in the region. In sub-Saharan Africa, OAT services were entirely suspended in some countries during COVID-19 and take-home OAT is rarely available in the region.

There were also positive changes in service delivery during the pandemic. Most importantly, longer take-home periods for OAT, and less restrictive initiation procedures, were set up in many countries, providing evidence that these are feasible and beneficial. Out of the 84 countries where OAT is available, 47 countries (with at least one country in every region) expanded take-home periods for OAT medications and 23 made distribution more accessible with home delivery or OAT distribution in outreach services.^[48] Nine countries expanded induction practices, including facilitated or rapid initiation.^[48] Innovative measures were introduced to compensate for decreased availability; for example, online consultations replaced some faceto-face meetings in the MENA region, and service providers set up online shops for injecting equipment in the United Kingdom and New Zealand. These COVID-19 adaptations in OAT, NSP and treatment delivery can increase access to services and should remain in place.

References

- HRI. Global State of Harm Reduction 2020. London: Harm Reduction International; 2020. https:// www.hri.global/files/2020/10/26/Global_State_ HRI_2020_BOOK_FA.pdf
- Degenhardt L, Peacock A, Colledge S, Leung J, Grebely J, Vickerman P, et al. Global prevalence of injecting drug use and sociodemographic characteristics and prevalence of HIV, HBV, and HCV in people who inject drugs: a multistage systematic review. The Lancet Global Health 2017;5(12):e1192–207. http://dx.doi.org/10.1016/ S2214-109X(17)30375-3
- Heard S, Iversen J, Kwon JA, Maher L. Needle Syringe Program National Minimum Data Collection

 National Data Report 2019. Sydney: Kirby Institute, UNSW Sydney; 2019. https://kirby.unsw.edu. au/sites/default/files/kirby/report/NSP-NMDC_Report-2019.pdf
- WHO. Global health sector strategy on viral hepatitis 2016-2021. Towards ending viral hepatitis. Geneva: World Health Organization; June 2016. http://apps.who.int/iris/bitstream/10665/246177/1/WHO-HIV-2016.06-eng. pdf
- Ambekar A, Chadda RK, Khandelwal SK, Rao R, Mishra AK, Agrawal A. Magnitude of Substance Use in India. New Delhi: National Drug Dependence Treatment Centre (NDDTC), All India Institute of Medical Sciences (AIIMS); 2019. http:// www.ndusindia.in/downloads/Magnitude_India_ FINAL_REPORT.pdf
- Sypsa V. Why do HIV outbreaks re-emerge among people who inject drugs? The Lancet HIV 2019;6(5):e274–5. https://doi.org/10.1016/S2352-3018(19)30079-7
- ECDC, WHO Regional Office for Europe. HIV/AIDS surveillance in Europe 2019 - 2018 data. Stockholm: ECDC; 2019. https://www.ecdc.europa.eu/ sites/default/files/documents/hiv-surveillance-report-2019.pdf
- EMCDDA. Drug-related infectious diseases in Europe. Update from the EMCDDA expert network, 2020, Technical report. Luxembourg: Publications Office of the European Union; 2020. https://www.emcdda.europa.eu/system/files/publications/13091/Technical-report_DRID2020.pdf

- Meyerson BE, Lawrence CA, Cope SD, Levin S, Thomas C, Eldridge LA, et al. I could take the judgment if you could just provide the service: non-prescription syringe purchase experience at Arizona pharmacies, 2018. Harm Reduction Journal 2019;16(1):57. https://doi.org/10.1186/s12954-019-0327-1
- Davis SM, Kristjansson AL, Davidov D, Zullig K, Baus A, Fisher M. Barriers to using new needles encountered by rural Appalachian people who inject drugs: implications for needle exchange. Harm Reduction Journal 2019;16(1):23. https:// doi.org/10.1186/s12954-019-0295-5
- Carruthers S. Needle and Syringe Programs in Australia: Peer-led Best Practice. Canberra.: Australian Injecting and Illicit Drug Users League; 2018. http://files.idpc.net/library/AIVL_needle_ and_syringe_report_final%20%281%29.pdf
- 12. UNODC. World Drug Report 2020. Booklet 2. Drug Use and Health Consequences . 2020 [cited 2020 Jun 6]; https://wdr.unodc.org/wdr2020/
- Taylor H, Curado A, Tavares J, Oliveira M, Gautier D, Maria JS. Prospective client survey and participatory process ahead of opening a mobile drug consumption room in Lisbon. Harm Reduction Journal 2019;16(1):49. https://doi.org/10.1186/ s12954-019-0319-1
- Medically Supervised Injecting Room Review Panel. Review of the Medically Supervised Injecting Room. Melbourne: Victorian Government; 25 June 2020. https://www2.health.vic.gov.au/Api/ downloadmedia/%7B52D63022-19E8-4347-9170-1ACDA991D926%7D
- EMCDDA. Drug consumption rooms: an overview of provision and evidence (Perspectives on drugs). Lisbon: EMCDDA; 2018. https://www.emcdda.europa.eu/system/files/publications/2734/ POD_Drug%20consumption%20rooms.pdf
- de Gee A, van der Gouwe D, Woods S, Charvet C, van der Poel A. Drug Consumption Rooms in the Netherlands. 2018 Update. Utrecht: Trimbos-instituut; 2019. https://www.trimbos.nl/docs/ eebe7cf1-179d-407b-94d0-8a202d8ec296.pdf
- EMCDDA. Health and social responses to drug problems: a European guide. Luxembourg: Publications Office of the European Union; October 2017. https://www.emcdda.europa. eu/system/files/publications/6343/TI_PUBPDF_ TD0117699ENN_PDFWEB_20171009153649.pdf



- WHO. Community management of opioid overdose. Geneva: WHO; 2014. https://apps.who.int/ iris/rest/bitstreams/611762/retrieve
- UNODC. World Drug Report 2020 Estimates of people who inject drugs, living with HIV, HCV & HBV, downloadable spreadsheet. Vienna: UN-ODC; 2020. https://wdr.unodc.org/wdr2020/en/ maps-and-tables.html
- Guimarães R, Monteiro L, Teles S. Risk behaviors for sexually transmitted infections in noninjecting drug users: A cross-sectional study. International Journal of STD & AIDS 29:658–64. 2018 Jun;29(7):658-664. doi: 10.1177/0956462417750332.
- Duran A, Rossi D, et al. High acceptability of rapid HIV test in Argentina. Experience during a seroprevalence study in vulnerable groups. 8th IAS Conference on HIV Pathogenesis, Treatment and Prevention; 2015; https://www.conicet.gov.ar/ new_scp/detalle.php?keywords=&id=24491&inst=yes&congresos=yes&detalles=yes&congr_ id=5470099
- 22. Langer AJ, Navin TR, Winston CA, LoBue P. Epidemiology of Tuberculosis in the United States. Clin Chest Med 2019;40(4):693–702. doi: 10.1016/j. ccm.2019.07.001
- 23. Deiss RG, Rodwell TC, Garfein RS. Tuberculosis and drug use: review and update. Clin Infect Dis 2009;48(1). doi: 10.1086/594126
- Friedland G. Infectious disease comorbidities adversely affecting substance users with HIV: hepatitis C and tuberculosis. J Acquir Immune Defic Syndr 2010;55 Suppl 1:S37–42. DOI: 10.1097/ QAI.0b013e3181f9c0b6
- Taarnhøj GA, Engsig FN, Ravn P, Johansen IS, Larsen CS, Røge B, et al. Incidence, risk factors and mortality of tuberculosis in Danish HIV patients 1995–2007. BMC Pulm Med 2011;11:26. doi: 10.1186/1471-2466-11-26
- 26. Saxton PJW, McAllister SM, Noller GE, Newcombe DAL, Leafe KA. Injecting drug use among gay and bisexual men in New Zealand: Findings from national human immunodeficiency virus epidemiological and behavioural surveillance. Drug and Alcohol Review 2020;39(4):365–74. doi: 10.1111/dar.13046
- Iversen J, Wand H, Topp L, Kaldor J, Maher L. Extremely low and sustained HIV incidence among people who inject drugs in a setting of harm reduction. AIDS 2014;(28):275–83. doi: 10.1097/ QAD.00000000000068

- Marzel A, Kusejko K, Weber R, Bruggmann P, Rauch A, Roth JA, et al. The Cumulative Impact of Harm Reduction on the Swiss HIV Epidemic: Cohort Study, Mathematical Model, and Phylogenetic Analysis. Open Forum Infect Dis 2018 [cited 2018 Jul 26];5(5). https://doi.org/10.1093/ ofid/ofy078
- 29. UNODC. HIV Prevention, Treatment, Care and Support for People Who Use Stimulant Drugs. 2019; https://www.unodc.org/documents/ hiv-aids/publications/People_who_use_ drugs/19-04568_HIV_Prevention_Guide_ebook. pdf
- UNODC. Risk and Transmission of HIV, HCV & HBV among stimulant drugs users. A review of the evidence (A). Part 1/5. Methodology and Summary. Vienna; UNODC, 2017. https://www.unodc.org/ documents/hiv-aids/2017/1_Stim_HIV_Syst_Lit_ rev_Part_1_methodology_and_summary.pdf
- Cortés E, Metaal P. Mercados de cocaína fumable en América Latina y el Caribe. Llamamiento a favor de una respuesta sostenible en materia de políticas. 2019.https://www.tni.org/ files/publication-downloads/tni-smokablecocaine_sp_web-def.pdf
- 32. CICAD. Análisis de seguimiento de usuarios de cocaínas fumables en programas de atención y tratamiento a dos años del ingreso. 2018. http://www.cicad.oas.org/oid/pubs/AnalisisdeSeguimento%20(1).pdf
- 33. EASL. EASL Policy Statement: Drug use and the global hepatitis C elimination goal . EASL; 2020. https://easl.eu/wp-content/uploads/2020/08/fullversion-easl-policy-statement-on-drug-use-andthe-global-hepatitis-c-elimination-goal.pdf
- 34. Grebely J, Drolet M, Nwankwo C, Torrens M, Kastelic A, Walcher S, et al. Perceptions and self-reported competency related to testing, management and treatment of hepatitis C virus infection among physicians prescribing opioid agonist treatment: The C-SCOPE study. Int J Drug Policy 2019;63:29–38. doi: 10.1016/j.drugpo.2018.10.012
- 35. WHO. Consolidated guidelines on HIV prevention, diagnosis, treatment and care for key populations - 2016 update. Geneva: WHO; 2016. http://apps.who.int/iris/bitstream/10665/246200/1 /9789241511124-eng.pdf?ua=1
- 36. WHO. Guidelines for the care and treatment of persons diagnosed with chronic hepatitis C virus infection. Geneva: WHO; 2018. http://apps.who. int/iris/bitstream/handle/10665/273174/978924155 0345-eng.pdf

- WHO. Combating Hepatitis B and C to Reach Elimination by 2030. Geneva: World Health Organization; May 2016. http://apps.who.int/iris/bitstream/10665/206453/1/WHO_HIV_2016.04_eng. pdf
- ECDC, EMCDDA. Prevention and control of infectious diseases among people who inject drugs. Stockholm, Sweden: European Centre for Disease Prevention and Control : European Monitoring Centre for Drugs and Drug Addiction; October 2011. https://www.emcdda.europa. eu/system/files/publications/638/ECDC-EMCD-DA_IDU_guidance_-_web_version_328027.pdf
- 39. INPUD. COVID-19 Crisis: Harm Reduction Resources for People who Use Drugs. London: International Network of People who Use Drugs. 2020 https://www.inpud.net/en/covid-19-crisis-harmreduction-resources-people-who-use-drugs
- 40. INPUD. COVID 19: Advice for People who Use Drugs leaflet. London: International Network of People who Use Drugs. 2020 https://www.inpud. net/en/covid-19-advice-people-who-use-drugsleaflet
- WHO. Coronavirus. Geneva: World Health Organization. 2020 https://www.who.int/emergencies/ diseases/novel-coronavirus-2019
- Vasylyeva TI, Smyrnov P, Strathdee S, Friedman SR. Challenges posed by COVID-19 to people who inject drugs and lessons from other outbreaks. J Intern AIDS Soc J Int AIDS Soc. 2020 Jul; 23(7): e25583.. doi: 10.1002/jia2.25583
- 43. EMCDDA. European Drug Report 2020: Trends and Developments. Luxembourg: Publications Office of the European Union; 2020. https:// www.emcdda.europa.eu/system/files/publications/13236/TDAT20001ENN_web.pdf
- Harris M. An urgent impetus for action: safe inhalation interventions to reduce COVID-19 transmission and fatality risk among people who smoke crack cocaine in the United Kingdom. Int J Drug Policy 2020 Sep; 83: 102829. doi: 10.1016/j.drugpo.2020.102829
- 45. HRI. Making the investment case: Cost-effectiveness evidence for harm reduction . London: HRI; 2020. https://www.hri.global/files/2020/04/21/ HRI_Cost_Effectivenes_Briefing_(APRIL_2020).pdf
- 46. EMCDDA. EMCDDA trendspotter briefing -Impact of COVID-19 on drug services and help-seeking in Europe. May 2020 https://www. emcdda.europa.eu/publications/ad-hoc/

impact-of-covid-19-on-drug-services-and-helpseeking-in-europe_en

- 47. APCASO. Rapid survey on the needs of young key populations and young people living with HIV in Asia and the Pacific in the context of COVID-19. Bangkok, 18 May 2020; https://apcaso.org/assessing-the-needs-of-young-key-populations-during-covid-19-outbreak-in-asia-and-thepacific/
- HRI. COVID-19. In: Global State of Harm Reduction 2020. London: Harm Reduction International; 2020. https://www.hri.global/files/2020/10/26/ Global_State_HRI_2020_BOOK_FA.pdf



Participation of Civil Society Organisations in policymaking



Introduction

The importance of civil society involvement in policy making, including the development and implementation of drug policies, is widely recognized (1). In Europe, both the European Union and the Council of Europe (2) have recognised the essential need for the contribution made by CSOs to the development and realisation of democracy, securing the transparency and accountability of public authorities. Reports such as those produced by the Civil Society Involvement in Drug Policy Project¹⁴ (CSIDP) provide relevant information on what kind of resources, capacities and mechanisms are needed to ensure meaningful civil society involvement (3). A continuous challenge is to identify and monitor if, and how, such meaningful involvement is assured.

This chapter intends to contribute to that by addressing the civil society involvement in policymaking in the cities and countries of C-EHRN FPs. Cooperation between CSOs and policymakers is evaluated by FPs at national and local levels. As explained in the introduction, in 2020 a stronger focus was added at the local level to be able to be closer to the experience of CSOs and to bring more insights into how official mechanisms get (or not) implemented on the ground. For that, the indicators used are those proposed by the code of good practice for civil participation in the decision-making process of the Council of Europe (4) and the assessment for the meaningful involvement of civil society in the area of drug policy in Europe by the Civil Society Forum on Drugs (CSFD) (1). The first set of indicators, defining levels of cooperation, was also used in 2019; the second, defining indicators to evaluate the aim and nature of cooperation, are used for the first time in 2020. As compared to 2019, new indicators were also included in the 2020 report to map CSO contributions to data collection and reporting, and participation in organised networks and national platforms. Due to the several changes regarding indicators, it is still not possible in this report to draw comparisons between 2019 and 2020.

Cooperation between CSO's and policymakers

Cooperation mechanisms

The cooperation between policymakers and CSOs to debate and give input on drug policies can assume different forms. At one end of the spectrum, cooperation is restricted to information exchange. At the other end, a solid partnership is established (5).

Following the definition of cooperative mechanisms of the Council of Europe (4), four different levels of cooperation can be considered:

- Information: This is a relatively low level of cooperation. It consists of a two-way process of information sharing and the provision of access to it between public authorities and CSOs;
- Consultation: This is an ad hoc mechanism through which public authorities ask CSOs for their expertise and opinion regarding a specific policy issue or development;
- Dialogue: This is a two-way communication mechanism built on mutual interests and potentially shared objectives to ensure a regular exchange of views; and,
- Partnership: This is the most comprehensive type of cooperation. This mechanism stipulates and articulates shared responsibilities for each step of the policymaking process: agenda-setting, policy drafting, and implementation of activities.



Cooperation at the national level

The great majority of the FPs reported the existence of structural cooperation between CSOs and policymakers on drug policy in their countries. Only FPs from Finland, Sweden and the Russian Federation reported having no national cooperative mechanisms (see Figure 5).

Figure 5: Is there any structural exchange mechanism between policymakers and CSOs in the field of drug policy in your country, at the national level?



Types of national-level collaboration

Information and dialogue were the most cited types of collaboration (57% each). Half of the respondents referred to having a consultation level of cooperation between CSOs and government at the national level, and only 13% referred to partnership.

Figure 6: Types of national collaboration between CSOs and governments



Reasons for a lack of national-level cooperation

The main reason given for a lack of cooperation were intolerant drug policies pursued by the respective countries or, otherwise, the lack of importance given to harm reduction.

- 66 Harm reduction is seen as a niche-scene and is often disregarded by policymakers. There is no "systemic way" to influence policymakers (FP Helsinki, Finland).
- **C** Because of the restrictive drug policy in Russia. (FP Saint Petersburg, Russian Federation)
- Government has a strict zero tolerance policy against all kinds of drugs. (FP Stockholm, Sweden)

Map 2 portrays the highest type of collaboration mentioned at the country level by respondents in 2020.

Map 2: Highest type of collaboration - country level, 2020



FPs were asked to evaluate the exchange between government and CSOs from the perspective of its aims and nature. The indicators were developed by the thematic working group of CSFD, based on a literature review on the quality of civil society involvement that the group conducted in the summer of 2020 (1).

The aim of the exchange between government and CSOs

FPs were asked to indicate to what extent the following statements applied to their country context. The exchange between government and CSOs aims to:

- Inform Civil Society (CS) of new policy developments
- Collect input and knowledge from CS at the grassroots level to learn more about new developments, trends and problems
- Share developments, trends and problems from the field and the grass root level
- **Discuss** which kind of drug policies are effective, beneficial or harmful
- **Develop** new strategies and approaches
- Improve access to, and the quality of, services (health, social and drug-related services)

The 5-point scale for answering included the following options: strongly agree (1); agree (2); undecided (3); disagree (4); and, strongly disagree (5).

Figure 7 shows the overall results. Over 60% of FPs agree (or strongly agree) that the exchange between government and CSOs aims at informing CSOs on new policy developments and collecting their input on new developments, trends and problems. About half agree that the aim is to share information on such developments and needs. About 35-40% think that these exchanges aim at discussing policies, developing new strategies and approaches, and improving services. According to these findings, the majority of respondents view civil society involvement as a one-way information flow from the government to civil society, rather than an interactive, constructive exchange of ideas and views between decision makers and civil society.

Figure 7: How much do you agree with the following statements about the aim of exchanges between government and CSOs?



Table 3 shows the answers per FP. It must be born in mind that the evaluations as to the aim of the exchange between government and CSOs are subjective and, thus, not easily comparable between countries and cities. However, if comparisons are attempted, the best overall average (means of agreement for all 6 statements) includes Bern (Switzerland), Barcelona (Spain), Nicosia (Cyprus), and Amsterdam (Netherlands). Again, in comparison, the overall situation was estimated to be worst in Helsinki, London and Bratislava.



country. The exchange between government and CS aims to:									
			Collect		.		Improve		
Country	City	Inform	input	Share	Discuss	Develop	services		
Albania	Tirana	3	3	2	2	2	2		
Austria	Vienna								
Belgium	Antwerp	4	2	2	5	5			
Croatia	Rijeka	1	2	2	3	2			
Cyprus	Nicosia	2	2	2	2	2	1		
Czech Republic	Prague	2	2	2	4	4	5		
Denmark	Copenhagen	2	2	3	2	4	2		
Estonia	Tallinn	2	2	2	3	2	2		
Finland	Helsinki	5	4	5	5	4			
France	Paris	2	2	2	2	2	2		
Georgia	Tbilisi	2	3	3	5	3			
Germany	Berlin	2	2	3	3	2	3		
Greece	Athens	3	2	2	1	2	2		
Hungary	Budapest	4	4	4	4	4	3		
Ireland	Dublin	3	3	3	3	3			
Italy	Milano /Roma	4	3	4	4	4			
Lithuania	Vilnius	2	2	2	4	3	3		
Luxemburg	Luxemburg	2	2	2	2	2			
Norway	Kristiansand		2	2	2	2			
Poland	Krakow	4	2	2	4	4	-		
Portugal	Vila Nova de Gaia	4	4	4	5	4	3		
Republic of N.									
Macedonia	Skopje	2	2	2	2	3	3		
Romania	Bucharest	4	2	3	3	4			
Russia	Saint Petersburg								
Scotland, UK	Glasgow	2	3	3	3	3	2		
Serbia	Novi Sad	2	2	4	4	2			
Slovakia	Bratislava	2	4	4	5	5			
Slovenia	Ljubljana	2	4	4	4	4			
Spain	Barcelona	1	1	1	2	1			
Sweden	Stockholm	4	4	4	4	4			
Switzerland	Bern	4	4	4	4	4			
The Netherland	-	2	1	1	2	3			
		2	2	3	3	3			
Ukraine United Kingdom	Kyiv	5	4	4		4			

Table 3: How much do you agree with the following statements about the aim? (per city)

The nature of the exchange between government and CSOs

Another set of indictors produced by the working group of the Civil Society Forum on Drugs (1) aimsto investigate the nature of the exchange between governments and CSOs. FPs were asked to indicate to what extent they agree with the following statements about the exchange between government and CS in their country:

- It is organised in a transparent way (e.g. it is easy to follow the decision making process)
- It is organised in a **balanced** way (represents different services well, communities, worldviews).
- It is organised in a timely manner (e.g. CS is informed in a timely way about any kind of new policy/development and the agenda of the meeting).
- Government officials are easily approachable by CSOs (e.g. they respond to emails/ phone calls).
- Decision makers are represented at the appropriate level (e.g. those who make deicisions are involved).
- The government is **open** to civil society initiatives (e.g. civil society initiatives are easily taken up by government).
- Adequate **funding** is provided (e.g. there is public funding for advocacy work).
- Civil society input is **heard** and taken into account when it comes to decision making.
- Civil society can speak openly and frankly and criticise without facing repercussions or budget cuts.





Over half (between 52-59%) of FPs agree that civil society can criticise the government without facing repercussions or budget cuts, and that government officials are easily approachable by CSOs. Only about a third agree that decision makers are represented at the appropriate level and that the government is open to civil society initiatives. Even less (about 20%) think that exchanges between CSOs and the government are transparent, or that there is a balance between different perspectives. Over 60% disagree that adequate funding is provided.

Table 4 shows the responses from each FP. The 5-point scale for responses include the following options: strongly agree (1); agree (2); undecided (3); disagree (4); and, strongly disagree (5). Again, FP evaluations on the nature of the exchange between government and CSOs are subjective and, thus, cannot be easily compared between countries and cities. However, when a compar-


ison is attempted, the best overall average (the mean of agreement for all 9 criteria) was in Bern (Switzerland), Luxembourg, Barcelona (Spain), and Berlin (Germany). Again, in comparison, the overall situation was perceived to be worst in Helsinki, London and Bratislava.

Table 4: How much do you agree with the following statements about the nature of the exchange between government and CS? (per city)

How muc	h do you agree w	ith the followi	ng stateme	nts abou	t the exchange b	etween g	overnme	nt and C	SOs in yo	
		_					_			No
Country	City	Transparent	Balanced	Timely	Approachable	Level	Open	Funding	Heard	repercussions
Albania	Tirana	4	4	4	4	4	5	5	3	
Austria	Vienna									
Belgium	Antwerp	4	4	4	4	4	4	4	4	
Croatia	Rijeka	4	4	5	4	3	4	5	4	
Cyprus	Nicosia	3	3	3	4	3	3	4	3	
Czech Republic	Prague	4	4	2	2	4	2	5	5	5
Denmark	Copenhagen	4	4	4	2	3	2	2	3	8
Estonia	Tallinn	2	3	3	2	3	3	3		
Finland	Helsinki	5	5	5	5	5	5	4	5	5
France	Paris	2	4	4	2	2	2	2	3	8
Georgia	Tbilisi	4	3	2	2	3	3	5		
Germany	Berlin	2	2	2	2	3	2	3	2	2
Greece	Athens	3	4	3	4	4	. 3	5	4	l.
Hungary	Budapest	4	2	4	2	5	5	5	5	5
Ireland	Dublin	3	3	3	2	2	3	3	3	6
Italy	Milano /Roma	4	4	4	5	5	5	5	4	l.
Lithuania	Vilnius	4	4	3	2	2	4	5	4	L
Luxemburg	Luxemburg	2	2	3	2	2	2	2	2	2
Norway	Kristiansand	3	2	2	3	2	3	2	3	}
Poland	Krakow	4	4	3	2	3	4	3	3	3
Portugal	Vila Nova de Gaia	4	5	4	2	2	3	4	4	l.
Republic of N.										
Macedonia	Skopje	3	4	3	3	3	3	4	2	2
Romania	Bucharest	4	4	4	4	4	4	4	4	L .
Russia	Saint Petersburg									
Scotland, UK	Glasgow	3	3	3	2	2	2	2	3	
Serbia	Novi Sad	4	2	4	4	4	4	4	4	
Slovakia	Bratislava	3	5	5	3	5	5	5	5	
Slovenia	Ljubljana	4	4	4	2	5	4	2	4	
Spain	Barcelona	2	2	4	1	1	2	4	3	6
Sweden	Stockholm	5	5	3	2	4	5	5	4	L
Switzerland	Bern	1	1	1	1	1	1	1	1	
The Netherlands	Amsterdam	4	4	2	2	2	2	2	2	2
Ukraine	Kyiv	4	3	4	2	3	3	4	3	
United Kingdom		4	5	5	5	5	5	5		

FP participation

About 80% of FP harm reduction organisations are directly involved in structural cooperation around drug policy with national policy makers.

Figure 9: Is your organisation involved in cooperation exchange at the national level?



Main forms of involvement

The main forms of involvement of FPs are either to provide information, or to participate in discussion forums. To a lesser extent, FPs take part in the direct drafting of policies and guidelines. Many feel, however, that their suggestions are not always included. The main forms in which FPs are involved in cooperation exchange with policymakers are highlighted below, with examples given by FPs.

 Providing information to the government on drug use trends; the needs of PWUD; the needs of harm reduction services; emergency situations; and to give suggestions on possible solutions to problems (e.g. FPs in Antwerp, Belgium; Rijeka, Croatia; Tallinn, Estonia; Berlin, Germany; Porto, Portugal; Krakow, Poland; Bratislava, Slovakia; Ljubljana, Slovenia; Amsterdam, the Netherlands; and in London, UK).

- We have a signalling function for the Ministry of Health. We provide bi-annual updates on the national state of affairs and challenges in the field. Also, if necessary, we have more regular contact. For instance, in between March and June 2020 we held regular contact with the Ministry to inform them on the impact of COVID-19 on harm reduction services. (FP Amsterdam, Netherlands)
- CAPDES is occasionally called to give some inputs on some particular subjects or topics like the evolution of field dynamics. Still, this tends to happen in an urgent situation or, for example, when the State authorities failed to provide detailed information to the media. (FP Porto, Portugal)
- Get By writing annual reports and projects in which we describe the problems we are facing and suggest solutions to these problems. Then the Government finances our efforts in carrying out these solutions and asks for information on our work. Also, we used to have a national office for combating drug use (it was suspended in 2019) which regularly initiated dialogue with NGOs, but now they mostly collect data on our work and our clients. From time to time, they will ask us to fill out a survey in order to get our perspective on a policy or situation. (FP Croatia)
- Participating in drug policy and related forums or committees (e.g. FPs in Prague, Czech Republic; Paris, France; Dublin, Ireland; Luxembourg; Kristiansand, Norway; Barcelona, Spain; Bern, Switzerland; and in Kiev, Ukraine).



- Nationally, we are part of an umbrella organisation (ANO) which chairs the drug policy forum of the government.
 (FP Prague, Czech Republic)
- Fédération Addiction ensures political representation of the health professional network at the national level when dealing with policy makers. We contribute to the decision-making process by communicating about the reality of the current situation in the field, and representing the interests of health practitioners and drug users. During the COVID-19 epidemic, weekly discussions have been organised between the public authorities and us. We've provided them with data about the way the professionals and services tackled the epidemic.

(FP Paris, France)

 Direct involvement in drafting policies and/ or guidelines related to drugs or related issues (HIV, HCV, homelessness, etc.) (e.g. FPs in Stockholm, Sweden; Athens, Greece; Vilnius, Lithuania; Skopje, Republic of North Macedonia; and in Novi Sad, Serbia).

- We are involved in formulating guidelines for needle exchange, OST. Usually, we are asked to give our input and opinion. Sometimes these inputs are included. (FP Stockholm, Sweden)
- C Regarding drug policy, we are taking part along with other CSO's in a group that frequently meets up with the national coordinator for narcotics for bottom-up design and implementation of the national drug strategy. Also, there are several contacts with state officials in the sector of health and social policy regarding the design and the implementation of more humane policies towards vulnerable groups. Also, there is the establishment of a proper collaboration with OST provider entities and other state supported harm reduction services. (FP Athens, Greece).
- Commission to standardisation process. (FP Novi Sad, Serbia)

As it is clear from the previous quotes, despite their participation, some FPs feel that their contributions are not taken seriously, or are not turning into practice.

Cooperation at the municipal level

Evaluation

Figure 10: Is there any structural exchange mechanism between policymakers and CSOs in the field of drug policy in your country, at the municipal level?



Reasons for the lack of municipal-level cooperation

The reasons given for a lack of structural collaboration at the municipal level, when it occurs, is that structural exchange mechanisms tend to be organised at the national level. Another reason is the restrictive posture of the government against drugs.

- In London, there is no mechanism to engage on drug policy as this is seen as a national issue and treatment is devolved to the local boroughs. That being said, there is an opportunity to engage with police in some areas outside of London, especially in relation to diversion schemes which divert people caught in possession of drugs at the pre-arrest stage – five areas of England and Wales have specific schemes. (FP London, UK)
- C The government has a strict zero tolerance policy against all kinds of drugs. (FP Stockholm, Sweden)
- C There are no coordination or policy-making councils at municipal levels. In general, decision-making on drug policy is very centralised and concentrated mainly within the Ministry of Interior and the Ministry of Justice; thus, not much is in place at municipal levels to take local decisions or to facilitate coordination. (FP Tbilisi, Georgia)
- C There are no actual official platforms to give out information about the current situation of drug users to policymakers. There are some "meetings" where the social and health care ministry calls the third sector to hear what they say and then often disagree, saying that it can't be done. There is no official structure for the voice of PWUD and/or the third sector that I know of. The meetings, when rarely organised, are exclusive to only some organisations and have zero influence on anything (FP Helsinki, Finland).



Types of municipal-level cooperation

When cooperation is present, the main types of collaboration follow the national pattern: the most mentioned ones are dialogue (57%) and information (54%), followed by consultation (46%) and, to a lesser extent, partnership(14%).

Figure 11: Types of collaboration between CSOs and government municipalities



(C In very few cases (Turin, Italy, is an example) there have been partnerships for the definition of HR basic assistance levels. It must be noted that in many cities no exchange mechanism is in place, and where it is in place it mostly relates to information and consultation levels. (FP Italy)

FP participation

About 75% of FP organisations are directly involved in structural cooperation around drug policy with municipal policy makers; this is slightly less than their reported cooperation at the national level.

Figure 12: Is your organisation involved in cooperation exchange at the municipal level?



Main forms of involvement

To a certain extent, the main forms of involvement of FPs at the municipal level are similar to their participation at the national level: providing information, participating in local discussion forums, and helping to draft local guidelines. A difference is that, besides being more targeted and specific to the local context, local participation is more personalised and can be more intense.

- Fixpunkt and other providers are deeply involved in drug policy in Berlin.
 (FP Berlin, Germany)
- We have as an organisation and I personally developed a good relationship with municipal authorities, even when they changed after elections. We have taken part in numerous committees regarding various subjects for drug policy and in the response that the city of Athens (mayor, city council, etc.) must have towards PWUD/PWID. Also, as an organisation, we took part in important meetings between the Mayor, his specialised counsellor for addictions and problematic drug use. Moreover, we are working closely together

at various levels with the people who run the Municipal shelter for homeless drug users and we constantly exchange information with the people who run it and the entities behind it. We have also, as an organisation, the luxury - let's say - to communicate directly with the Mayor for various things regarding our issues (FP Athens, Greece)

- Content of the second secon
- Vilnius city municipality has a council on drug prevention issues that meets regularly. Some NGOs are on it. But the council is rather nominal. Cooperation on specific issues with specific municipal officials does happen quite often. (FPVilnius, Lithuania)
- C The achievement has been the immediate improvement in hygiene and safer consumption for users, as well as more widespread health and public order benefits following monthly meetings with working groups of the third sector. Also, active elaboration in the updating of the REDAN Protocols. (FP Barcelona, Spain)
- C The exchange of opinions to find solutions for the housing of drug dependent people who are homeless. There has been an effort from NAAC's position to involve them in the process, although there is still a lot to do. (FP Nicosia, Cyprus)

Also, similar to the national level, participation at the local level is not always regarded as having a positive impact. Nevertheless, involvement at the local level seems to be slightly higher.

(The Porto municipality, in spite of having a small group addressing drug related problems, does not use this expertise in a regular and systematic way. Depending of the political circumstances, the local authorities can be open to a briefing and occasional meeting (normally coming from a community solicitation – situation occurred during the consumption room episode). Still, the traditional approach from the local level is to take decisions based on a very strict and rigid topdown model. Nevertheless, we must say that in the Lisbon municipality, there is a serious effort to establish a more horizontal model in the decision making process. The municipality has created a regular group for consultation (with regular and systematic meetings) named "Porto LX" where civil society NGOs have a seat and are called to express an opinion over different subjects regarding communities living in a vulnerable situation. (FP Porto, Portugal)

CSO contributions are not always taken seriously

Despite participation at different levels, some FPs feel that their contributions are not always taken seriously, and/or are not applied into practice. This was mentioned by FPs in Novi Sad (Serbia), Krakow (Poland), Rijeka (Croatia), and FPs from Athens (Greece) and Helsinki (Finland).



- Last year, we (MONAR Krakow) were twice involved by the National Bureau for Drug Prevention in work on issues related to harm reduction. One of them was a national meeting of people working in harm reduction programmes (something like a conference), which we shaped and programmed. The second event was a discussion on the national plan for HCV treatment. Unfortunately, neither our statements to abandon the criterion of excluding active drug users from treatment, nor any other findings, appeared in later materials. (FP Krakow, Poland)
- C The attitudes and lack of knowledge of policymakers in Finland is the biggest problem; evidence-based policymaking is getting more scarce and intimidation tactics of politicians is growing. Drug users and drugs are a good enemy. We know what things could help people (i.e. help with prevention overdose deaths), but we are not allowed to put them into practice. (FP Helsinki, Finland)
- **C** There is a major underrepresentation of the community. The general environment of CSO's in Greece is not exactly friendly with community leaders and they tend to try to influence and control every collective formed from the directly affected community. There is a huge degree of hypocrisy, both by state representatives and CSO's, towards PWUD/ PWID. Very frequently, we have been used as material for tokenism and our initiatives have been severely attacked by "relics" of civil society that pretend to be friendly with us and [say] that they fight for our own rights. We are excluded from places that by definition are inclusive and it is a common belief that we are not able enough to join decision making processes and groups for drug policy formation. Even in initiatives in which we are included, we are treated as being totally in-

capable of designing effective policy guidelines; and there is the constant urge to totally control us. Also, there is the ''Divide and Conquer'' dogma. Hypocrits that pretend to be our allies try to divide us in order to pass their own agendas. (FP Athens, Greece)

We feel like we can get in touch with policy makers, inform them of trends and problems in the field, but our inputs are rarely considered by policy makers. (FP Croatia, Rijeka)

Civil Society networks and platforms

The majority of FP organisations are part of a civil society network or national platform in the area of harm reduction, human rights, or development aid (see Figure 13). These networks are intended for exchange with other CSOs, either at the national or local level.



Figure 13: Is your organisation part of a civil society network?

Types of networks	Purpose of networks
National or local harm reduction networks and platforms.	Advocate for the sustainability of harm reduction services and for changes in drug policies and practices; produce and col- lect knowledge.
Networks or associations of people who use drugs (including EuroNPUD and INPUD).	Defend and protect the rights of PWUD, make their needs visible, and fight against stigma and criminalisation.
Networks of services working with key popula- tions, such as the homeless, sex workers, youth, or inmates.	Coordinate collaboration in cross-cutting fields; support each other's development and capacity building; implement joint advocacy campaigns; act with competent institutions in cases of human rights violations.
Networks of drug service providers (including, but not exclusively focused on, harm reduction)	Exchange knowledge and good practices, foster cooperation and development among services, as well as service quality.
Task force networks focused on specific diseases, such as STIs, HIV, TB, or HCV.	Exchange and develop knowledge and competencies in the care of specific diseases, and draft specific guidelines and policies.

The main types and purpose of networks in which C-EHRN FPs are involved:

In the case of the FP in Paris (Federación Addiction), the organisations is, per se, a national network.

Civil Society contribution to data reporting

Most C-EHRN FPs are currently contributing to data reporting in their country (other than the present monitoring).

Figure 14: Is your organisation contributing to data reporting in your country?



66 For those not contributing, the reported reasons are lack of capacity and time, or the fact that other organisations are responsible for national data collection on harm reduction (for instance, the EMCDDA Reitox Focal Points) (FPs from Stockholm, Sweden; Helsinki, Finland; Milan and Rome, Italy; Vilnius, Lithuania; Kristiansand, Norway; Krakow, Poland; Glasgow, Scotland; Novi Sad, Serbia; and in London, UK).

For those working on data reporting, the main types of contribution include:

- Provision of data related to their own service provision (number of people assisted, services distributed, treatment completion, etc.). The data is mostly shared with the EMCDDA national Reitox Focal Point, but sometimes is directly shared with local or national government;
- Being part of an Early Warning System for new psychoactive drugs.



Conclusions

Most FPs are directly involved in cooperation exchange with the government, either at the national or the local level. Most of them are also part of networks and contribute to data reporting in their country. Nevertheless, while structural cooperation between CSOs and governments exist in the majority of countries, FPs consider that most of it relates to lower levels of cooperation, such as information exchange and consultation. Both at the national and the local level, governments are much more likely to engage with CSOs to gather necessary data and information to solve specific problems rather than to jointly draft policies and guidelines. Even when higher levels of cooperation occur, several CSOs feel that their inputs are not taken into practice.

More than half of the FPs view government representatives as being easily approachable by CSOs and that civil society can speak openly and frankly and criticise the government without facing repercussions or budget cuts. Nevertheless, important challenges to civil society involvement are still present. The challenges most mentioned relate to a perceived lack of transparency from the government and a lack of balance between different perspectives (from services, communities, and worldviews), besides a lack of adequate funding to CSOs.

All-in-all, it seems that there is still a long path to travel for civil society be meaningfully involved in drug policies. Even in cases where FPs said that they were satisfied with civil society involvement, sometimes this was felt as window dressing. Such a lack of meaningful involvement might also mean that governments are not used to work in partnership with CSOs. More efforts are needed to highlight the importance of civil society participation and to guarantee its practical implementation.

Finally, it is worth stressing the need for quality standards/principles for civil society involvement. Great steps have already been developed by the CSFD in creating some indicators, used as part of C-EHRN monitoring for the first time this year. Further development of these indicators, and evaluating their ability to reflect civil society involvement, is needed. Measuring meaningful civil society involvement is a difficult task. As already mentioned in this report, FPs evaluations might be very subjective, and it is difficult to achieve comparable data in relation to other countries and sometimes even in relation to the previous year. Besides, monitoring CSO involvement might require a broader approach, such as by investigating the needs and views of government representatives, for instance.

References

- Sarosi P, van Dam A, Fulga V. Assessment Report & Literature Review - meaningful civil society involvement in the area of drug policy in Europe. Amsterdam; De Regenboog Groep/ Correlation-European Harm Reduction Network, 2020. http://www.civilsocietyforumondrugs.eu/wp-content/uploads/2020/11/CSFD-AssessmentReport-LiteratureReview-A4_final-1.pdf
- 2. EU. The role of civil society in drugs policy in the European Union. EU; 2006. https://eur-lex. europa.eu/legal-content/en/LSU/?uri=CE-LEX%3A52006DC0316
- Lahusen H, Keane M, Crook L, Košir M, Schiffer K, Ronconi S, et al. Civil Society Involvement in Drug Policy – A Road Map. Amsterdam; CSIDP, De Regenboog Groep / Correlation Network, 2018. https://csidp.eu/wp-content/uploads/2018/11/ Roadmap_web.pdf
- Council of Europe. Code of good practice for civil participation in the decision-making process. Strasbourg: Conference of INGOs of the Council of Europe; 2009, p.17. https://rm.coe.int/CoERM-PublicCommonSearchServices/DisplayDCTMContent?documentId=09000016802eede1
- Lahusen H, Verthein U, Martens M-S. Civil society involvement in drug policy in EU Member States. Assessment report. Amsterdam; Regenboog Groep / Correlation Network, 2018. https://www. drugsandalcohol.ie/29890/1/CSIAP_assessment_ report_2018.pdf



Essential Harm Reduction services



This is a new section in the C-EHRN Monitoring Report. The aim is to start monitoring developments on a range of essential harm reduction services in European cities within the previous 12 months. The state of harm reduction services in the responding cities is assessed from two directions: from the viewpoint of people using the service (service-users) on the one hand, and from different harm reduction services on the other. Respondents were also asked to estimate how their situation compares with the national situation and to name major needs of PWUD in their city.

Differences between services and user groups

The first question concerned the state of harm reduction services for different user groups. Altogether, 13 populations in need of services were named in the questionnaire. The 5-point scale for answering includes the following options (if the services are able to provide services for different groups): to a great extent (5); somewhat (4); very little (3); not at all (2); not relevant to my city (1).

As shown in Table 5, in most of the cities there are services provided to drug users who inject drugs (opioids, stimulants or NPS) and who experience homelessness. The scarcest situation with regard to city-level harm reduction services are for people who use drugs intranasally or by smoking, as well as for migrants and prisoners. However, for the latter two groups (prisoners and migrants) the city level might not be the most relevant when it comes to offering services.

In some of the cities, there are harm reduction services mainly for PWID and sex workers: this is the case in Vilnius (Lithuania), Tbilisi (Georgia) and Stockholm (Sweden). In addition to the named 13 populations, one more service user group was named: in Krakow (Poland), there are specific services for aging drug users. The second question assessed the situation in cities with regards to 19 different harm reduction services. In general, the most prevalent harm reduction services in 35 European cities were needle and syringe exchange programmes (NSP), opioid substitution treatment (OST), and outreach work. The least prevalent services were drug consumption rooms (DCR), drug checking through the use of fentanyl test strips, and naloxone provision in prisons.

In addition to the named 19 services, other activities were also recognised. In Tallinn (Estonia), officials have launched a programme which gives the police the possibility to refer detained drug users to a support programme instead of punishment. In Krakow, there is a special day shelter for homeless stimulant users in a drop-in centre to prevent psychoses.

It must be born in mind that the estimations as to the extent to which services are available are subjective and, thus, not easily comparable between countries and cities. However, if a comparison is made, the best overall average (mean of service provision for all 13 subsets; see Table 5) when it comes to serving all 13 groups of PWUD was in Bern (Switzerland), Paris (France) and Tirana (Albania). Again, in comparison, the overall situation was estimated to be worst in Vilnius, Stockholm and Budapest (Hungary).



	People	People	People	People who		Sex	People	Women	LGBQTI	Young	EU	Non-EU	People in
	who inject	who	who	smoke	use	workers	experienci			people who	Migrants	migrants	prison
	opiates (including	inject stimulant	smoke opiates	stimulants or new	intranasal		ng homeless			use drugs (under 18			settings
	synthetic	s or new	opiates	psychoacti	amphetami nes/cocain		ness			years of			
	opioids)	psychoac		ve	e/		11655			age)			
	opioid3)	tive		substances	cathinone,					age)			
		substanc			etc.								
		es											
Tirana	5	4	5	4	3	5	5	5	5	5	5	5	5
Vienna	5	4	3	3	4	3	5	4	3	4	4	3	2
Antwerpen	5	5	5	4	4	4	4	4	3	3	4	4	4
Rijeka	5	4	2	2	2	4	5	5	5	5	3	3	4
Nicosia	4	4	4	4	3	2	3	4	3	2		3	3
Prague	4	4	4	4	4	4	5	3	4	4	4	4	3
Copenhagen	5	5	5	5	2	4	4	4	4	3	4	4	1
Tallinn	5	5	5	4	4	4	3	4	3	2	1	1	3
Helsinki	5	5	4	4	4	4	4	4	3	4	4	3	3
Paris	5	5	5	5	5	5	5	5	5	4	4	4	4
Tbilisi	5	5	3	3	2	5	2	5	2	2	2	2	3
Berlin	3	3	5	4	3	5	4	4	4	3	4	4	4
Athens-	4	3	4	3	3	3	4	3	3	3	3	3	3
Budapest	3	3	1	3	3	3	3	3	3	3	2	2	2
Dublin	5	5	4	4	4	4	5	4	4	2	4	3	4
Milano	4	3	4	3	3	4	4	4	4	2	4	3	4
Bishkek	5	4	2	2	2	5	5	5	5	3	1	5	5
Vilnius	4	2	2	2	2	3	2	3	2	2	2	2	2
Luxemburg	4	4	4	4	3	5	5	3	4	3	4	4	5
Kristiansand	5	4	4	3	3	4	4	4	4	4	3	3	4
Krakow	5	5	2	4	3	3	5	3	3	5	4	5	4
Vila Nova de	5	3	5	3	3	4	4	3	3	3	4	4	2
Gaia													
Skopje	5	4	4	4	4	5	4	4	4	5	4	4	1
Bucharest	4	4	3	3	1	4	4	4	4	4	1	1	2
Saint	5	4	3	3	4	4	4	5	4	2	2	3	2
Glasgow	5	4	5	4	4	4	5	5	4	3	3	4	5
Novi Sad	5	4	3	3	3	5	5	5	5	5	5	5	3
Bratislava	4	4	3	3	3	4	4	4	4	2	1	1	2
Ljubljana	5	5	5	5	5	3	5	5	3	4	3	3	4
Barcelona	4	5	4	4	5	4	5	4	4	3	3	3	4
Stockholm	4	4	2	2	2	2	2	2	3	2	2	2	2
Bern	5	5	5	5	5	5	5	5	5	5	5	5	4
Amsterdam	5	4	5	4	5	5	5	4	5	4	5	4	3
Kyiv	5	4	2	2	2	5	4	4	4	4	3	4	4
London	5	5	3	3	3	4	5	4	4	4	5	5	4

Table 5: Are harm reduction services in your city able to provide services for the following populations

	Outre	Peer	NSP	Safer	Safer	DCR	Drop-in	Drug	Fentanyl	OST	OST	Nalox	Take-	Naloxo	Sexual	HIV	Specific	Online	Shelt	Legal
	ach	suppo		smoking	intranas		center	checki	Test		in	one	home	ne in	risk	service		harm	ers	suppor
	work	rt		kits	al kits			ng	strips		prison		naloxon	prison	prevent	S	ent	reducti		t
													е		ion	(preve	opportuni	on		
																ntion,	ties/inco			
																testing or	me generatio			
																treatm	n for			
																ent)	PWUD			
Tirana	5	5	5	5	4	1	4	1	1	5	5	3	3	3	5	· · · ·		3	3	3
Vienna	5	3	5	4	4	2	4	4	1	5	4	3	3	2	4	5	3	3	5	3
Antwerpen	2	4	4	3	4	2	5	2	2	5	5	2	2	2	4	5	3	3	4	3
Rijeka	2	4	5	2	2	2	5	2	2	5	5	3	2	4	4	4	4	4	5	4
Nicosia	2	2	4	2	2	2	4	2	2	4	4	4	4	2	2	4	2	2	2	2
Prague	5	4	5	3	3	2	5	2	4	4	3	3	3	2	4	5	3	4	4	4
Copenhagen	5	5	5	3	2	5	2	5	5	4	4	2	4	4	4	4	4	4	4	4
Tallinn	5	5	5	2	2	2	5	3	2	4	4	5	5	4	5	5	3	3	3	3
Helsinki	5	4	5	1	3	2	5	2	2	5	5	3	2	3	5	5	3	5	4	
Paris	5	5	5	5	5	4	5	5	4	5	5	4	5	4	4	5	3	4	3	4
Tbilisi	5	5	5	2	3	2	4	3	2	5	2	5	5	2	5	5	3	2	2	4
Berlin	5	3	5	5	4	5	5	2	2	5	5	3	3	2	4	5	4	3	4	4
Athens-	3	3	3	2	2	2	3	2	2	5	4	2	2	2	4	4	2	2	3	3
Budapest	3	2	3	2	2	2	4	2	2	4	2	2	2	2	3	3	3	3	4	
Dublin	5	3	5	4	4	2	4	2	2	5	5	4	4	2	4	4	3	4	5	3
Milano	2	3	4	3	3	2	4	3	2	5	5	4	4	3	3	3	3	3	2	
Bishkek	5	5	5	2	2	2	2	2	2	5	5	5	5	5	4	5	3	2	3	
Vilnius	3	2	4	2	2	2	2	2	2	4	3	4	4	2	4	4	2	2	2	2
Luxemburg	3	2	5	3	3	5	5	4	2	5	5	4	3	2	5	5	3	4	5	
Kristiansand	2	5	4	3	2	2	4	2	2	5	5	5	5	4	4	4	4	3	4	4
Krakow	3	3	5	2	2	2	5	4	3	5	5	2	2	2	4	5	3	3	5	5
Vila Nova de	5	4	5	4	3	2	4	2							4					
Gaia																				
Skopje	5	5	5	2	2	1	5	1	1	5	5	3	1		5	5	2	4	2	5
Bucharest	2	3	4	1	1	2	2	2		4	3			2	4	4			4	
Saint	5	5	5	2	2	2	3	2							4				3	
Glasgow	2	5	5	2	2	2	5	2							5				5	
Novi Sad	5	5	5	2	2	2	5	2							5				2	
Bratislava	5	3	4	2	2	2	4	2							3					
Ljubljana	5	3	5	1	4	1	5	5							4					
Barcelona	5	4	5	4	4	5	5	4			5	-			5				3	
Stockholm	2	5	5	2	2	2	4	2				-	-		2					
Bern	5	5	5	5	5	- 5	5	5							5				5	
Amsterdam	5	4	5	4	4	5	5	5											5	
Kyiv	5	5	4	2	2	2	3					_							4	
London	2	4	5	2	2	2		2												

Table 6: Are the following services available in your city for people who use drugs?

Highlight: Harm reduction in Tirana, Albania

People who inject drugs (PWID) are one of the largest key populations in Albania. A total number of 6,182 PWID (range of 3,626 to 8,737 with a 95% confidence interval) have been estimated to live in Albania. A crude estimate is that in Tirana there are 4,324 PWID. The HIV prevalence is low in Albania; it is estimated that 1% of PWID live with HIV.

Interview with Besnik Hoxha from Aksion Plus

Q: According to the data you reported for the monitoring, Tirana is able to reach various types of sub-groups of people who use drugs and also offer several types of harm reduction services.

Are these services based on the national drug strategy?

A: Yes, in some regions for specific issues. Yet there is a budgeting gap for some activities.

Q: Could you let us know how have these services developed? And for how long have you had them?

A: HR programmes started around 2000 and are delivered via: drop-in centres, mobile units, outreach workers, and secondary distribution.

Q: How are the services organised? (are the different population groups assisted in the same service or are there specific services for different key-pops?)

A: We mostly serve PWID and their injecting/sexual partners, in addition to their

family members. Other sub groups are also included: female sex workers and LGBTI.

Q: How many harm reduction services are there in Tirana? Is the coverage sufficient?

A: In Albania, HR services are provided to PWID by two NGOs, one that implements the OST programme in Tirana and eight other municipalities, and the other one that provides the needle and syringe programme in Tirana and three other municipalities. HIV prevention services to PWID are provided by two NGOs, Aksion Plus and Stop AIDS.

Aksion Plus implements the Methadone Maintenance Treatment (MMT) programme in Tirana, Durrës, Vlorë, Shkodër, Elbasan, Berat, Korçë, Sarandë and Fier. Nearly 950 clients attend OST centres, and in prisons as well. We provide MMT, as well as condoms, lubricants, and IEC materials at their drop-in centres. Services are provided by 15 office staff which includes a manager, medical doctors, nurses, a lab technician, psychologists, and social workers.

Stop AIDS implements needle and syringe programming in Tiranë, Durrës, and Elbasan. Stop AIDS provides needles and syringes, condoms, and HIV and other STI testing through both outreach and drop-in centres. Services are provided by office staff that includes a manager, medical doctor, nurses, a lab technician, psychologists, social workers and 25 outreach workers and peer educators.

Regarding testing of HIV and other STIs, they are conducted by almost all GFATM sub-recipientS focusing on PWID, MSM and FSW. Currently, under the GFATM support, HR services are being offered in nine cities of the country, in addition to eight prisons (pre-detention and prison settings).

Q: What types of services are those?

A: OST with nine centres across the country and a combination of HR services is being applied: drop-in Ccentres, mobile units, and outreach activities.

Q: Are peers involved in service delivery?

A: Yes. Former or current drug users are involved as outreach workers.

Q: Who pays for them? City, national sources, donors? And how sustainable they are?

A: They are paid by the Global Fund project. We are planning to involve local and central government to continue these services after the GFATM project. Some laws and bylaws are drafted for this purpose. WHO Albania is facilitating this process.

Q: Can you let us know your current challenges/future plans, what is missing?

A: Lack of funds, low salaries of the staff, staff burnout, lack of interest and will from the government to invest in harm reduction programmes, as well as a lack of interest from international donor agencies to support harm reduction programmes. Only UNFPA Albania has funded some complementary activities for PWID and other sub groups. In the presence of COVID-19, everything is getting more complicated and uncertain.



How do cities compare with the national situation?

Respondents were asked to estimate how their city compares with the national situation in terms of harm reduction coverage. A great majority (94%) answered that their city has better (79%) or similar (15%) service coverage than the rest of the country. This clear result can probably be explained by the fact that most of the partaking cities are capitals and/or bigger cities which typically have more PWUD and harm reduction services than smaller cities or rural areas.

Despite this – the relative situation being better in their city than nationally – a majority (59%) of respondents said that current harm reduction services are insufficient to meet the needs of PWUD.

Figure 15: Are harm reduction services in your city able to provide services for the following populations? (N=34)



Figure 16: Are the following services available in your city for people who use drugs? (N=34)



Figure 17: Do you feel the current harm reduction services at your city can meet with needs of PWUD?



Figure 18:. How does your city compare with the national situation in terms of harm reduction coverage?



Improvements needed to harm reduction services

Respondents were asked to freely describe the major needs that PWUD might have in their cities, and what is needed to improve the harm reduction services. The main needs and anticipated improvements that were mentioned are as follows:

- Tirana: shelters, employment opportunities for PWUD.
- Vienna: safer use measures in prison, employment opportunities for PWID, drug consumption rooms (DCRs).
- Antwerp: more services and funding.

- Nicosia: methadone maintenance, shelters, harm reduction in prison.
- Prague: more services, more OST, opening of a DCR.
- Copenhagen: More health services, substitution treatment for those without a Danish social security number.
- Tallinn: safer-use information also for non-injecting drug use, DCRs, shelters for PWUD (when not sober).
- Helsinki: drug checking service, DCR, better access to HCV treatment.
- Tbilisi: services for non-injectors, new types of OST (heroin and other), drop-in services at existing harm reduction sites.
- Athens/Thessaloniki: housing, psychosocial support with integrated services.
- Budapest: political support for harm reduction/no attacks against service providers, adequate funding for services, trained and motivated professionals/peer leaders.
- Dublin: Access to income generation for PWUD, DCRs, greater access to naloxone.
- Rijeka: drug checking, better availability of naloxone.
- Milan/Rome: 24/7 HR services, better service coverage, DCRs.
- Bishkek: OST with buprenorphine, gender-based services, a comprehensive client-oriented approach.
- Vilnius: Scaling-up of existing services, outreach work, drug checking.
- Kristiansand: DCR, low threshold outreach work, more OST, employment for PWUD.
- Krakow: DCR, naloxone distribution.
- Porto: DCR, take-home naloxone, HR services for women, improving shelters and services for the homeless, alcohol interventions, interventions for stimulant users, extension of drug checking.
- Skopje: Improving HCV prevention, safer smoking kits, drug checking, take-home naloxone.
- Bucharest: improving service coverage and quality, coordination with other services.



- St. Petersburg: more mobile units and dropin centres for PWUD.
- Glasgow: greater availability of injecting equipment, lower threshold to and wider choice of OST, DCR, drug checking.
- Novi Sad: take-home naloxone, shelters for PWUD, safer smoking kits and intranasal kits, water for injections, masks, shelters for PWUD, extending DCR opening hours, more DCRs for smoking.
- Stockholm: anonymous needle exchange services, quicker access to OST, naloxone without prescription (peer-to-peer distribution).
- Amsterdam: more services and outreach work among MSM who inject drugs.
- Kiev: community centres and drop-in centres, increased availability of services and right to choose between services, shelters, dispensing food, free rehabilitation.
- London: DCRs, drug checking, paraphernalia for smoking and intranasal use.
- **66** To allow several organisations that have experience and access to clients to deliver harm reduction services in the city. The choice of a service provider should not have one criterion (for selecting a winner) - a low price. This directly affects the quality of the services provided. The volume and burden on social workers (which now exist within the framework of state procurement) should be reduced, since they do not allow the provision of quality services and access 'hard-to-reach groups'. Accordingly, the price for services should be revised and provide for the possibility of high-quality service provision. Organisations that receive government money must register as VAT payers. This changes the accounting of the organisation, complicates it, and leads to the risks of auditing. The mechanism that is used in the transition of services to state funding should not change the structure of the work of non-governmental organisations, and vice versa, to contribute to the develop-

ment of civil society organisations in Ukraine; the development of the harm reduction package and algorithms of work with the involvement of community members." (FP Kiev, Ukranie)

- If we consider the marginalised PWUD that primarily use base-cocaine and heroin: the services generally meet their needs. Similarly, PWUD in the party scene have access to drug testing in two different locations and access to ample drug education. As far as we know, the only PWUD group that is not served sufficiently are MSM who inject their drugs in chemsex settings. They have limited access to services as they do not use the needle exchange services for homeless injectors, and they are also hard to reach for education on safer injection practices. Unsafe injection practices are common." (FP Amsterdam, Netherlands)
- **(** There is a need for "legal reform in respect of the interventions described as the main needs (see London in the list above) or an increase in local agreements with police and public health to permit these activities. Greater funding - local authorities became responsible for funding and commissioning of drug and alcohol services under the Health and Social Care Act 2012, while facing an estimated 37.3% reduction in central government funding between 2010-11 and 2015-16. As a result, "drug misuse treatment" has faced more reductions in funding than any other public health area in 2016-17 with a 14% reduction in funding between 2015-16 and 2016-17. Net expenditure on adult drug and alcohol services has decreased by 19% in real terms between 2014-15 and 2018-19." (FP London, UK)

In general, many respondents generally called for more funding and political support for harm reduction as well as wider understanding of harm reduction approach their cities and countries.

Conclusions

Often, country-based monitoring reports are only provided if a given country has, or has not, a certain harm reduction service. To be more analytical, it is important, 1) to go to the city/local/ regional level; 2) assess differences in service provision between cities and within one country; and, 3) assess which services are lacking in the city and if the existing services can meet demand.

The overall picture from this novel part of the C-EH-RN monitoring tool is that there is an insufficient amount of harm reduction services available in European cities, and that in many cities the existing services are largely focused on, and limited to, PWID and their services (especially NSP and OST). In most European cities, the harm reduction services lack funding and political support. In many cities, the integration of harm reduction services with other parts of the health and social care system is weak.

Info box: PWUD experiences

As part of C-EHRN monitoring in 2020, a separate questionnaire was sent to PWUD. Altogether, 38 PWUD from 36 different European cities responded.

Respondents were reached through the European Drug User Union (ENPUD) and its national member organisations. In what follows, assessments by PWUD are presented as to the state of harm reduction services in their city of residence.

In many respects, the answers from PWUD are in line with answers from C-EHRN focal points. As shown in Figure 19 below, PWUD also assess that needle exchange services and OST are relatively easy to access in their cities. Alternatively, drug consumption rooms, take-home naloxone, and drug checking services are rare.

NSP HIV care OST Prevention of sexual risk Outreach wor Drop-in Legal suppo Mobile Peer education Secondary NSF (P2P) Take-home Peer-to-pee Naloxor Online HR Housing Incor generation Drug Checking Female-specifi Safer nasa Safer smoking skits DCR Over-the r Naloxone Access is easy Access is difficult Not available

Figure 20 below shows that only 11 users out of 38 felt that PWUD are involved in the planning, delivery and evaluation of harm reduction services in their cities. A participatory approach in harm reduction service design is clearly something to be developed.

Figure 19: Please check the following harm reductio services and tell us how accessible they are for PWUD in your city (n=38)





PWUD were also asked to freely describe what difficulties they encounter in their daily lives. The following are some answers from different cities:

Prague (Czech Republic): "No place to inject, not enough OST possibilities, no substitution for meth, housing problems, no naloxone".

Helsinki (Finland): "Services are open at 'office hours', there is no harm reduction services at night time. HCV-testing is easy to get, but the path to treatment isn't available easily. Low threshold services need more service-hours."

Athens (Greece): "No money at all for community networks, we can't work voluntarily any more, we have produced debts in our family and our house."

Budapest (Hungary): "Basically 2-3 places remain in the city who work with active addicts, and they usually have narrow opening hours, limited staff and equipment because of their short budget. Even if there are other services, they are on the other side of the city, or you don't even know about them. The governments methadone programme has a limit that has been maxed out for years, so you can't get methadone (only on the black market), only Suboxone." **Dublin (Ireland):** "Homelessness is a big issue. Alcohol use among PWUD and an unreliable black market for benzos, Z-drugs and Pregabalin."

Vilnius (Lithuania): "Harm reduction services are very weak. Most of them are concentrated on injecting drug users, while more and more people are using drugs intranasally or orally, or smoking. As well, there is no peer-to-peer involvement at all in the existing services (excect "Young Wave", who are providing harm reduction in nightlife settings and who are not getting any funds/support from the government)."

Krakow (Poland): "The most significant problems are the issues related to OST and difficulties with them. In the OST (methadone) programme in Rydygier Hospital there has been conflict with the personnel, especially with the doctor who is head of the treatment programme. For the doctor consultation, patients have to wait for months. Rules for methadone patients are too strict and it is too easy to be relegated from treatment."

Bucharest (Romania): "Hard to get into rehab or get new needles other than in the pharmacy. NSPs are few and far away."

London (UK): "We have lost a lot of very knowledgeable workers when it comes to harm reduction. It's all about the recovery agenda and people successfully completing drug treatment. We need to go back 10-15 years when we were treated with respect as adults."

5

Hepatitis C



Introduction

People who inject drugs (PWID) account for the majority of new cases of hepatitis C virus (HCV) infections in Europe. In the WHO European region, an estimated two million PWID are living with active HCV infection, about 75% of whom are thought to live in Eastern European countries.

On the basis of the first C-EHRN monitoring of 35 European countries in 2019, as well as other information sources, it is evident that HCV testing and treatment for PWID remains insufficient: despite progress reported from several countries, further improvements of the existing continuum-of-care interventions for PWID are needed (1,2).

This section reports on the experiences of CSOs that provide harm reduction services as to the availability of, and access to, interventions that constitute the HCV continuum of care. It consists of four parts: 1) the use and impact of national strategies/guidelines on accessibility to HCV testing and treatment for PWID; 2) the functioning of the continuum-of-care in different countries and regions; 3) potential changes in the continuum of services compared to the previous year; and, 4) the role of harm reduction services and PWID NGO's in this context.

Unlike in 2019, the questions focus mostly on city level situations instead of national level. On many reported issues for the HCV chapter, however, we can assume that the city and national situation are the same, which enables us to make also comparisons in progress between 2019 and 2020.

When making comparisons, however, it should also be born in mind that there are some differences in participating countries and cities between 2019 and 2020.

Results

HCV prevention at current national policy level

Firstly, respondents were asked to assess the summarised public information on HCV at the EMCD-DA website about their country and whether it was up to date, or if anything had changed recently regarding new or updated hepatitis C strategies, guidelines, etc.

For the majority of countries the information on the EMCDDA website was up to date (see Figure 21). It must be noted, however, that Georgia, North Macedonia, Serbia and Ukraine, who also chose this answer, are not part of the EMCDDA's data gathering network. Countries that belong to the network and said their information needs to be updated included Austria, Croatia, France, Ireland and Lithuania.

Figure 21: Is the EMCDDA information in the summary of your country up to date, or did anything change recently regarding new or updated hepatitis C strategies, guidelines, etc.?¹⁵





National guidelines and real-life practices

One part of the C-EHRN monitoring survey assesses the use and impact of national strategies or guidelines on accessibility to testing and treatment for PWID. Respondents were asked to assess the use, and impact, of national strategies or guidelines on access to testing and treatment for people who use injectable drugs from the viewpoint of services working with PWUD.

Almost all countries use either their own national guidelines or EASL guidelines that include PWID (see Figure 22). Only three countries - Cyprus, Lithuania, and Russia - did not have any HCV guidelines which would include information related to related to PWID¹⁶.

Respondents were asked about the implementation of national HCV guidelines. Even if guidelines exist, they might have limited relevance in practice. A range of challenges - such as outdated guidelines and complicated testing and treatment systems - as well as a lack of services and other disparities between formal guidelines and reality were reported. Respondents (whose country has guidelines) were then asked to assess how these guidelines impact access to HCV testing, treatment and other services for PWID. Overall, many respondents saw a positive impact of the guidelines, but many also did not answer this question, which may indicate that the relevance of the guidelines is unclear. The least positive impact (48%) of the guidelines was seen by services provided by CSOs (see Table 7).

Figure 22: Do you think these guidelines impact accessibility to testing and HCV treatment of people who inject drugs (PWID) in your city?



Table 7: In Which areas did you notice that Guidelines had positive impact on access to testing, treatment and other services for people who inject drugs (PWID) (answered: n=23; not answered: n=11)

	%	n
HCV information & counseling	74	17
HCV treatment	70	16
HCV testing	61	14
Specialised HCV services	52	12
CSO services	48	11
Other	22	5

In some countries, respondents also saw a negative impact of the guidelines. Reasons for a negative impact included, "HCV treatment is prescribed only by specialists" (n=9), "HCV treatment is not possible outside the specialised healthcare system" (n=7), and "HCV testing is not possible outside the healthcare system" (n=3).

Ukraine reported that the current national guidelines mention PWID only with regards to testing, new guidelines being developed but they had not been approved for use yet at the time of data collection. The new guidelines will give priority to PWID to be treated:

C "Diagnostic analyses are necessary for treatment and are not cheap, so often PWID don't have access to treatment services. Also, there is a high level of stigma and discrimination from health professionals towards PWID. The current National Guidelines mention PWID only in regard to testing. The new modern guidelines have been developed by the Centre of Public Health but are not approved for use yet. The guidelines will give priority to PWID to be treated." (FP Kyiv, Ukraine)

Table 8: Restrictions applied in HCV treatment (n=13 countries)

	%	n
Other restrictions	40	6
Restrictions to injecting drug use	31	4
Accessible for former injecting drug users	23	3
Accessible for drugs users on opioid substitution treatment	23	3
Accessible only for F3 and F4	13	2
Accessible only for F2, F3 and F4	13	2
Accessible only for F1, F2, F3 and F4	13	2

Availability of, and access to, new drugs (DAA's)

According to the 2020 data, the new drugs for HCV treatment (DAA's) a¹⁷re available in all countries¹⁸. However, there are still a range of perceived restrictions to DAA access. Altogether 13 countries (19%) reported different restrictions in the use of DDA's.. A list of applied restrictions is presented below (see Table 8). Various restrictions are still common, even if there is a slight improvement in the overall situation. DATA REPORT

The biggest group, six countries, was in the category "other restrictions" which include:

- Austria: "DAAs reimbursed only when prescription comes from a specialized hepatological centre".
- Georgia: "Only Gilead products are accessible".
- Lithuania: "The main restriction is mandatory state health insurance which not all PWIDs have."
- Slovakia: "PWID patients, twelve months of abstinence confirmed every three months by toxicological examination".
- Sweden: "Abbvie's medicine for all genotypes is very limited access because of the high price."
- Cyprus: "Very restrictive criteria for the provision of DAAs in general, along with its very high price."

Respondents were asked in which countries people who use, or have used, drugs - either active users, former users or users on OST – are allowed formal access to HCV treatment. With regard to PWID access to DAA treatment, the worst situation is reported from Albania, Romania and Russia where DAAs are not allowed for any subgroup of PWID.

Active drug users are not allowed access to HCV treatment in 9 countries. In this regard, the situation has remained the same as in 2019; in addition to Albania, Romania and Russia, these countries include Estonia, Lithuania, Poland (official policy excludes active drug users from treatment but in practice they are able to access treatment), Serbia, Slovakia, and Ukraine.

Are DAAs used according to official policy?

The majority of C-EHRN focal points reported that in their cities (and presumably also on a country level) DAA's are used according to the official policy, but there were also four countries where there is a discrepancy between policy and practice:

- (Budapest/Hungary), most PWID cannot access treatment even if the official policy allows and requires it¹⁹.
- (Milano & Rome/Italy), even if there are no official restrictions in the access criteria, "in practice vulnerable groups, including PWID, are not supported in accessing testing and treatment for HCV."
- (Krakow/Poland), the situation is the opposite: although the guidelines exclude active drug users from treatment, in practice they are generally accepted for treatment.
- (Novi Sad/Serbia), "the national guidelines do not discriminate PWID, but in practice, the DAA is given to very few people, so those priorities do not include PWID".

Who is paying for HCV treatment?

HCV treatment with DAA's is reimbursed by health insurance or the public health service in most of the countries. Treatment with the new drugs is reimbursed with no limitations in 22 countries (67%) and with limitations in nine countries (27%). In two countries (Albania and the UK)²⁰, hepatitis C treatment with DAA t is not reimbursed.

Figure 23: Is treatment with the new drugs for hepatitis C (DAAs) reimbursed in your city (country)?





Those mentioning limitations in reimbursement include: Czech Republic, 'only one treatment possibility is covered as an unofficial rule (difficult to get reinfection treated again); Estonia and Finland, many PWID are denied treatment if they are actively using drugs, even if this is not in accordance with official guidelines; Georgia, patients need to pay the costs of diagnostics on reinfection; Poland, the waiting times for an appointment with a specialist are long, as are the waiting time for an assessment of the stage of liver fibroscopic examination; Slovakia, there are several difficulties (continuous proof of abstinence; the need to register with an unemployment office every two weeks; and difficulties in getting confirmatory tests) in getting reimbursement in practice.

Changes in the continuum of care

A well-functioning continuum of care, including low threshold and harm reduction services, is important for accessibility and impact of HCV testing and treatment. It is crucial to improve the low uptake of HCV testing and treatment among PWID by including the addiction treatment services, harm reduction facilities and drug user organisations in the continuum of services that provide HCV management within every European country.

C-EHRN monitoring contains a pattern of questions asking how the continuum of care is functioning in different countries and regions.

In most countries (85%), PWID can have a rapid test for HCV (see Figure 24) in low threshold settings at harm reduction services. Rapid tests are also quite commonly available in drug treatment (65%) and at infectious disease clinics (62%). PWID can get tested by a general practitioner in about half of the countries (44%; 51% in 2019). However, rapid testing for PWID at pharmacies has remained very rare. Similar to last year, confirmatory blood testing for HCV RNA is most commonly available for PWID at infectious disease clinics (97%) and gastroenterology clinics (65%) but, compared to last year, their availability seems to have improved at drug treatment clinics (50%; 35% in 2019) and at harm reduction centres (41%; 26% in 2019). The increase is encouraging but we must bear in mind that there are some differences in participating countries in 2019 and 2020.

As in 2019, PWID are most commonly treated for hepatitis C at infectious disease clinics (in 90% of reporting countries) and gastroenterology clinics (in 65% of reporting countries). In 32% of countries, treatment was provided at harm reduction services or community centres.

Point-of-care testing increases HCV testing and linkage to care. It is important that facilities offering testing are able to offer both HCV testing and treatment. There are still big differences within Europe as to where, and how, PWID can undertake a HCV test and Inequities in access exist across European countries, regions and cities. It can also be concluded that the integration of testing and treatment at the same location is still too rarely the case.









Figure 25: Where can people who inject drugs (PWID) perform a confirmatory blood test for HCV RNA?

Figure 26: Where can HCV infected people who inject drugs (PWID) perform a non-invasive diagnostic procedure for the evaluation of the stage of liver disease (i.e. Fibroscan®)?



Figure 27: In case where DAAs are accessible to people who inject drugs (PWID), where are they treated for HCV?



Are there written guidelines for the linkage-of-care?

Respondents were asked if the linkage-to-care for PWID is achieved by a written protocol or guidelines. More concretely, they were asked to assess if there is, for instance, an agreed protocol to refer clients from harm reduction services to other treatment and care systems. Respondents from 14 countries answered that the protocol/guidelines were clear, but in 13 countries they were regarded as unclear. Respondents of three countries could not make an assessment.

Figure 28: Is linkage-to-care for people who inject drugs (PWID) achieved by a written protocol/guidelines?



More or less action and coordination on HCV?

Focal points were also asked to compare the changes in HCV activities between 2018 and 2019 have PWID service providers in their country invested more or less attention in HCV awareness campaigns, testing at their own location, and treatment at their own location? Most commonly, these activities were reported had either remained at the same level or improved.

HCV awareness raising and testing at their own location was reported to have become worse in Croatia, and HCV treatment at their own location has become worse in Albania, Croatia and Germany.

Figure 29: Compared to 2018, have service providers for people who inject drugs (PWID) in your city invested attention in 2019 to the following activities?



When asked about the progress in coordination between health care and social care providers (especially NGOs and harm reduction services), in most of the cities and in all dimensions (information sharing, communication, service provision) the situation has remained the same or improved.

Negative progress was reported by four countries: information sharing has become worse in Croatia and Greece; communication has become worse in Croatia, Greece, Ireland and Sweden; and, servicer provision has become worse in Croatia, Greece and Sweden. Figure 30: Compared to 2018, did the coordination between health care providers (general practitioners, clinics) and social service providers (like non-governmental organizations, harm reduction, services) change in 2019 regarding HCV?



The role of harm reduction and drug user organisations

In countries with progressive HCV treatment policies, drug user interest groups²¹ have had a pivotal role in raising the issue with the public through awareness, and in advocating for the right of PWID to low threshold HCV testing and treatment. In 2020, the question was addressed at the city level whereas in 2019 it was at the country level. In 2020, 16 of the partaking cities had drug user groups actively working for political awareness with regards to HCV. In 10 cities the user groups were not active with regard to HCV, and in 7 cities (response option "other") no drug user group existed. In 2019, there were active user organisations in 15 countries.



Figure 31: Are drug user groups active for (political) awareness with regards to HCV in your city? $(n=33^{22})$



Conclusions

The results from 2020 show that PWID are still in an unequal position, and often deprived of proper HCV interventions, in different European cities and countries The overall picture has not changed much from that of 2019. There is still a lot to be developed in the continuum-of-care, and the integration of testing and treatment at one site is still too rarely the case.

Almost all countries have either their own national guidelines or EASL guidelines that include the HCV management for PWID. Only Cyprus, Lithuania, and Russia do not have such HCV guidelines.

The new drugs for HCV treatment (DAA's) are available in all countries, even if in practice there remains restrictions for PWID to access DAA's in some countries.

What is postive is the fact that 85% or respondents reported accesss to rapid test in harm reduction sites: Whilst still more needs to be done this is an encouraging result.

Active drug users are not allowed access to HCV treatment in 10 countries. In this regard, the situation has remained the same as was in 2019. In addition to Albania, Romania and Russia, these countries include Estonia, Lithuania, Poland ((although officially they do not have access but in practice yes), Serbia, Slovakia, Slovenia and Ukraine.

On the positive side, however, is that several countries report progress, more action and attention as well as better communication and information sharing is taking place on HCV. In 2020, countries that reported negative developments included Croatia, Greece, Ireland and Sweden.

References

- Tammi T, Rigoni R, Matičič M, Schäffer D, van der Gouwe D, Schiffer K, et al. Civil Society Monitoring of Harm Reduction in Europe, 2019. Data Report [Internet]. Amsterdam; 2020. Available from: https://www.correlation-net.org/wp-content/ uploads/2020/02/C-EHRN_monitoring_web.pdf
- Matičič M, Pirnat Z, Leicht A, Zimmermann R, Windelinck T, Jauffret-Roustide M, et al. The civil society monitoring of hepatitis C response related to the WHO 2030 elimination goals in 35 European countries. Harm Reduct J [Internet]. 2020;17(1):89. Available from: https://doi. org/10.1186/s12954-020-00439-3



Overdose prevention



Introduction

Preventing overdose and overdose deaths²³ is an ongoing public health challenge. Drug overdose (OD) is a major cause of death, especially among young people in Europe. In the region, people who use opioids are 5 to 10 times more likely to die than their peers of the same age and gender (1). In 2019, countries reporting to the EMCDDA (28 EU Member States, plus Turkey and Norway) registered more than 9,400 overdose deaths; that represents a slight decline from 2017, although underreporting is likely to occur (2). Half of these deaths were reported by the UK and Germany alone; such high numbers might be related to underreporting in other countries and a higher at-risk population in these two nations.

A number of evidence-based harm reduction interventions help in the avoidance of overdose and overdose deaths. Among them are access to naloxone (3-5) programmes that enable bystanders to provide first aid and administer naloxone before an ambulance arrives can save lives. We conducted a systematic review of the available studies on take-home naloxone to reverse opioid overdose and included 21 studies for analysis (with various study designs, Opiate Substitution Therapy (OST) (6,7), Drug Checking services, and Drug Consumption Rooms (DCRs) (8–10) which provides a drug consumption room in Barcelona among other services. The objectives of our study were to compare the client profile, the facility use, the drugs used, and the number of non-fatal overdose episodes between (1. OD prevention campaigns and training, as well as guidelines on how to prevent overdoses, are key in addressing and reducing the risk of an overdose (11,12). Despite the evidence, overdose prevention measures are not always implemented on the ground; at least, not to the extent that they are needed.

This chapter focuses on mapping the state, needs, and changes to overdose prevention in the last year at the local level in Europe. 35 C-EHRN focal points collected information on:

- The presence of OD prevention guidelines in official policies
- The context in which ODs are occurring (drugs involved and characteristics of OD cases)
- Challenges and desired improvements regarding OD prevention on the ground
- The state of trainings and campaigns for OD prevention
- The state and needs regarding naloxone access
- The state and needs regarding OST access.

As compared with the monitoring report of 2019, data on DCRs or OD prevention in prisons have not been included in this 2020 report; this data is already collected and reported elsewhere (13,14). Another difference to the 2019 report is that the focus of the 2020 report is mostly on the city - instead of the national level. This has allowed for more detailed qualitative information to be collected about how policies are implemented (or not) at the ground level. Such in-depth information is crucial to inform policy makers for better planning, and NGOs for advocacy purposes.

OD prevention in official policies

In 25 of the 35 countries where FPs responded the survey, OD prevention is mentioned in at least one official policy document. Most of the time, guidelines are set at the national level, but specific interest groups and associations also have guidelines for OD prevention. In fewer cases, guidelines exist at the regional and local levels only. Yet, in at least seven countries (Belgium, Croatia, Hungary, Poland, Romania, Slovenia, and Sweden), OD prevention is not yet featured in any official policy documents. Figure 1 shows a percentage comparison, and Table 1 the reporting that FPs gave



for their respective country, with total numbers per category. FPs from Austria²⁴, Georgia²⁵, Italy²⁶, North Macedonia²⁷, and the UK²⁸ mentioned a few specificities. In 2019, "national strategy", "protocol", or "separate strategy" were given as options to respondents on existing guidelines for OD prevention. Since more detailed categories were included in 2020, strict comparisons have not been made. Nevertheless, it is possible to say that with regards to the overall existence of guidelines for OD prevention, no significant changes seem to have occurred in 2020 when compared to 2019. Figure 32: Existing guidelines for overdose prevention



Country	FP City	National guidelines	Regional guidelines	Local guidelines	Specific groups have guidelines	In national drug strategy	No guidelines
Albania	Tirana					x	
Austria	Vienna						x
Belgium	Antwerp						x
Croatia	Rijeka						х
Cyprus	Nicosia					x	
Czech Re- public	Prague	x	x	x		x	
Denmark	Copenha- gen						
Estonia	Tallinn	x					
Finland	Helsinki			х	х		
France	Paris	х				х	
Georgia	Tbilisi	х			х		
Germany	Berlin			х			
Greece	Athens	х					
Hungary	Budapest						x
Ireland	Dublin	x					
Italy	Milan/Ro- mea		x	x			
Lithuania	Vilnius	х					
Luxembourg	Luxembourg	х			х	х	
Norway	Kristiansand	х					
Poland	Krakow						x

Table 9: Existing guidelines for overdose prevention per country

Country	FP City	National guidelines	Regional guidelines	Local guidelines	Specific groups have guidelines	In national drug strategy	No guidelines
Portugal	Porto				x		
North Ma- cedonia	Skopje				x		
Romania	Bucharest						x
Russia	Saint Peters- burg				x		
Scotland	Glasgow	х	х	х	Х	x	
Serbia	Novi Sad					x	
Slovakia	Bratislava					x	
Slovenia	Ljubljana						x
Spain	Barcelona	х	х	х		x	
Sweden	Stockholm						x
Switzerland	Bern					x	
Netherlands	Amsterdam					x	
Ukraine	Kiev	х			x	x	
United King- dom	London						
TOTAL		12	3	5	7	12	10

What ideal guidelines should contain

In an open question, C-EHRN FPs were asked to mention important issues missing in existent guidelines, and features that ideal guidelines should address. The results are as follows:

- Provision of Naloxone and Take Home Naloxone
- Low-threshold provision of OST
- Continuum of care and follow-up
- Coordination of continuous training for OD prevention
- OD prevention guidelines for non-opioids (stimulants, NSP, GHB, synthetic cannabinoids and polydrug use)
- Provision of services, such as DCRs and drug checking
- Assure emergency rescue access
- Being updated, to include evidence-base and recent knowledge

- C The guidelines are not always up-to-date. There is no current information about knowledge over recent years. (FP Austria)
- We still don't have Naloxone!!! Not in services, and not for take away.
 (FP Prague, Czech Republic)
- C There is very little funding, and there is no structure set for maintaining the naloxone training. Guidelines should include continuous training coordination. (FP Copenhagen, Denmark)
- **C** There are precise guidelines for the prevention of opioid overdoses, [but] there are no specific guidelines for the prevention of overdose from other psychoactive substances. (FP Tallin, Estonia)


- Guidelines are focused on very traditional (opiate use) forms of harm reduction. Inclusion of OD prevention among polydrug users, stimulant users, GHB users, and inclusion of services such as drug checking, would be an improvement. (FP Amsterdam, Netherlands)
- C The guidelines are for take home naloxone only. The main problem with the guidelines is that they require a medical institution and a medical provider to participate in the prescription and distribution of take home naloxone via low threshold centres which are not licensed medical care institutions. Low threshold centres need to enter into a cooperation agreement with a licensed clinic and the clinic's doctor must come to the low threshold centre to prescribe and hand out naloxone. The prescription is done anonymously (no personal data disclosed), but the doctor must see the PWUD. Also, medical clinics are not paid to do this. (FP Vilnius, Lithuania)
- Cover 50% of fatal ODs occur amongst those who have not been in contact with treatment services in the last five years and the various guidelines do not specifically address this issue, services often have high thresholds in relation to accessing OST with many requiring people to remain on daily/supervised consumption for prolonged periods. This has changed under COVID-19, but in our view this is a significant barrier to accessing OST which could act as a protective factor. (FP London, UK)
- Guidelines should include that an ambulance MUST come when someone calls because of an overdose, and should allow take-home naloxone.
 (FP Novi Sad, Serbia)

Overdose context

Drugs involved in ODs

FPs were asked how frequently they have heard of overdoses involving a list of specific drugs in their city in the last year. Results can be seen in Figure 33.

Altogether, heroin, fentanyl and other synthetic opioids make up almost half of the overdoses FPs have frequently heard about in their cities during 2020. Most of these opiate-related overdoses were linked to heroin, with a very small proportion related to fentanyl. Stimulants such as cocaine, crack cocaine and methamphetamine were mentioned as being involved in frequent overdoses by 20% of FPs.

In several cases, overdoses that FPs had heard of involved the use of multiple substances (category "others" in the graph). Other frequently mentioned substances included benzodiazepines, methadone, alcohol and other amphetamines, often in combination with other drugs. Less mentioned substances included synthetic cannabinoids and GHB.

Such findings follow a similar trend of the latest data collected by the EMCDDA on overdose deaths (1). Most overdose deaths were linked to the use of opioids, especially injected heroin, with a substantial number of deaths involving polydrug use, particularly heroin in combination with other central nervous system depressants, such as alcohol or benzodiazepines. Stimulants are implicated in a smaller number of overdose deaths in Europe, and often in combination with other substances.



Figure 33: Overdoses the 35 FPs heard of involving the following drugs in their city in the past year

Characteristics and circumstances of OD victims

In an open question, FPs were asked to describe the typical characteristics of OD victims that they know of, and the circumstances of their deaths. The most frequent characteristics related to ODs known to C-EHRN FPs were:

- Being in a situation of homelessness
- Using drugs alone (in a private setting or on the streets)
- Engaging in polydrug use
- Being recently released from prison, drug treatment, or other health treatment involving drug abstinence (such as in detoxification units in hospitals)
- Lacking proper nutrition and sleep
- Not calling for help/emergency for fear of the police
- Not having access to naloxone
- Comparison of the several circumstances, such as the release from prison or therapy, people who inject drugs or use multiple drugs run a much higher risk of having an overdose. So do our clients. Most ODs happen in private house-holds. They are typically not in OST and died in their own flat. OD is frequently caused by combined ingestion of Morphine/Heroin, Benzodiazepines and Alcohol. (FP Vienna, Austria)

- Content of the state of the
- Coccasional meth overdoses are always among MSM and characterised primarily through episodes of multiple days of meth use and no sleep. GHB overdoses happen in the party scene, among MSM and among marginalised polydrug users. (FP Amsterdam, Netherlands)
- Currently in Kiev, street methadone prevails and is mostly injected. We know of deaths when a person drank alcohol on top of it, fell asleep, and just died during sleep. Often people mix different psychoactive substances. People use alone and have no other people to help. They are also afraid to call an ambulance and do not have access to naloxone. (FP Kiev, Ukraine)

Having access to (Take Home) Naloxone is crucial for preventing opioid OD deaths. Having access to harm reduction information about polydrug use is essential and should be further developed. Finally, access to DCRs and residential DCRs (shelters or social housing where drug use is allowed) are strong protective factors against OD, especially for those using alone or in a situation of homelessness. DCRs are protective not only for people who use opioids, but also those using stimulants and other types of substances. Besides offering a safe and hygienic space for drug consumption, DCRs also help in combating isolation and improving community care (15). Such protective factors are also noted by C-EHRN FPs.

C Since the implementation of DCRs in the central city, most ODs are non-lethal and treated by the nurse staff according to the local guide-lines. We had some ODs in our local homeless shelters but all staff are trained in naloxone treatment.

(FP Copenhagen, Denmark)



Challenges in OD prevention

C-EHRN FPs were asked (open question) about the main challenges regarding OD response in their cities in the last year. To them, the main difficulties relate to the lack of (low-threshold) access to life-saving OD prevention programmes such as:

- Opioids (and stimulant) Substitution Treatment
- Naloxone
- Take Home Naloxone
- Drug checking
- DCRs
- OD prevention in prison and upon release from prison

Interestingly, as Chapter 2 shows, the European region is in a privileged position worldwide in terms of the availability of these harm reduction services: 91% of the countries in Europe has at least one operational OST programme and 86% has at least one OST operational in a prison. Numbers for DCRs, though, are much lower (23%), but still, Europe has most of the DCRs available in the world (14). Yet, on the ground, having at least one service is far from sufficient to handle the needs of PWUD throughout an entire nation. Furthermore, there are disparities within Europe in terms of service availability and the legal possibility of having, for instance, DCRs and drug checking.

Another crucial challenge relates to the stigmatisation and criminalisation of PWUD, including the fear of police intervention when calling an ambulance, and the refusal of ambulances to assist PWUD. This shows that, despite the general support for harm reduction in the European region, much needs to be improved in terms of supporting and securing the human rights of PWUD.

(C The biggest challenge is the lack of services aimed at OD prevention. We don't have Take-Home Naloxone programmes (Naloxone is only administered in the ER), no drug checking, no drug consumption rooms.

- 66 These aren't recognised as relevant by the Government, only needle exchange is. (FP Rijeka,Croatia)
- 66 We need to increase the OST capacity. And we also need to open more possibilities of substitution treatment for methamphetamine. (FP Prague, Czech Republic)
- Solution of the second seco
- 66 The ambulance just does not want to come [for PWUD].
 (FP Bratislava, Slovakia)
- (C The main challenge for this year in the prevention of overdoses would be the opening of the 24-hour supervised consumption rooms. As various studies have described, there is a clear link between DCRs and a decrease in mortality from overdoses. Another challenge would be to facilitate the acquisition of naloxone vials, for example, delivered free of charge in pharmacies or primary care centres. (FP Barcelona, Spain)
- Some problems are the political objection to DCRs, limited access to naloxone, and restrictive prescribing practices of OST. In addition, services can be quite paternalistic, not meeting people where they are at, coupled with a history of trying to encourage/make people reduce the OST dosage they are on. In some instances, police being called thereby reduces the likelihood of OD being reported. (FP London, UK)

Overdose prevention campaigns

Despite all the challenges and the rising number of overdoses, OD prevention campaigns are still not sufficiently put forward. In almost 60% of C-EHRN FP cities, there were no OD prevention campaigns in 2019. Only a quarter (26%) offered campaigns for OD prevention in general and/or for prevention of opioid OD. In very few cases, OD prevention campaigns approached the use of stimulants or NPS (6% each) and, in even fewer cases, the use of synthetic opioids (3%) (Figure 34).

Figure 34: Overdose awareness campaigns in the city in the past year (of 34 FPs)



Figure 35: Overdose response training in the city is available to:



Overdose prevention training

In more 29 of 34 FP cities, OD response training is available, targeting medical staff, harm reduction staff and people who use opioids. In less than a third of cities, OD prevention training also reaches friends and family of PWUD and, in only about 20%, people who use drugs other than opioids. Other possibilities include first aid training (to teachers, fire department, and company personnel) with techniques can potentially be used in an OD, and police training to deliver naloxone. Yet, in 7 of the cities, there is currently no OD prevention training, although information on OD prevention is somehow available. This was the case in Tirana, Rijeka, Athens, Budapest (where also no information is available), Porto (Portugal), Novi Sad, and Ljubljana.



Table 10. Overdess response	Au	an carllada iliber ya a ya a iber	
Table 10: Overdose response	training	availability per city	

Country	City	Harm Reduction Staff	Medical Staff	PWU opioids	Friends and family	Other PWUD	No training, but info
Albania	Tirana						x
Austria	Vienna	х	х	х		Х	
Belgium	Antwerp	x		х			
Croatia	Rijeka						x
Cyprus	Nicosia	х	Х				
Czech Re- public	Prague	x	x	x			
Denmark	Copenhagen	х	х	х			
Estonia	Tallinn	х	Х	х	x	Х	
Finland	Helsinki	х	х				
France	Paris	х	х	х	x		
Georgia	Tbilisi	х		х	x	Х	
Germany	Berlin	х	Х	х		Х	
Greece	Athens						x
Hungary	Budapest						х
Ireland	Dublin	х	х	х	x		
Italy	Milan/Rome		Х	х			
Lithuania	Vilnius	х	х				
Luxembourg	Luxembourg	х	x	х			
Norway	Kristiansand	х	x	х	x		
Poland	Krakow	х		х		x	
Portugal	Porto						x
North Macedonia	Skopje	x	x	x			
Romania	Bucharest		х				
Russia	Saint Petersburg			х		Х	
Scotland	Glasgow	х	х	х	x	х	
Serbia	Novi Sad						х
Slovakia	Bratislava		х				
Slovenia	Ljubljana						x
Spain	Barcelona	х	х	х	х	х	
Sweden	Stockholm			X ²⁹			
Switzerland	Bern	х	х	х			
Netherlands	Amsterdam	х	х				
Ukraine	Kiev	х		х	Х		
UK	London	х	x	х	x		

Desired improvements in overdose prevention

For most C-EHRN FPs, overdose prevention activities have not improved in their cities in the past year.

Figure 36: Have activities on OD prevention in your city improved in the past year?



In an open question, FPs described the main changes in OD prevention activities that they would like to see happening in their cities. These are as follows:

- Increased and low threshold access to Naloxone and Take Home Naloxone (without the need for prescription and free of charge for PWUD)
- Increased and continuous training on OD prevention, also for non-opioids, for harm reduction staff, PWUD and their contacts
- Campaigns and information on OD prevention for the general population
- Funding for Naloxone and OD prevention training
- Opening and scaling-up of DCRs and drug checking services
- Less restrictive prescribing practices for OST or HAT (Heroin Assisted Treatment)
- Attention to OD prevention for non-opioids
- Sensitise health professionals and first responders to fight stigma against PWUD.

- It would be desirable that naloxone is available for convicts, relatives of drug users, social workers, police-officers... (in all parts of Austria) paid by health insurance. (FP Vienna, Austria)
- 66 More services (specifically better availability of Naloxone), more training on overdose prevention in the general population and especially medical staff and people working with PWUD. (FP Rijeka, Croatia)
- 66 More training for harm reduction staff as well as provision of take home Naloxone without prescription or other requirements. (FP Nicosia, Cyprus)
- We need training on overdoses of substances other than opioids and synthetic opioids (how to prevent, how to help until an ambulance reaches the victim, what to do). (FP Tallinn, Estonia)
- 66 We wish for the introduction of Naloxone in Helsinki, the ability to use and give out Fentanyl-strips to PWUD, and establishing safe injection sites. (FP Helsinki, Finland)
- We need a broad campaign about overdose, including the general population, not only targeted groups. Trainings on Naloxone and OD management is needed for medical staff (besides first aid and hospitals). Also, police should know how to respond to overdose and should be equipped with Naloxone. (FP Tbilisi, Georgia)
- 66 We need the removal of barriers for take home naloxone and steady funding for take home naloxone distribution. (FP Vilnius, Lithuania)

- C Provide more information about OD prevention for PWUD and have naloxone available for everyone. Also, change the attitudes of medical and emergency ambulance staff toward PWUD. Sometimes PWUD prefer not to call an ambulance because of the attitude and stigma they face. (FP Bratislava, Slovakia)
- Content of the possibility to get Nalaxone for free in city pharmacies and clinics. Conducting a campaign for relatives of people dependent on drugs about measures that can be taken in case of an overdose. Medical training, as an ambulance often refuses to come in the case of an overdose. Availability of nasal naloxone, and overdose programmes for people who use stimulants. (FP Kiev, Ukraine)

FP context compared to national context

Almost half of the participants assessed that OD prevention in their cities is comparable to the national situation in their respective country. Also, almost half think that their city offers better OD prevention when compared to the national context. This shows that the OD prevention context described by C-HERN Monitoring is in good part based on the best examples available in a country. Since most FPs are based in a metropolis or a large city, it can be assumed that the current data might not reflect the context of smaller cities and rural areas. Figure 37: How does your city compare with the national situation in terms of OD prevention?

Correlation European

Harm Reduction Network



OD prevention for opioids

Naloxone availability

Naloxone is available in the majority (80%) of FP cities. Yet, in at least 6 cases, the life-saving drug was reported as not available. This was the case in Antwerp, Prague, Helsinki, Athens, Budapest, and Bucharest. The FP in Luxembourg was not aware of the current situation. According to the EMCDDA, the medication naloxone is included in the pharmacopoeia of all European countries, although in many cases only in injectable form and requiring a medical prescription (16). This lack of practical availability might be the reason why FPs mention naloxone as not being available in their cities.

Figure 38: Is naloxone available in your city?



To whom is naloxone available

For those 27 FPs reporting the availability of Naloxone, the drug is mostly available to medical staff at hospitals (93% of the cases), ambulances (85% of the cases), or medical staff in harm reduction services (67%). In about 60-70% of cases, it is also available to harm reduction staff and directly to PWUD. Only in 40% of cases is it available to family and friends of PWUD.

Figure 39: If available, who has access to naloxone?



Table 11:	How	naloxone	is	available	in	FP	cities
-----------	-----	----------	----	-----------	----	----	--------

How naloxone is available

When available in FP cities, Naloxone is mostly found in its injectable form (61%), although intranasal is also available in 50% of the cases. Training is available, both for peer administration (61% of cases) and staff administration (57%). Slightly more than half of respondents mentioned that Naloxone is available for take home (54%) and/or distributed by drug service providers (54%). Nevertheless, only in a few of the FP cities is naloxone reimbursed by health insurance (21%), or available in pharmacies without prescription (18%).

Table 11 shows the availability of naloxone reported per city (for the cases where availability was reported). It is interesting to note that according to the EMCDDA, only 12 countries have Take-Home Naloxone (THN) available: Austria, Denmark, Estonia, France, Germany, Ireland, Italy, Lithuania, Norway, Spain, Sweden, and the United Kingdom (including Scotland) (2,16). Despite this, FPs in the cities of Vilnius and Kristiansand did not report availability of TNH. On the other hand, FPs in Nicosia, Tbilisi and Kiev reported the drug to be available for take home.

Country	City	In-	Injec- tion	Training staff admin	Training peer admin	Take- home	In drug services	Paid by health insurance	Pharmacies without pre- scription
Albania	Tirana				x				
Austria	Vienna	x		x	x	х	х		
Croatia	Rijeka		х	x				х	
Cyprus	Nicosia	х		x		х	х		
Denmark	Copenhagen	х		х	х	х			
Estonia	Tallinn	х		х	х	х			
France	Paris	х	х	x	x	х	х	х	х
Georgia	Tbilisi		х	х	х	х	х		
Germany	Berlin	х		х	х	х		х	
Ireland	Dublin	х	х	х	x	х	х		
Italy	Milan /Rome	х	х			х	х		х
Lithuania	Vilnius			х			х		
Norway	Kristiansand	х			х		х		
Poland	Krakow		х						



Portugal	Porto	х	х						x
North Mace-	Skopje		x						
donia									
Russia	St. Petersburg		x		х		x		
Scotland	Glasgow		x	x	х	x	x		
Serbia	Novi Sad		x					х	
Slovakia	Bratislava		x						
Slovenia	Ljubljana	х							
Spain	Barcelona		x	x	х	x	x		
Sweden	Stockholm	х		x	х	x		х	
Switzerland	Bern	х	x	x			x	х	x
Netherlands	Amsterdam		x						
Ukraine	Kiev		x		х	x	x		
UK	London	х	Х	x	х	Х	x		

Challenges in naloxone availability

Despite reported availability, several challenges remain in gaining access to Naloxone. According to FPs (open question), the main challenges in their respective cities during the past year include:

- No access to Take Home Naloxone
- Naloxone available only at drug services, thus, not for all PWUD
- Need for medical prescription
- Administration by medical staff only
- Lack of insurance coverage for Naloxone
- Lack of funding for Naloxone
- Lack of funding and support for Naloxone training

Examples of such challenges given by C-EHRN FPs include the following:

- Solution (Content of the second se
- Solution of the second staff and at the hospital. (FP Bratislava, Slovakia)

- Solution of the state of the
- Solution of the second seco
- Solution of the second state of the second
- Conly medical staff in ERs have access to Naloxone and it can be administered only when a person is brought into the ER because of an OD.

(FP Rijeka, Croatia)

- (C In Austria, injectable naloxone is only available for medical staff, and handing out Naloxone requires a private prescription by a physician. This prescription refers to opioid addicts. The Naloxone will be handed out in a pharmacy. Clients who participate in the THN training get the prescription from a low-threshold medical institution for people with no insurance (Caritas Marienambulanz) with which we cooperate. After attending the training (which lasts around one hour), the participants get handed out the naloxone kit. (FP Vienna, Austria)
- (Intranasal naloxone is limited due to cost implications. Peer-to-peer supply is limited due to a lack of peer initiatives across London and England. (FP London, UK)
- In Denmark, funding from the state is given for the provision of naloxone to drug users in treatment. There is no systematic funding for provision and follow-up to people outside the treatment system.
 (FP Copenhagen, Denmark)
- Solution (Section 1) Naloxone without prescription is only accessible at harm reduction sites.
 (FP Tbilisi, Georgia)
- Cour main challenges are availability in pharmacies and in prison settings; training and advocacy among professionals; and the costs of the treatment. (FP Paris, France)
- In Ireland, there are relatively more restrictive pathways for accessing Naloxone. This requires a trained keyworker to initially conduct a risk assessment and to educate the client about Naloxone and train them, or their relatives, on how to administer either or both the nasal and injectable forms of Naloxone. Once this is com-

pleted, the client requires a doctor (usually their own GP, a GP working in specialised homeless services, or an OST addiction prescriber) to issue a prescription for Naloxone. Due to the scheduling of Naloxone in Ireland, the person to whom it is prescribed must not give the Naloxone to anyone else to hold for them. However, the HSE do allow for GPs to issue prescriptions retrospectively, within a 24 hour period, to allow, in particular, for the administration of Naloxone in an overdose scenario. (FP Dublin, Ireland)

G By order of the Ministry of Health, two ampoules of Nalaxone can be bought in a pharmacy without a prescription, but, in practice, they often refuse to sell without a prescription. When we heard about overdoses and tried to provide Nalaxone, we were informed that it was impossible to get it anywhere else but with us. (FP Kiev, Ukraine)

What needs to improve

C-EHRN FPs, were also asked to openly comment on the main changes needed regarding naloxone availability in their cities. These include the following:

- It must be available for friends and relatives of PWUD
- It must be available for take home
- It must be available in prisons and other custodial settings
- It must be available at in- and out-patient therapeutic institutions
- It must be available in pharmacies
- It must be increasingly available through peer distribution
- It must be paid by health insurance
- Access must not require a medical prescription
- Legal barriers for administration by non-medical staff and PWUD peers must be removed
- Training for peer administration must be provided



Changes between 2019 and 2020

Policy change in process

Over 60% of C-EHRN FPs reported not being aware of any policy change in process to increase access to naloxone in their city.

Those reporting policy changes in process were in Vienna (Austria), Nicosia (Cyprus), Paris (France), Norway, Porto (Portugal), Poland (at national level), London (UK), and in Kiev (Ukraine).

In Nicosia (Cyprus) take-home Naloxone programmes were under discussion in 2019 and finally became available in 2020.

More effort is done to provide naloxone to opioid users through drug treatment centres. In addition, efforts are being made so that take home Naloxone is available as an over-thecounter medicine. (FP Nicosia, Cyprus)

In Vienna (Austria) and Tbilisi (Georgia), naloxone has started being distributed by drug service providers. In Berlin (Germany), it has started to be prescribed to opioid users.

In Kiev (Ukraine), a limited amount of Naloxone is now available at pharmacies without prescription, and the lifesaving drug is also part of the first aid kit of police officers. In Vilnius (Lithuania), there are no changes yet, but there is a plan for police officers to carry intranasal naloxone.

Changes have been made to the order of the Ministry of Health, according to which Nalaxone can be bought in pharmacies of the city without a prescription in the amount of two ampoules. Changes have been made to the list of components of the policeman's first-aid kit and Nalaxone has been introduced there (even nasal, but it is not available in our city and the country).

(FP Kiev, Ukraine)

Figure 40: Is there a policy change in process to increase access to naloxone in your city?



The UK started discussions to expand Naloxone provision beyond drug treatment services, possibly including facilities dedicated to people experiencing homeless.

C Policy for the distribution of Naloxone is a national one – UK-wide – as naloxone is prescription only but can be legally supplied outside a prescription by 'drug treatment services'. There are discussions at a national level to expand this definition to other relevant settings, such as homeless services, but this is at a very early stage.
 (FP London, UK)

In Portugal, Naloxone was available to harm reduction providers in 2019 but only at the mobile DCR in Lisbon. In 2020, nasal naloxone was made available for a couple of months to harm reduction teams across the country, but the medicine expired in April 2020 and, therefore, teams are waiting for their new allocation. Injectable naloxone was available for a couple of months in 2020 for the use of HR teams.

C The National Drugs Agency (SICAD) will provide another allotment of nasal naloxone to HR teams across the country. They don't oppose naloxone being distributed to clients, family and friends, but there is no guidance or training available. (Porto, Portugal FP) In Paris (France), Naloxone started being reimbursed by health care insurance. Moreover, a national training platform for overdose prevention was created by a civil society-led initiative with support from the French government. Read more information on this case in Box 1.

Case 1: OD prevention national training in France

Since 2018, Fédération Addiction and its partners,-AFVD, Aides, APSEP, ASUD, association of general medicine, the college of therapeutic communities, ELSA France, , OFMA, Psychoactif, RESPADD, SAFE, SFETD, and SFSPO - have been involved in a working group on OD prevention supported by the French government. The group aims at encouraging people who use opioids, their contacts, and professionals to use appropriate strategies to reduce the risks of lethal opioid overdose. The group is coordinated by Dr Nicolas Authier, and composed of multidisciplinary professionals (pharmacists, doctors, first aid instructors, and harm reductionists).

In 2020, this civil society-led group set up a national training plan for OD prevention through the online platform <u>www.naloxone.fr</u>. The training is intended for people who use opioids, those who are in OST, as well as family and friends of users and professionals. The 7-step training includes:

- 1. Knowing essential information on opioid overdose
- 2. Protecting victim and witnesses
- 3. Calling emergency services
- 4. Administering naloxone-based medicine
- 5. Freeing respiratory tract and taking stock of the victim's condition
- 6. Putting victim in recovery position
- 7. If necessary, doing cardiac massage

Each stage of the training is composed of a video clip, accompanied by a recap and a quiz to evaluate knowledge gained. The platform also contains information such as epidemiological data, legal texts, and presentation of Naloxone-based medicines, among others.

It is also possible to perform a general test to obtain a training certificate. For individuals, the certificate enables them to verify new knowledge acquired with the training. For harm reduction workers, the certificate is considered sufficient and may be required to distribute Naloxone. In addition to the website, the group has produced a poster and a flyer for OD prevention. The poster describes four contexts in which Naloxone can be used in the event of an opioid OD, and it can be displayed in pharmacies and drug-related services, for instance. The flyer describes how to react if someone is having an opioid OD.





FP context compared to national context

More than half of respondents assessed that Naloxone availability in their cities is comparable to the national situation. 34% think that their city offers better coverage when compared to the national context, and 6% that it offers lower coverage. Similar to general OD prevention, the context described by the C-HERN Monitoring for Naloxone availability is in good part based on the best examples of availability in a country and, thus, current data might not reflect the context of smaller cities and rural areas.

Figure 41: How does your city compare with the national situation in terms of Naloxone availability?



Opioid Substitution Therapy (OST)

Medications available for OST

In most C-EHRN FP cities, both methadone (97% of the cases) and buprenorphine (88%) are available for OST, with methadone only (Bucharest, Romania) or buprenorphine only (Nicosia, Cyprus) being available in a few cases. When available, medical heroin (21%) and morphine (12%) complements the availability of methadone and/or buprenorphine. Medical heroin is available in Amsterdam (Netherlands), Bern (Switzerland), Glasgow (Scotland, since last year), Kristiansand (Norway), Luxembourg (Luxembourg), Berlin (Germany), and Copenhagen (Denmark). Morphine is available only in Berlin, Copenhagen, Vienna, and Bern, and in a few cases also in Kristiansand (Norway).





Other available medications for OST include:

- Suboxone (Budapest, Hungary and Ljubljana, Slovenia)
- Substitol (Ljubljana, Slovenia, Berlin and other cities/Germany)
- Diacetylmorphine (Berlin, and approx.. 9 other cities in Germany)
- Levomethadone and codeine (Berlin, Germany)
- Methylphenidate for peple who use methamphetamine (Prague, Czech Republic).

In St. Petersburg (Russia), OST is prohibited, and in Bratislava (Slovakia), the real availability of buprenorphine is limited: In theory, buprenorphine treatment should have been available since 1999 and in combination with Naloxone since 2008. But the reality is that since 2019, people on buprenorphine treatment don't have access to it because there is a lack of these pills in the Slovak market due to withdrawal of one distributor from the Slovak market last year. This medication is not registered as necessary, so Slovakia/ Ministry of Health doesn't need to secure the availability of this treatment. (FP Bratislava, Slovakia)

OST accessibility

Figure 43: What factors limit OST access in your city? Please rate them according to the level of the barrier they represent to achieving sufficient OST coverage.



Several factors limit OST accessibility in FP cities.

The main limiting factors are stigmatisation of PWUD and the high threshold to enter, or remain in, treatment. The thresholds considered most problematic are urine testing, the requirement that people abstain from using illegal drugs, the need for documentation, and the limited hours of service delivery. Other problematic thresholds mentioned by more than half of respondents include the lack of OST prescribers, long waiting lists, requirements for social coverage or medical insurance, and a requirement that people participate in meetings. Legal and age restrictions, an inadequate supply at pharmacies, and the costs of treatment, are other limiting factors. Figure 43 shows how FPs rated several pre-selected barriers to OST access.

What needs to improve?

C-EHRN FPs were aked to openly comment on the most needed improvements regarding OST access in their citities. Their main responses include:

- Increased OST coverage, especially in smaller cities
- Increased number (and options) of prescribers
- More options for medicines to be used as a substitute
- Lower threshold to start and continue treatment
- OST availability in prisons and other closed settings
- More attention to counselling and social support during treatment

- (6 In North Macedonia, about 1,600 people are treated with methadone, and 250 with buprenorphine, while the size of the population of people who inject drugs is estimated at about 6,800; the scope of programmes for treatment of addictions is about 27%. (FP Skopje, North Macedonia)
- C A huge problem of substitution treatment in Poland is the lack of access to prescription substitution drugs. This means that anyone who wants to be treated this way must agree to participate in large programmes. It also means that people from small towns have no chance of accessing substitution treatment because no one opens substitution programmes for only a few people. (FP Krakow, Poland)
- Control of the second prescribe larger OST doses and not push clients to reduce doses. (FP Budapest, Hungary)
- Maintenance programmes should be implemented with methadone for injection. Another option would be heroin-dispensing programmes. The hours of access to the methadone dispensing should also be extended, even creating a 24-hour dispensing centre. (FP Barcelona, Spain)
- C Patients should have more opportunities in choosing a physician who prescribes OST. (FP Vienna, Austria)
- Cone should have more options in regard to the substance given as a substitute. (FP Nicosia, Cyprus)



Correlation European

Harm Reduction

- C The service needs more counsellors who have experience and are able to advise clients (for example, why they feel bad, how can they improve their well-being, what is happening to them) and guide them. This part of the service is currently incomplete. (FP Tallinn, Estonia)
- C The main issue is that the treatment mainly consists of the medicine and there is very little social-counselling and psychological assistance available. (FP Helsinki, Finland)
- C The waiting list that comes and goes is an issue. But we cannot forget the criminal indifference of some service providers regarding the beneficiaries, and the huge level of stigma towards PWUD from the psychiatrists that usually run the substance-free treatment units. (FP Athens, Greece)
- It is necessary to expand the list of drugs for OST, and reduce the level of stigma, as well as introduce a more loyal attitude towards co-consumption. Support programmes need to be strengthened and we must start researching programmes of substitution therapy for stimulants.
 (FP Kiev, Ukraine)

Changes between 2019 and 2020

For the vast majority of participants (70%), OST has not improved in their city in the past year. The FPs reporting improvement were from Tirana (Albania), Vienna (Austria), Copenhagen (Denmark), Dublin (Ireland), Bern (Switzerland), Amsterdam (Netherlands), and Kiev (Ukraine). When present, most improvements occurred due to adjustments made to respond to the COVID-19 pandemic. It is, therefore, not clear if those will continue.

- COVID-19 pandemic, the waiting time for starting OST (Methadone) fell from 12to-14 weeks to 2-to-3 days. (FP Dublin, Ireland)
- Up till recently, European migrants, particularly those from Eastern Europe, had no access to methadone treatment. Due to COVID-19, a pilot started, allowing 17 European migrants access to methadone treatment. This pilot started because they had a harder time accessing heroin and methadone from the black market. It is not certain how this pilot will develop post-corona. Also, due to corona, people were given more weekly prescriptions (rather than having to pick up their methadone daily). This gives people more trust and freedom. On the other hand, we have heard of some people who miss the regular contact with their care. (FP Amsterdam, the Netherlands)

FP context compared to national context

About 60% of FPs consider that their city has better OST coverage when compared to the national situation. The main reason mentioned for the best coverage is that FP cities are bigger and, thus, count more services. In smaller and rural areas, OST coverage is compromised. The city situation was regarded as similar in 12 cases: Croatia, France, Georgia, Greece, Hungary, Italy, Norway, Poland, Spain, Switzerland, the Netherlands, and the UK. Nevertheless, even in these cases, a few differences occurred in terms of more, or more specialised, services in bigger cities.

- In other cities the situation regarding access to OST is a lot worse; there are long waiting lists for the treatment because of the lack of prescribers. Vienna has the best supplementation with OST. 85% of prescriptions are done by GPs (very easy to get access). Medication is delivered by pharmacies and there are useful take home regulations in terms of reintegration. (FP Vienna, Austria)
- Getter access to substitution treatment is available only in Warsaw, where there are many programmes offering this kind of treatment. Mostly bigger cities in Poland have similar access to OST, like in Krakow. (FP Krakow, Poland)

Figure 44: How does access to OST in your city compare with the national situation?





- In big cities, like Rijeka, inadequate supplies at pharmacies isn't a problem, but in smaller cities it is a big problem. The pharmacies in smaller cities often don't have OST, so it has to be specifically ordered which puts the people who use OST in a bad position where they sometimes have to wait for days for their therapy to arrive. (FP Rijeka, Croatia)
- In Prague, there are several OST programmes.
 The capacity is around half of what is needed.
 In some parts of the Czech Republic, there are no OST programmes at all, or they are far from the homes of PWID.
 (FP Prague, Czech Republic)
- C The largest cities are similar but access to medical doctors to prescribe is problematic, especially in rural communities. (FP Copenhagen, Denmark)

Conclusions

There is an urgent need for the development of specific OD prevention measures that can address the current challenges seen in the field.

OD prevention measures should be included in national drug policy strategies, action plans and guidelines. *Policies and guidelines* should be updated with recent evidence and include:

- Guidelines for low-threshold access to Naloxone (e.g. over-the-counter, without the need for a medical prescription, community-based, and paid by health insurance).
- The obligation to provide people suffering from an OD stigma and punishment-free emergency services.
- OD prevention for non-opioids (such as stimulants, NSP) and address polydrug use.

Regarding services, there is a need for:

- Setting, and scaling, up DCRs and residential DCRs (in shelters or social housing), especially given FP observations that many ODs happen to people using alone or who are homeless.
- Setting, and scaling, up drug checking services.
- Scaling-up Take-Home Naloxone programmes and an increase in low threshold access to Naloxone for PWUD and people likely to witness an overdose.
- Lowering the threshold of OST programme initiation and continuation.
- Investing in substitution treatment for stimulant drugs.
- Further developing, and providing, OD prevention for stimulants and polydrug use.

In terms of campaigns and training,

- Funding and developing targeted OD prevention campaigns, including prevention for OD with opioids, stimulants, and NPS.
- Funding and developing OD prevention campaigns for the general public.
- Sensitising health care professionals, and other first responders, to properly assist PWUD, without stigma, refusal, or punishment.

References

- 1. European Monitoring Centre for Drugs and Drug Addiction. European Drug Report 2020: Trends and Developments. Luxembourg: Publications Office of the European Union; 2020.
- 2. EMCDDA. Drug-related deaths and mortality in Europe.Update from the EMCDDA expert network. July 2019. Luxembourg; 2019.
- Minozzi S, Amato L, Davoli M. Preventing fatal overdoses : a systematic review of the effectiveness of take-home naloxone. [Internet]. Luxembourg: EMCDDA Publications Office; 2015. Available from: http://dx.publications.europa. eu/10.2810/396726
- 4. Strang J, McDonald R, Hedrich D, Simon R. Preventing opioid overdose deaths with take-home naloxone. Lisbon; 2016.
- Langham S, Wright A, Kenworthy J, Grieve R, Dunlop WCN. Cost-Effectiveness of Take-Home Naloxone for the Prevention of Overdose Fatalities among Heroin Users in the United Kingdom. Value Heal J Int Soc Pharmacoeconomics Outcomes Res. 2018 Apr;21(4):407–15.
- Winter RJ, Stoové M, Degenhardt L, Hellard ME, Spelman T, Jenkinson R, et al. Incidence and predictors of non-fatal drug overdose after release from prison among people who inject drugs in Queensland, Australia. Drug Alcohol Depend [Internet]. 2015;153:43–9. Available from: http:// www.sciencedirect.com/science/article/pii/ S0376871615003105
- Volkow ND, Wargo EM. Overdose Prevention Through Medical Treatment of Opioid Use Disorders. Ann Intern Med [Internet]. 2018 Jun 19;169(3):190–2. Available from: https://www. acpjournals.org/doi/abs/10.7326/M18-1397
- Montero-Moraga JM, Garrido-Albaina A, Barbaglia MG, Gotsens M, Aranega D, Espelt A, et al. Impact of 24-hour schedule of a drug consumption room on service use and number of non-fatal overdoses. A quasiexperimental study in Barcelona. Int J Drug Policy [Internet]. 2020;81:102772. Available from: http:// www.sciencedirect.com/science/article/pii/ S0955395920301134

- Newman M. Could drug consumption rooms save lives? BMJ [Internet]. 2019;366. Available from: https://www.bmj.com/content/366/bmj. l4906
- Kappel N, Toth E, Tegner J, Lauridsen S. A qualitative study of how Danish drug consumption rooms influence health and well-being among people who use drugs. Harm Reduct J [Internet]. 2016;13(1):20. Available from: https://doi. org/10.1186/s12954-016-0109-y
- Wallace B, Pagan F, Pauly B (Bernie). The implementation of overdose prevention sites as a novel and nimble response during an illegal drug overdose public health emergency. Int J Drug Policy [Internet]. 2019;66:64–72. Available from: http://www.sciencedirect.com/science/article/pii/S0955395919300258
- Eukel HN, Skoy E, Werremeyer A, Burck S, Strand M. Changes in Pharmacists' Perceptions After a Training in Opioid Misuse and Accidental Overdose Prevention. J Contin Educ Health Prof [Internet]. 2019;39(1). Available from: https://journals. lww.com/jcehp/Fulltext/2019/03910/Changes_in_ Pharmacists_Perceptions_After_a.3.aspx
- EMCDDA. Drug consumption rooms : an overview of provision and evidence. [Internet]. [Luxembourg]: European Monitoring Centre for Drugs and Drug Addiction; 2017. Available from: http:// www.emcdda.europa.eu/system/files/publications/2734/POD_Drug consumption rooms.pdf_en
- Harm Reduction International. The Global State of Harm Reduction 2020 [Internet]. London;
 2020. Available from: https://www.hri.global/ files/2020/10/26/Global_State_HRI_2020_BOOK_ FA.pdf
- 15. Woods S. Drug Consumption Rooms in Europe: Organisational Overview. 2014;
- EMCDDA. Take Home Naloxone [Internet]. 2020 [cited 2020 Dec 22]. Available from: https://www. emcdda.europa.eu/publications/topic-overviews/take-home-naloxone_en

7

Highlights in New Drug Trends



Introduction

The continued appearance and use of NPS on the global and European markets remains a major concern for policymakers, law enforcement officers, but also for CSOs working in the field. International sources have warned of potential health risks for quite some time (1-3). But since the number of new drugs entering the market is still high (in 2019, 55 NPS were identified for the first time within the EU; in other words, 1 new substance per week). Although the body of knowledge regarding NPS is growing, we still lack essential information about many of these substances, e.g. regarding effects, risks, etc. Given that CSOs work closely with PWUD, they are able to gather essential information about these new substances, information which is difficult to gather by, for example, scientists, or law enforcement officials. Besides NPS, civil society monitoring also included other developments regarding the use of drugs, for instance new patterns of drug use, new routes of administration, the use of known substances by a different group of PWUD or the combined use of different substances (new and/or known). So the focus of this activity includes a broader field than just the use of new drugs. New approaches in this field are needed to regularly update existing data on new drug trends and drug use patterns. Harm reduction and community organisations working closely with PWUD may see changes in drug use much more quickly than other organisations working in this field. Therefore, it is considered important, and of great value, to establish a mechanism to pick up, monitor and report on emerging drug trends at a much more rapid pace. The fact that the data collected by C-EHRN may be anecdotal, small-scale, or is appearing for a short period of time, is considered not an obstacle, but as complementary to other data sources.

Adjustments to the questionnaire and introduction of focus groups

In January 2020, C-EHRN in Amsterdam discussed in a core group the monitoring activities of the previous year with a focus on what needed improvement, such as the quality and reliability of the data gathered. As for the New Drug Trends section, several occasions during 2019 had already allowed discussions to take place on issues related to monitoring new drug trends by civil society organisations (4).

These discussions and brainstorming sessions resulted in several adjustments to the questionnaire compared to that used in 2019: a shortened questionnaire; and a focus on the local level (the city where the FP is located) rather than at a regional or national level. In so doing, the quality and reliability of the data collected will increase, and thus its interpretation. Furthermore, it was decided to not only use online questionnaires, but to also promote the idea of Focal Points organising focus groups or group discussions with local experts in the field, including PWUD. However, because of the coronavirus outbreak in March of 2020 in most EU countries, it was considered not feasible to conduct face-to-face interviews and, therefore, FPs continued to use the 'safe' online questionnaires.

After a number of reviews, the 2020 questionnaire was sent out to FPs in May, and in September analysis of the filled-in questionnaires began. The request to fill in the CSO monitoring questionnaire was sent to the 37 C-EHRN FPs, of which 33 filled in the questionnaire using Survey Monkey. Below are some of the highlights gained from the monitoring of New Drug Trends in 2020.



New drug trends in C-EHRN FP cities

New drug trends can arise in many ways: a new or unknown substance that arrives on the market; an already known drug but used by a new group of people; a new route of administration of a substance; the combined use of different substances, sustained changes in behaviour and language relating to new drug use, etc.

Emergence of new substances

The first question asked was related to the emergence of a new or unknown substance in the local market. Since it is new or unknown to the target groups or CSOs, its use holds extra risks for the user. Therefore, timely monitoring of the use of such substances, as well as gathering information on risks and effects, is essential to provide harm reduction services for users of such substances as early as possible. CSOs are the agencies closest to PWUD and are, therefore, the first who can, and should, report on such issues.

Figure 45: In the previous year, have you witnessed any new developments regarding the use of drugs in your city amongst your target group(s): (a) The emergence of a new or unknown substance?

A small group of FPs have, therefore, seen the emergence of a new substance entering the local market. This may be an indication that although many new substances are newly offered on the local market every year, most of them are not used, at least not soon after they appear on the sellers markets. It can also be the case that substances already disappear from the market even before they are detected. It is worth noting that this period, in which almost a quarter of the FPs reported new developments regarding the use of drugs in thier city, includes the period from when the COVID19 pandemic began. It is not fully understood yet the full impact of the pandemic on drug routes and drug markets in Europe.

The FP in Antwerp, Rijeka, Tallinn, Budapest, Skopje, Bucharest, Glasgow, and Stockholm reported the emergence of new substances on the local markets: these include synthetic cannabinoids (mentioned by Rijeka, Bishkek and Budapest FPs); ketamine (Antwerp); Isotonitazene (Tallinn); 2C-B (Skopje); crack cocaine (Glasgow); and oxycodone (Antwerp). It seems that most of these new drugs on the market are used by marginalised and homeless people, and/or people with mental disorders. Reasons for using these new substances include the unavailability of desired drugs, and curiosity.



This question had 33 responses: Yes: 8 FPs (24%)

Case 1

Name of new substance: "Riječki šit" ('shit from Rijeka'), also called Paški, Đefri, Škipa, Gashijev šit or Ciganski šit.

What is it? It is supposed to be a mix of everything available (synthetic cannabinoids, kerosine, rat poison, antidepressants, and ketamine.

First heard of: Around March 2020.

Reason for using: Curiosity and desire to experience pleasure.

Effects: 'it makes you feel like a child again, and everything is okay and funny; hallucinations (optical and sound)'.

Unwanted effects: Strong fear and paranoia.

Number of people using: Unknown (maybe 50-100).

Known substances newly used by your target groups

Learning about known substances being used for the first time by the target group of CSOs may indicate, for example, changes in the availability or quality or price of the drugs they usually take. This may be of particular interest to CSOs working in this field, since shifting from one substance to another may lead to changes in drug patterns, risk behaviours and possible unknown health risks. Timely and up-to-date information on this phenomenon may inspire CSOs to rapidly address any negative consequences of use of these drugs by the target group. Figure 46: In the previous year, have you witnessed any new developments regarding the use of drugs in your city amongst your target group(s): (b) The emergence of a known substance but used for the first time among (one or more of) your target group(s) in your city (e.g. GHB use among people who traditionally only used heroin)? (Please mention only the most remarkable or worrying changes).

This question had 33 responses: Yes: 11 FPs



Two-thirds of respondents mentioned no new developments regarding the use of known substances for the first time by any of their target groups.

FPs in Vienna, Antwerp, Nicosia, Tallinn, Krakow, Porto (Vila Nova de Gaia), Glasgow, Barcelona, Stockholm, Amsterdam, and London mentioned the emergence of a known substance by a new target group. Examples include: young people starting to use benzodiazepines in Vienna; ketamine in Glasgow; and in Stockholm, young people have started to use NPS; synthetic cathinones in Tallinn, or synthetic cannabinoids in Krakow by street opioid users; methamphetamine use by heroin/crack cocaine users in London; 3-MMC in chemsex scenes in Amsterdam and GHB in LGBT scenes in Porto; 'Shabu' (methamphetamine) by Asian immigrants.



Case 2:

Known substance used for the first time by target group

What substance: synthetic cathinones ('bath salts').

Used now by: Street opioid users.

Since when: First quarter of 2019.

In what form: powder.

Route of administration: mostly injecting.

Reason for using: Unavailability of street fentanyls.

Source: NGO Convictus, Tallinn (Estonia).

New routes of administration

Known PWUD using new or different routes of administration may also be a sign of changes in the quality of drugs they use, or may reflect a shift from recreational use to more problematic use. As such, it may also involve extra health related risks, e.g. when PWUD shift from smoking cocaine to injecting. Quick interventions, based on this monitoring, may minimise harm done to these PWUD.

Figure 47: In the previous year, did you witness in your target group(s) the emergence of a new or different route of administration of specific substances?

This question had 33 responses: Yes: 7 FPs.



Examples reported are young people sniffing MDMA instead of oral intake (Vienna); OST injecting (methadone (heptanon), buprenorphine) (Rijeka); Oxycodone bought online and injected or snorted (Helsinki); 'bio-drugs' taken orally; crack cocaine by means of smoking (Tbilisi); opioid by injection (Stockholm); and chemsex drugs among MSM (Amsterdam).

Case 3:

PWUD using different routes of administration

By whom: People who inject drugs, people who are homeless.

What substance: Cocaine.

Route of administration: Injected, or smoked if crack.

Since when: Over the last year or two this has increased.

Reasons: Increased availability, enhances effects of other drugs that are used.

Source: Scottish Drugs Forum, Glasgow (UK).

New combinations of substances

Combined use, or the use of more than one psychoactive substance during one session, always includes additional health risks as the effects may be much different from what is expected. Being aware at an early stage of a trend like this may help CSOs to develop early interventions to support PWUD.

Two FPs mentioned the combined use of ketamine and cocaine (Glasgow and Vienna). Other combinations mentioned include: cannabis and amphetamines (Tbilisi); methadone and amphetamines (Barcelona); and combinations of various fake and real benzodiazepines, as well as the combined use of Lyrica (gabapentin) and opiates (Stockholm). Figure 48: In the previous year, did you witness in your target group(s) new combinations of substances?

This question had 33 responses: Yes: 5 FPs, Vienna, Tbilisi, Stockholm, Glasgow, Barcelona.



Changes in the target groups

Figure 49: In the previous year, did you witness any changes in the existing target groups you provide services for (e.g. younger, new immigrant groups)?

This question had 33 responses: Yes: 5 FPs.



Examples include: OST users experience more problems in everyday life (Nicosia); the MSM chemsex scene is increasing (Budapest); a decrease in NPS use in favour of increased use in pharmaceutical opioids (Krakow); immigrants returning home from other EU countries (Bucharest); and immigrants inhaling opioids (Stockholm).

New target groups

FPs in Antwerp, Rijeka, Nicosia, Tbilisi, Athens/Thessaloniki, Krakow, Porto, Bucharest, Glasgow, Belgrade, Barcelona, and Stockholm started services last year for new groups of PWUD. These include: MSM; non-EU immigrants; migrants; students from Asia; young people using non-injectable drugs; PWUD in the chemsex scene; immigrants from countries of the former USSR; homeless populations; gender specific services; people returning from other countries; and people who buy drugs online.

Figure 50: In the previous year, did you start providing services for any new group(s) of PWUD?

This question had 33 responses: Yes: 12 FPs



Case 4:

New groups of PWUD for which services were provided

New group of PWUD: Chemsex community.

Activities conducted by our NGO:

Peer-to-peer support, linkage with mental health professionals.

Chemsex is a quickly rising trend in the gay community and involves very risky behaviour related to blood borne infections. As an association of HIV+ people, we took that very seriously in terms of prevention and peer support. Also, there are no specialised services for that and people do not have somewhere to get support.

The NGO reached more than 80 people from this community over the past year.

Source: NGO Positive Voice, Athens/ Thessaloniki (Greece).



PWUD whose needs are not met

Figure 51: In the previous year, did you come across any new group(s) of PWUD for whom your organisation, or any other organisation, are currently not providing any services?

This question had 33 responses: Yes: 3 FPs.



FPs in Skopje, Bucharest, and Stockholm mention groups for whom no services are yet provided: people who use non-opiate drugs (MDMA, amphetamine, LSD, magic mushrooms, cocaine) and young people and PWUD at parties.

Case 5:

PWUD whose needs are not being met

PWUD: People who use non-opiate drugs (MDMA, amphetamine, LSD, magic mushrooms, cocaine).

It is not a new group, but we haven't developed services for them yet.

How long has this group been around? More than 3 years.

What is needed for this group of PWUD includes the following:

- 1. Social services, legal services, services from a psychologist and psychiatrist.
- 2. Additional harm reduction outreach activities at music events such as:
 - Setting up an info-stand with educational materials on latest trends in

drugs and the risk of drug use; sexual health issues; and information on prevention of HIV, HCV and other blood borne infections.

- Highly experienced and educated outreach team that will provide: advice on drug use and risks associated with it; medical advice, including first aid; legal advice when approached by the police, etc.
- Display posters with a chart of drug combination risks which will be visible and easily accessible.
- Have an internet connected laptop for team members to look up pictures and information on specific substances; also to check if a substance is already registered in an online database of lab tested substances.
- Have harm reduction supplies such as: condoms and lube; straws for snorting; syringes for measuring a liquid dose; vitamins and minerals (Vitamin C, Magnesium).
- Have a list of local services to give to people seeking help with substance use or sexual health issues.

Challenges

The new shortened questionnaire provided improved data collection. For example, the scope was now more clearly defined (fewer questions) and narrowed down (to the local level). A number of issues previously faced were, therefore, solved. However, the new questionnaire continued to have some flaws that came to light especially while analysing the data received and include the following:

Corona crisis

The corona crisis may have led CSOs staff to have to spend their valuable time in addressing the additional needs of PWUD in their services, rather than answering questions and doing research. This may have led to the fact that a number of FPs mentioned that nothing happened, or filled in very little while answering the questions.

No focus groups (as planned)

Based on feedback from the previous monitoring round, the questionnaire was too long. Consequently, focus groups/group discussions were introduced with relevant local stakeholders to discuss new drug trends and to jointly answer the questions. A number of FPs had already agreed to do the data collection in this way. However, during the period of data collection, most, if not all, participating cities suffered from corona-related measures. This included severe restrictions in organising gatherings in many cities, and the number of people allowed to participate in those face-to-face meetings.

Issues in interpreting the data

Issues were faced in interpreting the data received. For example, if a FP provides very short answers to the questions, does that mean that little information was available, or that the way of questioning is not suitable for gathering data on new drug trends? On the other hand, some FPs replied to all questions with a lot of information. Does that mean that in these cities a lot is going on, or merely a very dedicated and enthusiastic person filled in the answers?

What people think is in the substance versus lab-tested information

Another limitation of the way data is collected is the differences between self-reports of the content of, for example, a new substance (see Case Study 1) versus information that comes from a lab analysis of the sample. Self-reports of substance content can

give crucial information on new substances, such as its general effects and appearances, however as consumers' experience is strongly influenced by the set and setting (context and psychological state), they also have to be carefully assessed. Comparisons between the strength perceived by users and the effective rate of psychoactive substance as determined through a full-blown analysis have demonstrated that they can be different.

PWUDs share information with Harm Reduction service providers, this is very early drug trend information that can aid the development of as close to real time policy responses at a local level. The gathering of intelligence from Harm Reduction services is important; however, we know it is not research, it is not lab-tested information and it is not empirical data. Harm Reduction Intelligence gathering is important information and requires a continually developing system of local corroboration and verification."

Definition of 'a trend'

Caution is also required when talking about a trend. Without a clear definition, questions regarding trends may leave space for different interpretations between different FPs. It would be useful to develop indicators, such as the number of persons involved, the length of the period in which the phenomenon is observed, and others. Although the term has been excluded from the questionnaire, reference to trends is likely to really mean 'observations'.

Reliability of some answers

As has been the case in 2019, sometimes information given by FPs is very different from what other (official) sources report. We see this as helpful information that can inform the debate on drug policy at local/national level.

Staff at the Correlation HQ have limited capacity to follow-up on the questionnaire. Follow-up has been undertaken concerning all non-returned questionnaires and, in a few cases, when replies have been



unclear. However, there is no capacity to do this in all cases. It is wise to allocate more time for this, as it may reduce some of the limitations mentioned elsewhere.

Conclusions

Data collection regarding new drug trends by CSOs has improved compared to previous years, although the level of responses in terms of input or quality still differs greatly. C-EHRN has contributed to a better understanding of emerging drug trends, at least at the local level. The importance and added value of CSOs providing around-level information on developments regarding drug use at local level can be clearly seen, but work still needs to be done to make full use of these resources. Also, continuous attention should be paid to the auality of the data, and to streamlining the information, such as in the level of detail of information provided. Finally, adequate and qualitative data collection is more time consuming that initially expected. Disseminating updates on new drug trends for all FPs several times a year is, at present, not possible. There is lack of capacity within C-EHRN to do this, but also the monitoring of new drug trends is not the core business of CSOs in the field and, therefore, to do so is usually an extra task on top of the already heavy workload on most professionals working with PWUD. Furthermore, COVID-19 has made the workload even areater.

While reports of emerging new substances in some cities (such as synthetic cannabinoids or synthetic cathinones, as well as the increased use of methamphetamine and GHB in chemsex scenes in a number of cities) are in line with other sources, such as the European Drug Report (1) or the World Drug Report (3) other FPs mentioned lesser known NPS for the first time last year, such as Isotonitazene by the Estonian FP in Tallinn. Data, however, is still rather limited since most of FPs do not have much input. This is something that C-EHRN plans to improve next year by changing the data collection methods, possibly including online group discussions and interviews.

References

- 1. European Monitoring Centre for Drugs and Drug Addiction. European Drug Report 2020: Trends and Developments. Luxembourg: Publications Office of the European Union; 2020.
- 2. UNODC. Current NPS Threats [Internet]. Vienna, Austria; 2020. Available from: https://www.unodc. org/unodc/en/scientists/current-nps-threats.html
- 3. UNODC. World Drug Report 2020 [Internet]. Vienna, Austria: UNITED NATIONS; 2021. Available from: https://wdr.unodc.org/wdr2020/
- Tammi T, Rigoni R, Matičič M, Schäffer D, van der Gouwe D, Schiffer K, et al. Civil Society Monitoring of Harm Reduction in Europe, 2019. Data Report [Internet]. Amsterdam; 2020. Available from: https://www.correlation-net.org/wp-content/ uploads/2020/02/C-EHRN_monitoring_web.pdf



COVID-19 and Harm Reduction



Introduction

Since the beginning of 2020, European countries have experienced an unprecedented public health threat with the emergence of COVID-19. People who use drugs (PWUD) and harm reduction services in Europe are uniquely vulnerable during the pandemic. European countries implemented a variety of virus containment strategies, such as border closures, service reductions, increased police presence, as well as diversion of sterile supplies and staff to hospitals [1]. In addition to these disruptions, PWUD may face an increased risk of exposure to COVID-19 due to crowded living conditions and may be at higher risk of severe complications should they contract the virus [1,4]. Preliminary research on the pandemic's effect on PWUD and drug services in Europe has been conducted by the European Centre for Monitoring of Drugs and Drug Abuse (EMCDDA). To complement this, C-EHRN has collected the ground-level experiences of Civil Society Organisations (CSO's) providing vital harm reduction services from May to July 2020 in cities across the European region, as well as data from PWUD themselves. This data from C-EHRN Focal Points (FP's) and PWUD examines both the challenges and successes during the first wave of the pandemic and provides timely information on the real-life situation, with valuable lessons learned and areas of focus for future advocacy.

The COVID section of the C-EHRN Monitoring Tool consists of ten questions on COVID-related impact on harm reduction services and on PWUD in FP organisations or cities. A separate questionnaire was sent to PWUD with four questions covering the COVID pandemic. Data was collected from May to July 2020, reflecting the first wave of the pandemic.

Results

Challenges faced by harm reduction services

The first part of C-EHRN monitoring assessed the various challenges faced by harm reduction services during the pandemic. CSOs were asked whether their harm reduction services were affected, and which modifications to services they undertook during the first wave of the pandemic.

Figure 52: Has the COVID-19 pandemic affected daily harm reduction practices?



The majority of FPs reported that the pandemic had indeed affected their daily harm reduction practices. However, FPs from three cities - Antwerp, Helsinki and London - reported that the COVID-19 pandemic did not affect daily practices in their organisations.

Harm reduction facilities were then asked which challenges their services had faced during the first wave of the pandemic.





Figure 53: Challenges faced by harm reduction services during the first wave of the pandemic

The majority of FPs reported rapid adaptation of services. FPs in Vienna, Athens and St. Petersburg reported closure during the pandemic, and FPs in Tirana, Rijeka, Berlin, Porto and St. Petersburg reported that PWUD could not access services in their city due to lockdowns.

No FPs reported limitations in harm reduction supplies from the services that remained open. The responses shown in Figure 2 correlate with preliminary findings from EMCDDA's research which suggests there has been a decline in the availability of European drug services during the early months of the pandemic. Services still providing care to PWUD needed to rapidly implement new hygiene and social distancing measures. Additional challenges found by the EMCDDA include a lack of adequate protective equipment for staff, challenges linked to remote technology, staffing shortages, and managing the demand for substitution treatment [3].

Major adaptations noted in the comments by FPs included service windows, new online services, home delivery services, a new need for food and water distribution, cessation of drop-in services, and disruption in HIV/HCV services. FPs in Bucharest, Porto and Copenhagen reported an increase in programmes and opening hours.

- **66** Our facilities were open throughout the COVID-19 period and care was maintained at different stages. The entire service was adapted to the new situation. From an initial stage, access to consumption spaces was maintained, limiting the number of people; the delivery and receipt of material that is sterile or already used for injectors as well as for smokers; delivery of containers; nursing care; Hep C and supervised antiretroviral treatments; replacement of naloxone kits; and the delivery of food packs. Subsequently, other spaces and services were adapted, week-by-week, in order to expand attention to all areas: showers; laundry service; permanence and food in the drop-in space with safe distances; use of masks; and all measures including disinfection and cleaning with specific products." (FP Barcelona, Spain)
- (6 We had to adapt our services. No drop-in or groups. Focus was on providing services on a one-to-one basis on outreach, in-reach/fixed site. (FP Dublin, Ireland)
- Increase in programme and opening hours, development of new components (mask, disinfectants, and massive food distribution, "home" delivery.
 (FP Bucharest, Romania)
- (6 We expanded daily opening. Limited access to the clinic service, but expanded access to outreach. Isolation facilities for the homeless, including undocumented migrants (after a while). Night shelter for street homeless regardless of nationality. (FP Copenhagen, Denmark)

We adapted our services – not only with protective material – but, for example, in GIRUGaia we began intensive telephone contact with our clients 2 times a week []To respond to users' needs, and to help them during the social isolation period, Kosmicare has made available online its psychological support and harm reduction counselling sessions. This service was created to support people that are experiencing difficult and/or challenging situations related to the use of drugs, that are facing some negative drug-related psychological consequences or difficulties managing their drug use. (FP Porto, Portugal)

Positive changes in harm reduction during the pandemic

While the COVID-19 pandemic has wrought with many hardships for PWUD and harm reduction CSOs, it has also provided a unique opportunity for innovative practices, as well as a few potential "silver linings". Preliminary findings have shown a large demand for opiate substitution treatment (OST) and the transition to online harm reduction services.

FPs were asked about the potential positive changes in harm reduction services at their organisation during the first wave of the COVID-19 pandemic, with their responses shown in Figure 54.

Education on COVID-19 was endorsed by the majority of FPs. Many of the other changes centred around OST, such as increased length of prescriptions and take-home doses; new phone or telemedicine services; or increased interest in, and enrolment onto, OST. In Prague and Novi Sad, FPs reported that new forms of OST were available. Outreach services and access to housing and shelters, and naloxone distribution also ranked highly. The responses per city/organisation can be seen in Table 12. Other positive changes mentioned by FPs include more shelters for people experiencing homelessness; new methylphenidate prescriptions for stimulant use (Prague); more attention and funding to outreach work and the supply of sterile syringes (Athens); nasal Naloxone training for staff and PWUD (Lisbon); distribution of food and hygiene products (Bucharest); increased volunteering for outreach (Novi Sad); and better hygiene and tranquillity in DCRs and drop-in sites due to a lower number of service users (Amsterdam).

Figure 54: Positive changes or innovations in harm reduction services





Table 12: Positive changes reported by FPs in response to the first wave of the COVID-19 pandemic

	Im- prove- ment in OST ser- vices	In- creased length of prescrip- tions and take- home doses	New phone or telemedi- cine services for OST	In- creased interest and en- rolment in OST	Home delivery of OST	New forms of OST available	Improvement regarding access to housing and shelters	Increased or started Naloxone distribu- tion	New out- reach services added
Albania, Tirana		x	x						
Austria, Vienna		х	x		x		x		x
Belgium, Antwerp	x	x	x	x	x				
Croatia, Rijeka			x						
Cyprus, Nicosia								x	х
Czech Republic, Prague		x	x			x	x		
Denmark,		~	^			^	^		
Copenhagen	х	х		х	х				х
Estonia, Tallin			x		x				
Finland, Helsinki		x							
France, Paris		х	x				х	x	x
Georgia, Tibilisi		х							x
Germany, Berlin	x	x	x	x	x		x		
Greece, Athens									х
Hungary, Budapest									x
Ireland, Dublin	x	x	x	x	x		x	x	x
Italy, Milan/ Rome	x	x	x	x				x	
Lithuania, Vilnius								x	
Luxembourg, Luxembourg	x	x							
Norway, Kristiansand									x
Poland, Krakow	x	x	x	x					x
Portugal, Porto		x	x	x	x		x	x	
North Macedo- nia, Skopje									x
Romania, Bucharest									x
Russia, St Peters- burg									x
Scotland, Glasgow		x	x	x	x	x	x	x	x
Serbia, Novi Sad		x			~				
Slovakia,									
Bratislava Slovenia,	Х								
LJubliana						x			x

	Im- prove- ment in OST ser- vices		New phone or telemedi- cine services for OST	In- creased interest and en- rolment in OST	Home delivery of OST	New forms of OST available	Improvement regarding access to housing and shelters	Increased or started Naloxone distribu- tion	New out- reach services added
Spain, Barcelona	x	х	x	х	x		x		
Sweden, Stockholm		x							
Switzerland, Bern	x	х			x				
Netherlands, Amsterdam	x	x	x	x	x		x		x
Ukraine, Kiev		х	x	х	х				
UK, London	x	x	x	x	x		x	x	

FPs were asked to comment on other positive changes in harm reduction services during the first wave of the COVID-19 pandemic.

- Set New sheltering possibilities for the homeless, new substance methylphenidate available in the pharmacology assisted treatment to metham-phetamine. (FP Prague, Czech Republic)
- 66 All OST was delivered to users' own homes. Outreach OST to the homeless. (FP Copenhagen, Denmark)
- Generation and more money directed to the street work and to the supply of sterile injections. (FP Athens, Greece)
- 66 Mobile DCR supported the emergency shelters set up in Lisbon. Nasal naloxone training of staff and some clients of the emergency shelters in Lisbon. Increased length of prescriptions and take-homes - only in drug treatment centres, not low threshold. (FP Porto, Portugal)
- (Massive food and hygiene product distribution. (FP Bucharest, Romania)

In the first wave of the epidemic, a 24-hour homeless shelter for about 20 people opened in Ljubljana. The shelter was open for about two months. Besdies, due to the suspension of public transport, some users did not have access to substitution medicines, so our association took over the distribution of medicines for these users. (FP Lubljiana, Slovenia)

- Clients volunteer to help deliver materials in the field. (FP Novi Sad, Serbia)
- Specific and personalised health follow-ups for our clients, especially those who had symptoms. (FP Barcelona, Spain)
- **66** Larger enrolment in OST by >50% but no new patients in. (FP Stockholm, Sweden)
- 66 Better hygiene in DCRs, more tranquillity/less aggression and frustration in drop-in centres, and not only improved access but also improved quality (more independent bedrooms) at shelters. (FP Amsterdam, Netherlands)



Difficulties faced by PWUD during the pandemic from the perspective of harm reduction service providers

FPs were asked to identify potential difficulties faced by PWUD in their countries during the pandemic and to rank them according to difficulty.

Figure 55: Main difficulties faced by PWUD from the perspective of harm reduction CSOs



Figure 56: Social isolation for PWUD during the pandemic



In contrast, limited access to OST in their country was ranked by FPs as the least problematic. It was only noted as very problematic by respondents in Slovenia, as shown in Figure 57. This might be related to the fact that several positive developments regarding access to OST occurred during the first wave of the COVID-19 pandemic, as previously mentioned.

Figure 57: Limited access to OST during the pandemic



Limited access to drugs, adulterated or low-quality drugs, and an increase in prices were perhaps not as widespread and/or problematic as anticipated by preliminary research [1]. Social isolation, as well as an increase in mental health disorders, may be related to the severity of the COVID-19 outbreak, as well as the degree of lockdown and limitations to services in the region.

Voices of PWUD during the pandemic

Figure 58: Harm reduction access during the pandemic by PWUD



To complement this data from the harm reduction FPs, a new survey this year for PWUD included four questions on harm reduction during the pandemic. PWUD were asked to rank their access to various harm reduction services in their city during the pandemic. OST, housing, online harm reduction, and outreach work were ranked by PWUD as easiest to access. The majority of PWUD respondents noted a reduction in the types of services available, though some new services (such as delivery) were noted.

- **C** Services were cut in half during COVID. (Helsinki, Finland, PWUD)
- C Everybody closed the services and left alone the drug users in the streets to police appetite, drug users were searching for WATER and a TOILET not drugs. Very big disappointment from everybody. (Athens, Greece, PWUD)
- General Heroin was delivered to patients' homes by bicycle courier. Fewer drug consumption rooms were open.
 (Zurich, Switzerland, PWUD)
- Conline harm reduction services were more available; the OST programme gave us Methadone for all week; we had bigger amounts of sterile equipment, but when we spent it was not easy to get a new one because of the state of emergency and police hours. (Novi Sad, Serbia, PWUD)

Changes in drug use by PWUD during the pandemic

PWUD were asked if their drug use, or that of their peers, was affected by the COVID-19 pandemic in terms of which drugs were used, and how they prepared or took them. Early research by the EM-CDDA suggested an overall decline in drug use among PWUD, but with heterogenous results and significant variability by country and drug type [1]. Many respondents noted an increase in alcohol use. No change in drug use was reported by ten participants. Respondents noted an increase in al-


cohol use in Vienna, Budapest, and Glasgow (n=3). Participants from Copenhagen, Athens, Helsinki, Budapest, Kristiansand, Stockholm and Porto (n=7) noted that drugs were more expensive. In particular, the respondent from Finland noted the price of buprenorphine had "doubled." Participants from Copenhagen, Athens, Luxembourg, Porto and Stockholm (n=5) all noted a decrease in the quality of substances available. More buprenorphine on the streets was noted by participants from Prague, whereas more methadone in the street supply was noted in Novi Sad. Cannabis was noted as harder to come by from respondents in Dublin and Porto.

- For a period, the drugs were bad and expensive.
 (Copenhagen, Denmark, PWUD)
- C Amphetamines got more common, Buprenorphine and Benzos were not available. If you got buprenorphine, the price was double. (Helsinki, Finland, PWUD)
- I have been drinking a lot more alcohol and buying more prescribed tablets on the street (Tramadol).
 (Glasgow, Scotland, PWUD)
- (6 Higher prices, and worse quality and emergence of new NPS. (Stockholm, Sweden, PWUD)
- (C Less cocaine was consumed and increasingly at home. There were more purchases in Darknet. (Zurich, Switzerland, PWUD)

Health of PWUD during the COVID-19 pandemic

Participants were asked to comment on the mental and physical health of themselves or their peers during the COVID-19 pandemic. An increase in mental health disorders was noted by thirteen PWUD out of the 24 respondents.

- Social distancing has affected my psyche, I have been too scared to meet anyone or even go outside. I also have used more alcohol. One of the most traumatised things in my life. (Helsinki, Finland, PWUD)
- (C More need of psychologist and psychiatric services. (Tallinn, Estonia, PWUD)
- (Mental health issues are prevailing in the community due to many factors and the lack of psychosocial support. The lockdown had a tremendous affect on mental health, and people are more disorganised and chaotic. (Athens, Greece, PWUD)
- ((It was hard for me, because I couldn't go to a psychiatrist. One my friends died during the pandemic and it was a very difficult moment for me. (Novia Sad, Serbia, PWUD)
- C Peoples' mental health has been affected negatively and suicide rates have gone up. (Stockholm, Sweden, PWUD)

Physical health was also noted to be affected by a number of respondents, some of whom attributed it to the use of tobacco and alcohol.

- (6 I have COPD and it has got worse as I am smoking a lot more tobacco. (Glascow, Scotland, PWUD)
- (C I was sick before the pandemic, having a bad septic infection and then septic arthritis. It was hard to get treatment and get myself a hip replacement during the pandemic. (Bucharest, Romania, PWUD)

Other findings noted by respondents include a closure of health services, stress from lack of work, and a rise in violence and peer-on-peer robbery and violence.

PWUD life under the COVID-19 pandemic

Respondents were asked to comment on other ways the COVID-19 pandemic has affected PWUD in their city. Frequent themes mentioned by PWUD include:

- Problems with the police (e.g. in Antwerp, Athens, Budapest, and Novi Sad).
- Difficulty gathering and meeting to obtain drugs (e.g. in Tirana and Vienna).
- Lack of tourists/lack of money in general (e.g. in Moscow, Glasgow, Amsterdam, Luxembourg and Dublin).
- Plight of the homeless (e.g. in Ljubljana, Krakow and Cologne).
- Lack of drop-in centres for social contact (e.g. in Greece).
- It was almost impossible to meet up; therefore, buying drugs was more difficult.
 (Vienna, Austria, PWUD)
- 66 Mostly people who used buprenorphine tried to switch to benzos before they, too, stopped being sold. (Helsinki, Finland, PWUD)

- 66 Most of the respondents experienced difficulties related to homelessness, not substance use per se. (Krakow, Poland, PWUD)
- (C New people couldn't enter an OST programme because of Covid. (Vilnius, Lithuania, PWUD)
- (6 It has been a lot more difficult to get around with money. No source of income. No easy access to drop-in centres, so no social control. (Amsterdam, Netherlands, PWUD)
- Some of my friends were penalised for walking during the police hour. (Novi Sad, Serbia, PWUD)
- (C There is loads more begging, some sex workers have been forced to start begging as they have no customers. This is causing a lot of conflict with the established beggars fighting over the best spots. (London, UK, PWUD)

Overdose and the COVID-19 pandemic

Harm reduction focal points were asked about whether they noted an increase in overdose (OD) during the pandemic. This was hypothesised by many experts early in the pandemic to be a potential risk due to more PWUD using alone, less access to naloxone, an increase in adulterated substances, etc. [1].

Figure 59: Increase in OD observed during the pandemic





While the majority of the C-EHRN FPs did not note an increase in overdose, one-third did not know. FPs in Kiev, Novi Sad, and Stockholm reported increased rates of overdose during the pandemic.

Government response for PWUD during the COVID-19 pandemic

FPs were asked to evaluate whether their government responded to protect PWUD and harm reduction professionals during the pandemic in their country.

Figure 60: Government response to protect PWUD and harm reduction services during the pandemic



When asked to comment and give examples of the most frequent government responses to protect harm reduction and PWUD, FPs mentioned a number of items. Five FPs mentioned guidelines provided by their government (Vilnius, London, Paris, Helsinki, and Nicosia). FPs from ten countries reported their government assisted with the expansion of OST or the loosening of regulations regarding take-home doses (Tallin, Paris, Berlin, Tbilisi, Dublin, Rome, Kristiansand, Barcelona, Amsterdam, and Kiev). Distribution of protective and hygiene equipment was reported by seven FPs (Nicosia, Prague, Tallin, Porto, Barcelona, Amsterdam, and Bern). New shelters for the homeless were mentioned by five FPs (Vienna, Germany, Dublin, Barcelona, and Amsterdam).

FPs were asked to rate the government response to the pandemic on a scale from very poor to above average.

Figure 61: Focal Point rating of Government responses for PWUD



III=Above Average III=Average III=Below Average III=Very Poor

- All medical staff had access to necessary protective materials. Medical services must continue, this includes OST and heroin maintenance treatment. The National Government offers financial security to ensure that municipalities do not cut funding of low threshold services, thus ensuring continued services. Additional funding was made available for services to facilitate COVID transitions and the opening of additional shelters for the homeless. (FP Amsterdam, Netherlands)
- C During the pandemic, care and open consumption spaces have been maintained, as well as care in all areas, adapting it to the new situation and preventive measures. Lockdown spaces and shelters have been opened for people who live on the street and specifically with consumption spaces for people who live on the street and use drugs. Specific supervised programmes have been started for people with problematic alcohol consumption, avoiding withdrawal syndromes and allowing their lockdown. Rapid induction into OST programmes have been managed and these programmes have been adapted

to enable longer take-home care to guarantee the possibility of lockdown while maintaining treatment. The availability of all available sterile consumable material in the quantities required has been maintained, as well as sexual prevention material, for smoked consumption and naloxone kits in order to prevent possible adverse reactions. There has been permanent contact by the government and monitoring of the adaptation of the different facilities, giving updated information on the different measures that had to be adopted at all times to prevent infections. (FP Barcelona, Spain)

- (C There was implemented 5 days dose delivery for OST clients for the first time in Georgia. We advocated for it for years, but COVID gave us this possibility. (FP Tbilisi, Georgia)
- C The Ministry of Health released recommendations to low threshold centres on protective measures. They were not easy to implement and practically no funding was allocated to buy masks or disinfectants. Vilnius municipality gave access to COVID-19 testing for low threshold workers. (FP Vilnius, Lithuania)
- (CRapid response, OST in low threshold, more shelters, information. (FP Berlin, Germany)
- C A message came from the National Institute for Health Development that during the pandemic, drug users must receive services, and we did not close any services, we adapted them; for example, counselling services (went online, but also met with clients physically if necessary), to reduce direct contacts and risks. Clients received an unlimited amount of injecting accessories

and methadone for a longer period of time at home. We were provided with disposable masks and gloves by the Government. (FP Tallinn, Estonia)

- (C There was no specific approaches to treatment which meant that different cities had different approaches. There was no national strategy for COVID testing of drug users. (FP Copenhagen, Denmark)
- (C The help came after long time, almost after the main crisis. Distribution of protective equipment. The help from the government is still chaotic even in these days. (FP Prague, Czech Republic)

Role of CSO's during the pandemic

The next section asked the FPs to comment on the role of CSOs in advocating for, or increasing, services for PWUD during the pandemic. Most frequently mentioned was working with the government, followed by service provision and advocacy for PWUD.

Interfacing with the government and/or municipality during the pandemic was most frequently reported by thirteen FPs as the role for CSOs. The next most frequently mentioned was providing direct services to PWUD. Providing advocacy for PWUD was noted by ten FPs. Distributing information about COVID-19, and hygiene, was reported by nine FPs. Five FPs specifically mentioned addressing homelessness as part of their role during the pandemic. Two FPs, in Vienna and Vilnius, reported that CSOs did not play a role during the pandemic. The FPs in Helsinki and Stockholm reported that their role involved interacting with the media.



C The CSO including my organisation have been highly committed to increasing services for PWUD. Our work helped the public authorities to take measures. (FP Paris, France)

Figure 62: Role of CSOs reported during the pandemic



We adapted our services so that all drug users could get what they needed. We promoted the reduction of risk behaviours related to drug use during a pandemic, as well as guidance on behaviours and activities that are important for controlling viruses and reducing the risks associated with Covid-19. We are a source of information, communicating with other institutions such as the Unemployment Insurance Fund, shelters, health care institutions, medical institutions, doctors, etc. (FP Tallin, Estonia)

- COur organisation took part in regular calls with the city of Bratislava, when we spoke about implementing quarantine measures for homeless people. Our goal was to not base any new service related to COVID-19 for homeless people on abstinence as one of the requirements. Also, we became members of a working group at the ministerial level regarding homeless people and COVID-19, so people who use drugs will not be missed out of the services. (FP Bratislava, Slovakia)
- (C The CSOs played a big role in participating in meetings in order to establish the needs and gaps in services during the pandemic, the lockdown, and to approach problems affecting PWUD living on the street." (FP Barcelona, Spain)
- We drew the attention of the authorities and the public to the fact that there are social groups for which the COVID-19 epidemic caused specific problems (e.g. homeless people who could not stay there without a home, or the difficulties of people living with HIV access to consultations and medical examinations).

(FP Krakow, Poland)

((It was also pointed out that these programmes will contribute to the prevention of COVID among people who use drugs and other marginalised communities by informing, distributing the necessary materials. The government has announced a call for funding civil society organisations to prevent COVID and support marginalised communities. We distributed food packages, hygiene products, brochures, vitamins to people who use drugs and sex workers. (FP Skopje, North Macedonia)

- We created an online database to find services that are open during the lockdown; we organised video conferences for service providers; we translated guidelines; we created posters; HepaGo is doing street outreach, etc. (FP Budapest, Hungary)
- More counselling about health matters and education about hygiene, distribution of protective materials (surgical masks, latex gloves, disinfectants). But mostly, every CSO worked based on their intuition since there were no specific guidelines for CSOs, and especially none for harm reduction services. (FP Rijeka, Croatia)
- Many organisations in Ukraine have not stopped their work and have adjusted the services to the conditions that have developed. For example, our organisation fully provided PWUD with their previous access to services, and also carried out transportation for clinic registration for HIV and receiving ART. We also began to advise people on rights and help them get vital services. (FP Kiev, Ukraine)

Potential for innovation and lessons learned during the pandemic

Nearly all FPs endorsed the pandemic as an opportunity to practice innovative harm reduction. Of those that responded "no", the FP in Vilnius noted it was, "too difficult just to focus on the basics", and the respondent in Ljubljana who responded "no" commented, "it was improvisation."

Figure 63: Opportunity to practice innovative harm reduction services



FPs were asked to freely comment on innovation during the pandemic. Most frequently mentioned was a transition to online services by eight FPs. New home delivery services, and positive changes to OST, were the next most frequently mentioned by five FPs. New outreach services were mentioned by three focal points, and two CSOs mentioned new food services. The FP from Tallin noted that their clientele seemed pleased by the switch to online services and would like this to continue. In Amsterdam, the FP noted they were able to expand OST to non-Dutch European migrants, whereas the respondent in Copenhagen commented that there was still no OST available for migrants. Extending drug-checking services to more vulnerable people was under discussion according to the focal point in Porto. Respondents from Barcelona noted a more sympathetic attitude of the state towards PWUD living on the street, and also reported some more personalised follow-up cases and interventions to improve their situation.



It has widely allowed for more digitalisation and more attention to hygiene. Some DCRs, for instance, have improved their ventilation and access to soap and water, or they intend to do so. Also, some services have really picked up outreach care very well; for instance, offering meals, work, medication or even paraphernalia at home. Several wish to continue these services. In Amsterdam, they have started a pilot offering European migrants methadone treatment. Thus, access to methadone has increased for a small group of people.

FP Amsterdam, Netherlands)

- 66 New phone for outreach work can be a major tool to approach clients and to go deep into the relationship with them. Possibility of extending the drug checking service to more vulnerable people. (FP Porto, Portugal)
- (6 It usually takes a lot longer for our clients to embrace innovation, but in a pandemic and curfew due to Covid, maximum reduction of public transport has made it easier for our clients to get services through the innovations we have introduced. (FP Skopje, North Macedonia)
- **C** There was no opportunity to implement innovative harm reduction service practise because we were focused on crisis intervention and survival. (FP Bratislava, Slovakia)

PWUD and lessons learned from the pandemic

PWUD were asked to comment on their perspective for lessons learned from the pandemic. The importance of harm reduction was mentioned most frequently (6 PWUD), followed by expansion of OST and increase in OST flexibility (5 PWUD). The importance of peers was noted by two respondents, and the damage of stigma to the most vulnerable in the community during a pandemic was noted by two participants.

Content of the provided in the Corona crisis. Content of the provided the provid

(Zurich, Switzerland, PWUD)

- We could have learned a lot if PWUD were involved in health and social policies, in service monitoring and evaluation...this way we only learned global normalising lessons that may serve most but are very insufficient for marginalised groups. (Porto, Portugal, PWUD)
- 66 We should have some savings of money, drugs and sterile equipment in case something like this happens again. (Novi Sad, Serbia, PWUD)

Conclusions

Overall, FPs from countries with more strict lockdown procedures reported more reliance on home delivery and mobile outreach services. Positive FP rating of government responses correlates well with other positive outcomes reported during the pandemic. FP and PWUD from Barcelona, Zurich, Tallinn, Dublin, Berlin, and Amsterdam all rated their government responses as "above average" and noted other positive outcomes such as expanded OST, better sheltering for the homeless, etc. Conversely, in regions where the FP rated government responses as "below average", the FP and PWUD also reported more difficulties, such as problems with the police, and with sterile supplies for harm reduction.

In Western Europe, FP responses varied depending on the city, with many positive reports coming from Barcelona, Dublin, Amsterdam, Zurich, Paris, Berlin, and Lisbon detailing many positive harm reduction changes, whereas others referenced a more heterogeneous picture during the pandemic. Many FPs noted rapid law changes to expand OST, and improvement in housing for the homeless. Scandinavia was more mixed, with more positive responses regarding the pandemic coming out of Copenhagen and more negative comments from Stockholm and Helsinki. Central Europe (in the EU) reported very few government responses for PWUD aside from Prague, and more negative responses regarding the pandemic's effects on harm reduction and PWUD were stated, though some positive innovations were noted. In Southern and Eastern Europe, responses were very mixed. FPs in Skopje and Tbilisi reporting more positive responses, whilst those in Athens and St. Petersburg painted a more grim picture during COVID-19 times. The Balkan states were also mixed, with the FP in Tallin reporting a dramatically more positive picture than the FP in Vilnius regarding harm reduction services during the pandemic.

Important lessons to be learned from the initial response during COVID-19 is that services can be adapted rapidly and that expansion of OST, outreach services, and home delivery are vital components of harm reduction during a pandemic. Social isolation and mental health are important concerns for PWUD, with increased outreach and digital connection/phone services being important considerations to overcome these problems. Certain pandemic responses detailed by FPs demonstrate a window of opportunity to overcome political will in implementing rapid-scale changes to harm reduction services by rolling out new services or service changes. Important future considerations for future second and third waves of the pandemic are to prioritise expansion of outreach services, OST, and housing for vulnerable populations in connection with vital harm reduction services. Other infectious disease needs (such as HIV/HCV testing and treatment) must not be forgotten, and advocacy must continue to maintain the needed sterile supplies for harm reduction provision during a pandemic. Harm reduction services are a vital component in a pandemic response in caring for the vulnerable population of PWUD in Europe, and continued research as to the longer-term influence of the pandemic on these vital services will be important for future C-EHRN monitoring.



References

- EMCDDA. Impact of COVID-19 on patterns of drug use and drug-related harms in Europe.
 2020. https://www.emcdda.europa.eu/system/ files/publications/13130/EMCDDA-Trendspotter-Covid-19-Wave-2_1.pdf
- EMCDDA. EMCDDA update on the implications of COVID-19 for people who use drugs (PWUD) and drug service providers. 2020. https://www.emcdda.europa.eu/system/files/ publications/12879/emcdda-covid-update-1-25.03.2020v2.pdf
- EMCDDA. Impact of COVID-19 on drug services and help-seeking in Europe. 2020. https:// www.emcdda.europa.eu/system/files/publications/13073/EMCDDA-Trendspotter-Covid-19_ Wave-1-2.pdf
- Holloway, Ian W., Anne C Spaulding, Ayako Miyashita Ochoa, Laura A Randall, Adrian R King, HBOU Study Team, and Paula M. Frew. 2020. COVID-19 Vulnerability among People Who Use Drugs: Recommendations for Global Public Health Programmes and Policies. Journal of the International AIDS Society 23 (7): e25551. doi: 10.1002/jia2.25551

Endnotes

- 1. Although Scotland is part of the UK, it is treated separately in this report.
- 2. The names of participant can be found under "contributors".
- FP names, organisations, cities and countries can be found under "contributors". Both Italian FPs submitted one combined questionnaire. We have also received contributions from: (1) Chinara Imankulova, of AFEW Kyrgystan, Bishkek, Kyrgyz Republic; and, (2) Karen Mamo from Malta. These last two, however, where not included in the current report due to geographic coverage (1) and time restrictions (2).
- 4. Data extracted from the network member survey conducted in 2020.
- https://www.correlation-net.org/wp-content/uploads/2020/05/C-EHRN_Monitoring_2020-Questionnaire.pdf
- Unless otherwise stated, data and trends on harm reduction service provision presented in this chapter are based on Harm Reduction International. The Global State of Harm Reduction 2020. London; 2020. https://www.hri.global/files/2020/10/26/Global_State_HRI_2020_BOOK_FA.pdf
- 7. Europe refers to all countries included in this report. Where other regions (Asia, Eurasia, Latin America and the Caribbean, Middle East and North Africa, North America, Oceania, Sub-Saharan Africa, and Western Europe) are mentioned, reference is being made to regional categories used in the Global State of Harm Reduction report. For a detailed list of countries in each region, please check Harm Reduction International's Global State of Harm Reduction web page: https://www.hri. global/global-state-of-harm-reduction-2020
- The table reports on availability, which does not necessarily mean accessibility. There might be serious barriers in access to these services on the ground
- WHO has set a NSP coverage target of 300 syringes per person who injects drugs per year to reach hepatitis elimination goals by 2030; World Health Organization. Combating Hepatitis B and C to reach elimination by 2030. Advocacy Brief. Geneva; World Health Organization, May 2016, Table 1, p3. https://apps.who.int/iris/bitstream/handle/10665/206453/WHO_HIV_2016.04_eng.pdf[4].

- 10. After the Global State of Harm Reduction 2020 was published, C-EHRN FP from Norway reported that the country has also started providing Heroin Assisted Treatment.
- Afghanistan, Australia, Canada, Denmark, Estonia, Italy, India, Mexico, Myanmar, Norway, Puerto Rico, Ukraine, the United Kingdom, the United States and Vietnam.
- 12. Naloxone is available in HR units but officially peers cannot distribute it, only nurses.
- Coronavirus disease (COVID-19) is an infectious disease caused by a newly discovered coronavirus. Most people infected with the COVID-19 virus will experience mild to moderate respiratory illness and recover without requiring special treatment. Older people, and those with underlying medical problems such as cardiovascular disease, diabetes, chronic respiratory disease and cancer, are more likely to develop serious illness.[41]
- 14. https://csidp.eu/
- 15. Includes also countries that are not part of the EMCDDA's data gathering network.
- 16. In 2019, Poland reported not having any guidelines, but in 2020 they reported having 'other guidelines': Program lekowy B.71 (Ministry of Health). https://www.intermedis.pl/pliki/2015_Program_lekowy_B_71_leczenie_pWZW_C_kuracja_bez_interferonu.pdf
- 17. Several different types of restrictions could be reported by one country.
- 18. Still in 2019, DAA's were not available in North Macedonia but are now available.
- 19. "This is because they will not walk to the hospital on their own initiative, you need to do outreach and motivate them - and there is no investment into this, only a few professionals do this work." (from the answer by the Hungarian FP).
- 20. In UK, treatment is provided free at the point of delivery so no reimbursement is needed.
- 21. In some countries, HCV activist groups different from drug user groups - have also played a major part but questioons about the these groups was not included in the C-EHRN monitoring survey..
- 22. Missing information from Skopje/Northern Macedonia.



- 23. Overdose deaths' refers to deaths that are caused directly by the consumption of one or more illicit drug. The term 'overdose prevention' refers to preventing both overdose without fatal consequences and overdose deaths.
- 24. In Austria, there is no specific strategy for OD prevention. There are, however, guidelines for dealing with illegal drugs and PWUD, and harm reduction is mentioned in documents such as: "Österreichische Suchtpräventionsstrategie, Strategie für eine kohärente Präventions- und Suchtpolitik Leitlinie Qualitätsstandards für die Opioid-Substitutionstherapie (OST)".
- 25. National harm reduction guidelines include OD management.
- 26. There is no national drug plan in Italy. The last one (2010) has never been approved by the regions due to a conflict between regions and the government. Some regions/cities have recommendations about OD prevention.
- 27. HOPS published guidelines in 2015 intended for physicians working in emergency services, hospital emergency departments, and CSOs working in the field of harm reduction.
- 28. There are OD prevention guidelines in Scotland and Wales, whilst there is take home naloxone guidelines in England, but not OD prevention. The London FP is not aware of any guidelines in Northern Ireland.
- 29. Only those PWU opioids who are enrolled in OST or a registered NSP.



C-EHRN envisions a fair and more inclusive Europe, in which people who use drugs, including other related vulnerable and marginalized people, have equal and universal access to health and social services without being discriminated against and stigmatized.

We advocate for a harm reduction approach that is based on solid evidence and on human rights principles, and addresses both health and social aspects of drug use.